Speed of random walks, isoperimetry and compression of finitely generated groups

By JÉRÉMIE BRIEUSSEL and TIANYI ZHENG

Abstract

We give a solution to the inverse problem (given a prescribed function, find a corresponding group) for large classes of speed, entropy, isoperimetric profile, return probability and L_p -compression functions of finitely generated groups. For smaller classes, we give solutions among solvable groups of exponential volume growth. As corollaries, we prove a recent conjecture of Amir on joint evaluation of speed and entropy exponents and we obtain a new proof of the existence of uncountably many pairwise nonquasi-isometric solvable groups, originally due to Cornulier and Tessera. We also obtain a formula relating the L_p -compression exponent of a group and its wreath product with the cyclic group for p in [1, 2].

1. Introduction

An important topic in group theory is the description of asymptotic behaviors of geometric and probabilistic quantities, such as volume growth, isoperimetric profile, Hilbert and Banach space compression on the geometric side, and speed, entropy and return probability of random walks on the probabilistic side. The study of these quantities falls into three types of questions. First given a group, compute the associated functions. Secondly the inverse problem: given a function, find a group with this asymptotic behavior. Thirdly understand the relationship between these quantities and their interactions with other topics in group theory, such as amenability, Poisson boundaries, classification up to quasi-isometry, etc. This paper contributes to solve the second question for finitely generated groups of exponential volume growth.

The first solution to an inverse problem for a large class of functions concerned compression gaps for non-amenable groups. The notion of uniform embeddings of groups into Hilbert spaces, more generally Banach spaces, was introduced by Gromov in [Gro93]. This notion has numerous geometric

Keywords: random walks, speed, entropy, uniform embedding of groups, wreath products, isoperimetry

AMS Classification: Primary: 20F69, 60B15, 46B85, 20E22.

⁽C) 2021 Department of Mathematics, Princeton University.

applications. Yu has proved that a finitely generated group uniformly embeddable into a Hilbert space satisfies the Novikov conjecture [Yu00]; later this result was extended by Kasparov and Yu to uniformly convex Banach spaces [KY06]. These results motivate further studies of embedding properties of groups into Banach spaces. The topic of distortions of bi-Lipschitz embeddings of finite metric spaces is originally studied in Banach space theory; it has found important applications in computer science algorithms since the work of Linial, London and Rabinovich [LLR95]. Similar to distortions of bi-Lipschitz embeddings, the \mathfrak{X} -compression gap measures quantitatively, for an infinite finitely generated group equipped with the word metric, the least possible distortion when one embeds it into a Banach space \mathfrak{X} ; see definitions recalled before the statement of Theorem 1.1. Arzhantseva, Drutu and Sapir proved that essentially any sublinear function is the upper bound of a Hilbert compression gap of width $\log^{1+\epsilon}(x)$ of some non-amenable group [ADS09]. Their construction does not provide a solution to the other inverse problems, because non-amenability forces volume growth to be exponential, speed and entropy growth to be linear and return probability to decay exponentially.

In the amenable setting, a partial solution to the inverse problem is known for volume growth, entropy or speed. Bartholdi and Erschler have proved [BE14] that for any regular function f(n) between $n^{0.7674\cdots}$ and n, there is a group with volume growth $e^{f(n)}$ up to multiplicative constant in front of the argument. This statement at the level of exponents was first obtained in [Bri14]. For any function between \sqrt{n} and n^{γ} for $\gamma < 1$, there are a group and a finitely supported measure with this entropy up to multiplicative constant by Amir-Virag [AV17]; see also [Bri13] for a statement with precision $n^{o(1)}$. Amir and Virag also showed that for any function between $n^{\frac{3}{4}}$ and n^{γ} , $\gamma < 1$, there are a group and a finitely supported measure with this speed up to multiplicative constant. These examples are all permutational wreath products, which are extensions of groups acting on rooted trees. They generalize constructions of Grigorchuk, who provided the first uncountable family of pairwise non-quasiisometric groups [Gri85]

In [NP08], [NP11], Naor and Peres have established quantitative connections between compression of uniform embeddings of groups into Banach spaces and speed functions of random walks. Their work has renewed interest in the longstanding question (attributed to Vershik) on what speed functions are possible for simple random walks on groups; see [AV17]. It also motivates our search for a flexible construction where the metric structure is tractable.

This paper develops the construction of new families of groups, not related to groups acting on rooted trees or their extensions, for which speed, entropy, return probability, isoperimetric profiles, Hilbert and some Banach space compression can all be computed explicitly. Before stating our result, let us recall some necessary definitions. Let Δ be a finitely generated group equipped with a generating set T, and let μ be a probability measure on Δ . Let X_1, X_2, \ldots be a sequence of independent and identically distributed random variables with distribution μ . Then $W_n = X_1 \cdots X_n$ is the random walk on Δ with step distribution μ . Its law is the *n*-fold convolution power μ^{*n} . Its speed function (rate of escape) is the expectation

$$L_{\mu}(n) = \mathbf{E}|W_n|_{\Delta} = \sum_{g \in \Delta} |g|_{\Delta} \, \mu^{*n}(g),$$

where $|\cdot|_{\Delta}$ is the word distance on the Cayley graph (Δ, T) . Its Shannon entropy is the quantity

$$H_{\mu}(n) = H(W_n) = -\sum_{x \in \Delta} \mu^{*n}(x) \log \mu^{*n}(x).$$

The pair (Δ, μ) has Liouville property if the Avez asymptotic entropy $h_{\mu} = \lim_{n \to \infty} \frac{H_{\mu}(n)}{n}$ is 0. By the work of Avez [Ave76], Derrienic [Der76], Kaimanovich-Vershik [KV83], this is equivalent to the Liouville property that all bounded μ -harmonic functions are constant.

The *return probabilities* of the random walk are

$$\mathbf{P}[W_{2n} = e] = \mu^{*(2n)}(e),$$

where e is the neutral element in Δ . The ℓ^{p} -isoperimetric profile is defined as

$$\Lambda_{p,\Delta,\mu}(v) = \inf\left\{\frac{\sum_{x,y\in\Delta} |f(xy) - f(x)|^p \,\mu(y)}{2\sum_{x\in\Delta} |f(x)|^p} : f \in \ell^p(G), \ 1 \le |\mathrm{supp}(f)| \le v\right\}.$$

The Kesten criterion for amenability [Kes59] states that Δ is non-amenable if and only if the return probability of a non-degenerate symmetric random walk on Δ decays exponentially. For amenable groups, the return probability and isoperimetric profiles provide quantitative information on the Følner sets in the groups. The asymptotic behavior of the return probability and the ℓ^2 -isoperimetric profile (spectral profile) essentially determine each other, as shown by Coulhon and Grigor'yan [CG97].

The *compression* of an embedding Ψ of Δ into a Banach space \mathfrak{X} is the function

$$\rho_{\Psi}(t) = \inf \left\{ \|\Psi(x) - \Psi(y)\|_{\mathfrak{X}} : \|x^{-1}y\|_{\Delta} \ge t \right\}.$$

The embedding is said to be uniform if $\rho_{\Psi}(t) > 0$ for all $t \ge 1$ and equivariant if Ψ is a 1-cocycle; see Section 6. The couple of functions (g_1, g_2) is an \mathfrak{X} compression gap of Δ if any 1-Lipschitz embedding $\varphi : \Delta \to \mathfrak{X}$ satisfies $\rho_{\varphi}(t) \le g_2(t)$ for all $t \ge 1$ and there exists a 1-Lipschitz embedding $\Psi : \Delta \to \mathfrak{X}$ such that $\rho_{\Psi}(t) \ge g_1(t)$ for all $t \ge 1$. The \mathfrak{X} -equivariant compression gap is defined in the same manner, restricting to equivariant embeddings. Let $L_p = L_p([0,1],m)$ be the standard Lebesgue space. By [NP11, Th. 9.1], when Δ is amenable, for all $p \geq 1$, (g_1, g_2) is an L_p -compression gap of Δ if and only if it is an equivariant L_p -compression gap of Δ .

Among these quantities, the compression gap is obviously independent of the choice of the measure μ , and up to multiplicative constants, it is invariant under quasi-isometry. The ℓ^p -isoperimetric profiles and return probability associated with symmetric probability measures of finite generating support are also known to be stable under quasi-isometry (see Pittet and Saloff-Coste [PSC00]), but the stability question regarding speed and entropy is open.

The groups we construct are diagonal products of lamplighter groups. Given a family of groups $\{\Gamma_s\}$ all generated by the union of two finite groups A and B, a factor of the diagonal product is the lamplighter group $\Gamma_s \wr \mathbb{Z} = (\bigoplus_{\mathbb{Z}} \Gamma_s) \rtimes \mathbb{Z}$ endowed with generating set consisting of the shift on \mathbb{Z} , a copy of the lamp subgroup A at position 0 and a copy of the lamp subgroup B at position $k_s \in \mathbb{Z}$. The diagonal product is the subgroup of the direct product generated by the diagonal generating set. Such a group is determined once we input a family of marked groups $\{\Gamma_s\}$, usually finite and satisfying some conditions, and a sequence (k_s) of "scaling factors" (see Section 2). When the groups $\{\Gamma_s\}$ are chosen among quotients of a residually finite group Γ that is generated by finite subgroups A and B, the choice of parameters heuristically permits one to interpolate between $\Gamma \wr \mathbb{Z}$ and the wreath product $(A \times B) \wr \mathbb{Z}$ with finite lamp group.

Constructions with expander sequences as input. The construction where the groups $\{\Gamma_s\}$ are taken to an expander sequence permits to show our main result. Denote $f(x) \simeq_C g(x)$ if $\frac{1}{C}g(x) \leq f(x) \leq Cg(x)$ for all $x \geq 1$. We write $f(x) \simeq g(x)$ and call f and g equivalent if there exists C with $f(x) \simeq_C g(x)$. Write $\log_*(x) = \log(x+1)$.

THEOREM 1.1. There exists a universal constant C > 1 such that the following holds. For any non-decreasing function $f: [1, \infty) \to [1, \infty)$ such that f(1) = 1 and x/f(x) is non-decreasing, there exists a group Δ of exponential volume growth equipped with a finite generating set T and a finitely-supported symmetric probability measure μ on Δ such that

- the speed and entropy functions satisfy $L_{\mu}(n) \simeq_C H_{\mu}(n) \simeq_C \sqrt{n} f(\sqrt{n});$
- the ℓ^p -isoperimetric profile satisfies $\Lambda_{p,\Delta,\mu}(v) \simeq_C \left(\frac{f(\log(v))}{\log v}\right)^p$ for any $p \in [1,2]$;
- the return probability satisfies $-\log\left(\mu^{*(2n)}(e)\right) \simeq_C w(n)$, where w(n) is defined implicitly by $n = \int_1^{w(n)} \left(\frac{s}{f(s)}\right)^2 ds$;
- $\left(\frac{1}{C\epsilon}\frac{n/f(n)}{\log_*^{1+\epsilon}(n/f(n))}, C \cdot 2^p \frac{n}{f(n)}\right)$ is an equivariant L_p -compression gap for Δ for any $p > 1, \epsilon > 0$.

When the function f is not asymptotically linear, i.e., $\lim_{x\to\infty} f(x)/x = 0$, the group Δ can be chosen elementary amenable with asymptotic dimension one. In this case (Δ, μ) has the Liouville property, and the equivariant L_p -compression gap is also valid for p = 1.

Since the constant C is universal, this result is new even when f is asymptotically linear.

The first statement asserts that any regular function between diffusive \sqrt{n} and linear n is the speed and entropy function of a random walk on a group. This statement on possible speed functions should be compared with known constraints on speed functions. From subadditivity, $L_{\mu}(n+m) \leq L_{\mu}(n) + L_{\mu}(m)$ for any convolution walk on a group Δ . By Lee-Peres [LP13], there is a universal constant c > 0 such that for any amenable group Δ equipped with a finite generating set T, for any symmetric probability measure μ on G, $L_{\mu}(n) \geq c\sqrt{p_*n}$, where $p_* = \min_{g \in T} \mu(g)$. On the other hand, we obtain that any function g(n) such that $\frac{g(n)}{\sqrt{n}}$ and $\frac{n}{g(n)}$ are non-decreasing is equivalent to a speed function. It improves on Amir-Virag [AV17] by the range between diffusive and $n^{\frac{3}{4}}$ for speed, and by the range close to linear for speed and entropy. The constant in Theorem 1.1 is universal, whereas the constants in [AV17] diverge when approaching linear behavior. Moreover, if only concerning speed and entropy, we can find such a group Δ in the class of 4-step solvable groups. This is the case when the groups Γ_s are lamplighters over finite *d*-dimensionnal lattices with $d \geq 3$; see Theorem 3.8 and Example 3.3.

As mentioned earlier, the third statement on return probability can be derived from the ℓ^2 -isoperimetric profile estimate in the second statement via the Coulhon-Grigor'yan theory; see Section 4.2. For $p \in [1, 2]$, any regular function between constant and n^{-p} is equivalent to $\Lambda_{p,\Delta,\mu} \circ \exp$ for some group Δ , and any regular function between $n^{\frac{1}{3}}$ and linear n is equivalent to $-\log \mu^{*n}(e)$. Again, this should be compared with known constraints for isoperimetric profile and return probability for groups with exponential volume growth. By Coulhon and Saloff-Coste [CSC93], for any symmetric probability measure μ on Δ , $-\log \mu^{*n}(e) \geq cn^{\frac{1}{3}}$ and $\Lambda_{p,\Delta,\mu} \circ \exp(x) \geq c'x^{-p}$, $p \in [1, 2]$, where the constants c, c' depend on the volume growth rate of (Δ, T) and $p_* = \min_{g \in T} \mu(g)$.

From the result on ℓ^1 -isoperimetric profile, we derive Corollary 4.7 that any sufficiently regular function above exponential is equivalent to a Følner functions. The result extends [OO13, Cor. 1.5]. It also answers [Ers10, Question 5] positively that there exists elementary amenable groups with arbitrarily fast growing Følner function, while simple random walk on it has the Liouville property. Groups of subexponential volume growth and arbitrarily large Følner function were first constructed by Erschler [Ers06]. When f is not asymptotically linear, the fourth statement asserts that any unbounded non-decreasing sublinear function h(n) is equivalent to the upper bound of an equivariant L_p -compression gap of width $\log^{1+\epsilon} h(n)$ of an amenable group. Recall that equivariant and non-equivariant compression are equivalent for amenable groups [NP11]. Regarding non-equivariant compression, it is an amenable analogue to the main result in [ADS09] and slightly improves on it as the width depends on the upper bound. It also provides other examples of amenable groups with poor compression, after [Aus11], [OO13] and [BE17]. It also follows easily that any function below \sqrt{x} is the upper bound of an equivariant L_p -compression gap of a non-amenable group, simply considering the direct product of an amenable solution with the free group on two generators. The equivariant compression of a non-amenable group is at most \sqrt{x} , and this bound is attained for free groups by [GK04].

In order to obtain the equivariant L_p -compression gap with upper bound x/f(x) bounded, we actually need to choose the family $\{\Gamma_s\}$ among quotients of a Lafforgue lattice with strong Property (T) [Laf08]. In this case, we also obtain an upper bound on the compression exponent of Δ into any uniformly convex normed space; see Corollary 6.2.

Solvable examples with finite dihedral groups as input. The large scale geometry of solvable groups of exponential growth has attracted much attention in the past decades. Algebraically, solvable groups, in particular polycyclic groups, are natural "small" amenable groups to be investigated after the Gromov polynomial growth theorem [Gro81]. Remarkable quasi-isometry rigidity results have been obtained by Farb and Mosher [FM98], [FM99] for solvable Baumslag-Solitar groups and by Eskin, Fisher and Whyte [EFW12], [EFW13] for lattices in the three dimensional solvable Lie group Sol and lamplighters. In general, solvable groups do not come naturally with actions on geometric spaces whose rich structure would facilitate an investigation of the geometry of the group. Many fundamental problems are open; see the survey [FM00].

When the input sequence $\{\Gamma_s\}$ consists of finite dihedral groups, our construction of diagonal products yields 3-step solvable groups. This collection of groups demonstrates the richness of large scale geometry in the class of solvable groups. In particular, their geometric properties differ significantly from polycyclic groups. On the other hand, these groups are closely related to lamplighters, and they might serve as candidates to be considered in the quasi-isometry classification program of solvable groups.

For random walks on these solvable groups, we have the following.

THEOREM 1.2. Let $\epsilon > 0$. There exists a constant C > 0. For any function f such that $\frac{f(n)}{\log^{1+\epsilon}(n)\sqrt{n}}$ and $\frac{n^{\frac{3}{4}}}{f(n)}$ are non-decreasing, there are a 3-step solvable group Δ and a finitely-supported symmetric probability measure such that the

SPEED OF RANDOM WALKS

Exponent of	$\mathbf{E}\left W_{n}\right _{\Delta}$	$H(W_n)$	$-\log\mu^{*2n}(e)$	$\Lambda_{p,\Delta,\mu} \circ \exp$	$\alpha_p^{\#}(\Delta)$
$\{\Gamma_s\}$ expanders	$\frac{1+\theta}{2+\theta}$	$\frac{1+\theta}{2+\theta}$	$\frac{1+\theta}{3+\theta}$	$\frac{-p}{1+\theta}$	$\frac{1}{1+\theta}$
$\{\Gamma_s\}$ dihedral	$\frac{1+3\theta}{2+4\theta}$	$\frac{1}{2}$	$\frac{1}{3}$	-p	$\max\left\{\frac{1}{1+\theta},\frac{2}{3}\right\}$

Figure 1. Exponents for sequences of parameters $k_s = 2^{2s}$ and $l_s = \text{diam}(\Gamma_s) \simeq 2^{2\theta s}$, where $\theta \in (0, \infty)$. The isoperimetric profile and compression exponent are all valid for $p \in [1, 2]$. The compression for expanders is valid for $p \in [1, \infty)$.

speed function is

$$\mathbf{E} |W_n|_{\Lambda} \simeq_C f(n)$$

and the entropy and return probability satisfy

$$\frac{1}{C}\sqrt{n} \le H(W_n^{\Delta}) \le C\sqrt{n}\log^2 n \text{ and } \frac{1}{C}n^{\frac{1}{3}} \le -\log\mu^{*n}(e) \le Cn^{\frac{1}{3}}\log^{\frac{4}{3}}n.$$

The factor $\log^{1+\epsilon} n$ is only technical. It follows from Proposition 3.11 that there is no gap isolating the diffusive behavior \sqrt{n} , but the analysis is simplified by this mild hypothesis. In contrast, it is known that for simple random walk on a polycyclic group of exponential growth, the speed is diffusive (i.e., equivalent to \sqrt{n}) and the return probability decays like $e^{-n^{1/3}}$. The exponents when $\{\Gamma_s\}$ are expanders or dihedral are given in Figure 1.

Questions on the relationship between the five quantities in Figure 1 are more easily asked at the level of exponents. The *exponent* of a function f is $\lim \frac{\log f(n)}{\log n}$ when the limit exists. For a compression gap of width less than any power, the lower exponent of the upper bound coincides with the definition of the \mathfrak{X} -compression exponent introduced in Guentner-Kaminker [GK04],

 $\alpha_{\mathfrak{X}}^*(G) = \sup \{ \alpha_{\mathfrak{X}}(\Psi) : \Psi \text{ is a Lipschitz map } G \to \mathfrak{X} \},\$

where the compression exponent $\alpha_{\mathfrak{X}}(\Psi)$ of the map $\Psi: \Delta \to \mathfrak{X}$ is defined as

$$\alpha_{\mathfrak{X}}(\Psi) = \sup \{ \alpha \ge 0 : \exists c > 0 \text{ s.t. } \rho_{\Psi}(t) \ge ct^{\alpha} \text{ for all } t \ge 1 \}.$$

When \mathfrak{X} is the classical Lebesgue space L_p , we write $\alpha_p^*(G)$ for the L_p -compression exponent. The equivariant compression exponent $\alpha_{\mathfrak{X}}^{\#}(G)$ is defined similarly, restricting to *G*-equivariant embeddings Ψ . When *G* is amenable, $\alpha_p^*(G) = \alpha_p^{\#}(G)$; see [NP11, Th. 1.6].

Theorems 1.1 and 1.2 together permit to solve a recent conjecture of Amir [Ami17].

COROLLARY 1.3 (Joint behavior of speed and entropy exponents). For any $\theta \in \left[\frac{1}{2}, 1\right]$ and $\gamma \in \left[\frac{1}{2}, 1\right]$ satisfying

$$\theta \leq \gamma \leq \frac{1}{2}(\theta+1),$$

there exist a finitely generated group G and a symmetric probability measure μ of finite support on G, such that the random walk on G with step distribution μ has entropy exponent θ and speed exponent γ .

The case where both exponents belong to $\begin{bmatrix} \frac{3}{4}, 1 \end{bmatrix}$ was treated by Amir [Ami17]. Proposition 3.17 gives a more precise statement regarding functions rather than exponents.

For the diagonal products constructed with finite dihedral groups that appear in Theorem 1.2, we estimate their L_p -compression exponents for $p \in [1, \infty)$; see Theorem 8.1. Explicit evaluation of compression exponents yields the following result. It answers [NP08, Question 7.6] positively within the class of finitely generated 3-step solvable groups. Moreover, with certain choices of parameters, such groups provide the first examples of amenable groups whose L_p -compression exponent, p > 2, is strictly larger than the Hilbert compression exponent.

THEOREM 1.4. For any $\frac{2}{3} \leq \alpha \leq 1$, there exists a 3-step solvable group Δ such that for any $p \in [1, 2]$,

$$\alpha_p^*(\Delta) = \alpha_p^{\#}(\Delta) = \alpha$$

Moreover, there exists a 3-step solvable group Δ_1 such that for all $p \in (2, \infty)$,

$$\alpha_p^*(\Delta_1) \ge \frac{3p-4}{4p-5} > \alpha_2^*(\Delta_1) = \frac{2}{3}$$

Both speed and compression exponents depend explicitly on the parameter sequences $(k_s), (l_s)$; see Figure 1 for some concrete choices of parameters. A key metric property of the diagonal product Δ that we rely on, in the estimates on speed and compression, is that placing the two types of generators k_s apart in $\Gamma_s \wr \mathbb{Z}$ essentially has the effect of rescaling the copies of Γ_s by a factor k_s . Moreover, under suitable assumptions, the metric in the diagonal product can be understood to behave similarly to a direct product.

In [Aus11, §5], Austin remarked that compression upper bounds from classical Poincaré inequalities and Markov type inequalities could be viewed as related to random walks (i.e., obstructions can be detected efficiently by suitable random walks); it would be interesting to find examples of finitely generated amenable groups for which obstructions genuinely unrelated to inequalities concerning random walks are needed. Austin conjectured that a group with a sequence of cubes $(\mathbb{Z}_m^n, \ell^{\infty})$ embedded would be a candidate for such type of obstructions. In some sense our construction of diagonal product Δ with finite dihedral groups as input realizes this idea. Because of the presence of scaled ℓ^{∞} -cubes of growing sizes in Δ , in the proof of Theorem 1.4 we apply deep results of Mendel and Naor [MN08] on metric cotype to estimate distortion of these embedded finite cubes. As a corollary, we obtain a new proof of the following result of Cornulier-Tessera [CT13].

COROLLARY 1.5 (Cornulier-Tessera [CT13]). There exist uncountably many pairwise non-quasi-isometric finitely generated 3-step solvable groups.

The original proof used asymptotic cones. Our method is completely different, using compression as quasi-isometry invariant. Theorem 1.4 and Corollary 1.5 do not hold for 2-step solvable groups. By Baumslag [Bau72], any finitely generated metabelian group embeds into a finitely presented metabelian group, so there are countably many classes of isomorphism of metabelian groups.

A formula relating the L_p -compression exponent of $H \wr \mathbb{Z}$ to H. The method we apply to estimate compression exponent of Δ in the dihedral case fundamentally differs from the case with expanders. Along the way, to better understand the compression of Δ , which is a diagonal product of groups $\Gamma_s \wr \mathbb{Z}$, it is instructive to first evaluate compression exponent of general wreath product $H \wr \mathbb{Z}$. This has been an object of intensive study; see [AGS06], [NP08], [ANP09], [Tes11], [CSV12]. We develop a novel approach and derive the following general formula which, in particular, extends the result in [NP11] on the L_p -compression exponent of $\mathbb{Z} \wr \mathbb{Z}$.

THEOREM 1.6. Let $p \in [1,2]$, and let H be a finitely generated infinite group. Then the equivariant L_p -compression exponent of $H \wr \mathbb{Z}$ is

$$\alpha_p^{\#}(H \wr \mathbb{Z}) = \min\left\{\frac{\alpha_p^{\#}(H)}{\alpha_p^{\#}(H) + \left(1 - \frac{1}{p}\right)}, \alpha_p^{\#}(H)\right\}.$$

Organization of the paper. The detailed description of diagonal products is given in Section 2. The construction of diagonal products is reminiscent of the piecewise automata group of Erschler [Ers06] and the groups of Kassabov-Pak [KP13], which permit us to obtain oscillating or "close to non-amenable" behaviors, but where more precise estimates are not known.

A technical assumption on the family $\{\Gamma_s\}$ and a list of examples satisfying it appears in Section 2.1. The key estimate relating the metric of a diagonal product to that of a direct product is established in Section 2.2 under the assumption that the sequence (k_s) is strictly doubling. In Section 2.3, we describe some metric spaces naturally embedded in the diagonal product. It will be used in Sections 6 and 8 on compression gaps.

Section 3 is devoted to the evaluation of speed and entropy functions of random walks on Δ . We first treat in Section 3.1 the case, including expanders, where the groups $\{\Gamma_s\}$ have uniform linear speed up to diameter. Theorem 3.8 gives the first point of Theorem 1.1. The case of dihedral groups is studied in

Section 3.2, proving two thirds of Theorem 1.2. Evaluation of speed and entropy of diagonal products relies on estimations of traverse time of the simple random walk on \mathbb{Z} , which are recalled in Appendix A. The joint prescription of speed and entropy of Corollary 1.3 is obtained in Section 3.3. Section 3 is not used further in the paper, and a reader not interested in speed or entropy can omit it.

Isoperimetric profiles and return probabilities are studied in Section 4. The second point of Theorem 1.1 is derived as Theorem 4.6. It is proved together with Corollary 4.7 regarding Følner functions in Section 4.1. The third point is derived as Theorem 4.8 in Section 4.2 using Coulhon-Grigor'yan theory. Dihedral groups are treated in 4.3 finishing the proof of Theorem 1.2. A reader interested mainly in compression can formally omit Section 4, though the test functions of Proposition 4.4 will be used in Section 6.

Obstructions for embeddings into Banach spaces are reviewed in Section 5. They are based on Poincaré inequalities on finite metric spaces embedded in the group. The classical spectral version stated in Section 5.1 will be used in Sections 6 and 7. Markov type inequalities of Section 5.2 will be used in Sections 7 and 8, and the Mendel-Naor metric cotype inequalities presented in Section 5.4 will be used in Section 8.

In Section 6, we consider diagonal products where $\{\Gamma_s\}$ are quotients of a Lafforgue lattice with strong Property (*T*). We first establish in Section 6.1 an upper bound on compression exponent valid in any uniformly convex Banach space, and then in Section 6.2 we derive the proof of the fourth part of Theorem 1.1, in the form of Theorem 6.11. This is done after three preliminary steps: first provide an upper bound when all quotients $\{\Gamma_s\}$ are finite, and secondly an upper bound when one of them is the whole group Γ . Thirdly an explicit 1-cocycle, related to isoperimetry, is constructed to get a lower bound.

Section 7 is devoted to Theorem 1.6. It requires none of previous sections except for Poincaré and Markov type inequalities of Section 5, but it uses several facts about stable random walks on lamplighters over a segment, gathered in Appendix C. It also serves as a warm-up for Section 8.

The compression of diagonal products with dihedral lamp groups is studied in Section 8. Theorem 1.4 is proved there, as well as some explicit bounds for L_p -compression p > 2, stated in Theorem 8.1. As before, we first derive some upper bound using metric cotype of Section 5.4 and Markov type inequalities of Section 5.2 and then describe an explicit embedding into $L_q \ q \ge 2$. Section 8 formally uses only Sections 2 and 5, but is best understood reading also Sections 6 and 7.

Finally we point out a few open questions in Section 9. Appendix B explains a natural approximation of regular functions by piecewise constant and linear functions. It is used repeatedly to prove Theorem 1.1 in Sections 3, 4 and 6.

Acknowledgments. We thank Goulnara Arzhantseva, Yves Cornulier and Mikaël de la Salle for interesting comments, Masato Mimura for a simplification in Example 2.3 and pointing out the LEF property of Fact 2.10, as well as an anonymous referee for useful comments.

2. The construction and metric structure

2.1. The construction with diagonal product. The wreath product of a group Γ with \mathbb{Z} is the group $\Gamma \wr \mathbb{Z} = \Gamma^{(\mathbb{Z})} \rtimes \mathbb{Z}$, where $\Gamma^{(\mathbb{Z})}$ is the set of functions $f:\mathbb{Z} \to \Gamma$ with finite support $(f) = \{j \in \mathbb{Z} : f(j) \neq e_{\Gamma}\}$. An element is represented by a pair (f, i). We refer to f as the lamp configuration and i as the position of the cursor. The product rule is

$$(f,i)(g,j) = (f(\cdot)g(\cdot - i), i + j).$$

The neutral element is denoted as (e, 0), where support(e) is the empty set. For $j \in \mathbb{Z}$ and $\gamma \in \Gamma$, we denote by $\gamma \delta_j$ the function taking value γ at j and e_{Γ} elsewhere.

Let $A = \{a_1, \ldots, a_{|A|}\}$ and $B = \{b_1, \ldots, b_{|B|}\}$ be two finite groups. Let $\{\Gamma_s\}$ be a sequence of groups such that each Γ_s is marked with a generating set of the form $A(s) \cup B(s)$, where A(s) and B(s) are finite subgroups of Γ_s isomorphic respectively to A and B. We fix the isomorphic identification and write $A(s) = \{a_1(s), \ldots, a_{|A|}(s)\}$ and similarly for B(s).

Fix a sequence $(k_s)_{s\geq 0}$ of strictly increasing integers. Take the wreath product $\Delta_s = \Gamma_s \wr \mathbb{Z}$, and mark it with generating tuple \mathcal{T}_s

$$\mathcal{T}_s = \big(\tau(s), \alpha_1(s), \dots, \alpha_{|A|}(s), \beta_1(s), \dots, \beta_{|B|}(s)\big),$$

where $\tau(s) = (e, +1)$ and

$$\alpha_i(s) = (a_i(s)\delta_0, 0), 1 \le i \le |A|, \ \beta_i(s) = (b_i(s)\delta_{k_s}, 0), 1 \le i \le |B|.$$

With slight abuse of notation, we use the symbol Δ_s to denote the marked group. Alternatively, the marked group Δ_s can be viewed as the projection

$$\boldsymbol{\pi}_s: \mathbf{G} = \mathbb{Z} * A * B \to \Gamma_s \wr \mathbb{Z},$$

where \mathbf{G} is the free product of

$$\langle \boldsymbol{\tau} \rangle = \mathbb{Z}, \langle \boldsymbol{\alpha}_i, 1 \leq i \leq |A| \rangle \simeq A \text{ and } \langle \boldsymbol{\beta}_i, 1 \leq i \leq |B| \rangle \simeq B$$

and the projection sends $\boldsymbol{\tau}$ to $\boldsymbol{\tau}(s)$, $\boldsymbol{\alpha}_i$ to $\boldsymbol{\alpha}_i(s)$ and $\boldsymbol{\beta}_i$ to $\beta_i(s)$.

The diagonal product Δ of the, possibly finite, sequence of marked groups $\{\Delta_s\}$ is the quotient group $\mathbf{G}/\bigcap_s \ker(\boldsymbol{\pi}_s)$, with the projection map $\boldsymbol{\pi}: \mathbf{G} \to \Delta$. It is marked with generating tuple $\mathcal{T} = (\tau, \alpha_1, \ldots, \alpha_{|A|}, \beta_1, \ldots, \beta_{|B|})$. A word in \mathcal{T} represents e_{Δ} if and only if the same word in \mathcal{T}_s represents e_{Δ_s} for each s. We use $\boldsymbol{\pi}_s: \Delta \to \Delta_s$ to denote the projection from Δ to the component Δ_s . The group Δ is completely determined once the family of marked groups $\{\Gamma_s\}$ and the sequence of distances (k_s) are given. An element g of Δ is completely determined by the family of projections $\pi_s(g) = (f_s, i_s)$, and one immediately checks that the projection onto \mathbb{Z} is independent of s. Therefore we write $((f_s), i)$ for a typical element of Δ , where $f_s \in \Gamma_s^{(\mathbb{Z})}$ and $i \in \mathbb{Z}$.

ASSUMPTION 2.1. Throughout the paper, we assume the following:

- $k_0 = 0$ and $\Gamma_0 = A(0) \times B(0) \simeq A \times B$.
- We assume that

$$\Gamma_s/[A(s), B(s)]^{\Gamma_s} \simeq A(s) \times B(s) \simeq A \times B.$$

Here $[A(s), B(s)]^{\Gamma_s}$ is the normal closure of the subgroup generated by the commutators $[a_i(s), b_j(s)]$. We call the quotient group $\Gamma_s/[A(s), B(s)]^{\Gamma_s}$ the relative abelianization of Γ_s .

The first assumption is mainly for convenience of notation. It follows easily from Lemma 2.7 below that the marked group $(A \times B) \wr \mathbb{Z}$ with usual generating set $(k_0 = 0)$ is a quotient of Δ as soon as (k_s) is unbounded.

The second assumption is non-trivial and restrictive. It requires that the relative abelianization, which is always a quotient of $A \times B$, is in fact isomorphic to $A \times B$. As we will see below, we can find interesting families of groups satisfying Assumption 2.1.

Notation 2.2. Take a family $\{\Gamma_s\}$ of quotients of an infinite group Γ , and parametrize the group Γ_s by its diameter $l_s = \operatorname{diam}(\Gamma_s)$ with respect to the generating set $A(s) \cup B(s)$. Taking the value $l_s = \infty$ corresponds to the choice $\Gamma_s = \Gamma$, otherwise $l_s < \infty$, Γ_s is a finite quotient group of Γ . We say that the sequences $(k_s), (l_s)$ parametrize the diagonal product Δ . Formally (k_s) can take the value ∞ . We use the convention that if $k_s = \infty$, then Δ_s is the trivial group $\Delta_s = \{e_{\Delta_s}\}$.

In this paper we will take a group Γ and a family $\{\Gamma_s\}$ of quotients of Γ from the following list of specific examples.

Example 2.3. The groups $\{\Gamma_s\}$ can be taken to form a family of expanders. For example, we obtain the following sequence of finite groups from the Lafforgue super expanders. By Lafforgue [Laf08], for any local field F, the group SL(3, F) has Property (T) in any uniformly convex Banach space \mathfrak{X} . A fortiori taking $F = \mathbb{F}_p[[X^{-1}]]$, this is also the case for the non-uniform lattice $\Lambda = SL(3, \mathbb{F}_p[X])$, generated by the union of the two following finite subgroups:

$$A = \left\langle \left(\begin{array}{rrrr} 1 & 1 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{array} \right), \left(\begin{array}{rrrr} 1 & X & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{array} \right) \right\rangle \simeq \mathbb{F}_p^2,$$

$$B = \left\langle \left(\begin{array}{rrr} 0 & 1 & 0 \\ 0 & 0 & 1 \\ 1 & 0 & 0 \end{array} \right) \right\rangle \simeq \mathbb{Z}/3\mathbb{Z}.$$

There also exist positive constants c, c_1 such that the "congruence subgroups"

$$\Lambda_m = \mathrm{SL}(3, \mathbb{F}_p[X]/(X^m - 1))$$

satisfy $cm - c_1 \leq \log |\Lambda_m| \leq cm + c_1$ for all $m \geq 1$.

To make sure the second part of Assumption 2.1 holds, we take Γ (resp. Γ_m) to be the diagonal product of Λ (resp. Λ_m) with $A \times B$. Since Λ is a quotient group of Γ of index at most |A| |B|, it follows from the hereditary properties (see [BdlHV08, §1.7]) that Γ has Property (*T*) in any uniformly convex Banach space \mathfrak{X} . Since $\{\Gamma_m\}$ is a sequence of finite quotient groups of Γ , by [Laf08, Prop. 5.2], there exists a constant $\delta(\Gamma, \mathfrak{X}) > 0$ such that for any function $f: \Gamma_m \to \mathfrak{X}, m \in \mathbb{N}$,

(1)
$$\frac{\frac{1}{|\Gamma_m|^2} \sum_{x,y \in \Gamma_m} \|f(x) - f(y)\|_{\mathfrak{X}}^2}{\leq \frac{1}{\delta(\Gamma,\mathfrak{X})} \frac{1}{|\Gamma_m|} \sum_{x \in \Gamma_m} \sum_{u \in A(m) \cup B(m)} \|f(x) - f(xu)\|_{\mathfrak{X}}^2}.$$

In particular, $\{(\Gamma_m, A(m) \cup B(m))\}$ form a family of expanders in ℓ^2 with spectral gap uniformly bounded from below by $\delta(\Gamma, \ell^2)$.

We will refer to this family $\{\Gamma_m\}$ as the Lafforgue super expanders. Each Γ_m is marked with generating set $A(m) \cup B(m)$. Note that by construction,

$$|cm - c_1| \le \log |\Lambda_m| \le \log |\Gamma_m| \le cm + c_1 + \log(3p^2).$$

From the inequality (1), by [HLW06, Th. 13.8] there exists constant $c_2 > 0$ depending only on r and $\delta(\Gamma, \ell^2)$ such that the ℓ^2 -distortion satisfies

$$c_2 \log |\Gamma_m| \le c_{\ell^2}(\Gamma_m) \le \operatorname{diam}(\Gamma_m) \le r \log |\Gamma_m|,$$

and by [ADS09, Cor. 3.5], there exists $c_3 > 0$ depending only on r and $\delta(\Gamma, \mathfrak{X})$ such that

$$c_3 \log |\Gamma_m| \le c_{\mathfrak{X}}(\Gamma_m).$$

See Section 5 for the definition of distortion. Lafforgue super expanders are a crucial tool to study compression in arbitrary uniformly convex Banach spaces.

In most statements of this paper, it is sufficient to use the classical Property (T) rather than its strengthening in uniformly convex Banach spaces. Therefore we can also take $\Lambda = \text{EL}(3, \mathbb{F}_p(X, Y))$, which has Property (T) by Ershov-Jaikin-Zapirain [EJZ10] generated by the union of its finite subgroups

$$A = \left\langle \left(\begin{array}{ccc} 1 & a & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{array} \right), a \in \{1, X, Y\} \right\rangle \simeq \mathbb{F}_{p^{3}}^{3}$$
$$B = \left\langle \left(\begin{array}{ccc} 0 & 1 & 0 \\ 0 & 0 & 1 \\ 1 & 0 & 0 \end{array} \right) \right\rangle \simeq \mathbb{Z}/3\mathbb{Z},$$

and with "congruence subgroups" EL $(3, \operatorname{Mat}_{m \times m}(\mathbb{F}_p)) \simeq \operatorname{SL}(3m, \mathbb{F}_p)$ the subgroup generated by the matrices $e_{i,j}(a)$ for $a \in \operatorname{Mat}_{m \times m}(\mathbb{F}_p)$ and $i \neq j$ in $\{1, 2, 3\}$, which are identity plus the matrix with only non-zero entry a in position i, j.

Examples 2.4. Some choices of families $\{\Gamma_s\}$ permit us to obtain diagonal products in the class of solvable groups.

(1) An obvious choice is to take $\Gamma_s = D_{l_s}$ as a finite dihedral group of size $2l_s$ generated by two involutions with $A = \mathbb{Z}/2\mathbb{Z}$, $B = \mathbb{Z}/2\mathbb{Z}$ and $\Gamma = D_{\infty}$. Then the diagonal product Δ is 3-step solvable.

(2) Another possibility is to take $\Gamma = \mathbb{Z}/2\mathbb{Z} \wr D_{\infty}^{k}$ as the lamplighter on an ordinary k-dimensionnal lattice. We can take $A = \mathbb{Z}/2\mathbb{Z} \wr (\mathbb{Z}/2\mathbb{Z})^{k}$ and $B = \mathbb{Z}/2\mathbb{Z} \wr (\mathbb{Z}/2\mathbb{Z})^{k}$ in the obvious manner. The quotients $\Gamma_{s} = \mathbb{Z}/2\mathbb{Z} \wr D_{m_{s}}^{k}$ are lamplighter over an ordinary discrete k-dimensionnal torus. The diagonal product Δ is 4-step solvable.

To check that this example satisfies Assumption 2.1, we denote $\pi : D_m \to \mathbb{Z}/2\mathbb{Z} \times \mathbb{Z}/2\mathbb{Z}$ as the abelianization $\pi(x) = a^{\epsilon(x)}b^{\eta(x)}$. We set

$$\pi_A : \mathbb{Z}/2\mathbb{Z} \wr D_m^k \to \mathbb{Z}/2\mathbb{Z} \wr (\mathbb{Z}/2\mathbb{Z})^k \simeq A$$
$$(f, x) \mapsto (f_A, a_1^{\epsilon_1(x)} \cdots a_k^{\epsilon_k(x)}),$$

where

$$f_A(a_1^{\epsilon_1}\cdots a_k^{\epsilon_k}) = \prod_{x:\ \epsilon_i(x)=\epsilon_i, \forall i} f(x).$$

It is clear that $[A, B]^{\Gamma_m} \subset \operatorname{Ker} \pi_A$ and that we can proceed similarly for B. Finally, one can check that $\pi_A \times \pi_B$ projects onto $A \times B$.

Remark 2.5. The requirement that Γ is a quotient of a free product of *finite* groups is crucial, but we can generalize our construction to more than two finite factors, positioned on an arithmetic progression of common difference k_s . This is straightforward for metric estimate, isoperimetry and compression, but it raises some technical questions regarding random walks, as the walk inherited on the lamps is not simple anymore. For simplicity, and by lack of relevant examples, we avoid this generality.

2.2. Description of the metric. We describe the metric structure of the Cayley graph (Δ, \mathcal{T}) .

2.2.1. Local coincidence in relative abelianizations. We will be able to estimate the metric in our diagonal products for sequences (k_s) growing exponentially because different factors are largely independent. But the diagonal product differs from the usual direct product. For instance, the lamp configurations of relative abelianizations behave jointly.

Notation 2.6 (Projection maps under Assumption 2.1). Let $\theta_s : \Gamma_s \to \Gamma_s/[A_s, B_s] \simeq A(s) \times B(s)$ denote the projection to the relative abelianization. The projection map extends point-wise to the lamp configuration function $f_s : \mathbb{Z} \to \Gamma_s$,

$$\left(\theta_s\left(f_s\right)\right)(x) = \theta_s\left(f_s(x)\right).$$

We call $\theta_s(f_s)$ the lamp configuration of the relative abelianization.

Let θ_s^A and θ_s^B denote the compositions of θ_s with the projection to A(s) and B(s) respectively. Then we have a decomposition of $\theta_s(f_s)$ into a commutative product of functions

$$\theta_s\left(f_s\right) = \theta_s^A\left(f_s\right)\theta_s^B\left(f_s\right).$$

For any element $g = ((f_s), i)$ in the diagonal product Δ , all the relative abelianization lamp configurations are determined by the first one.

LEMMA 2.7. Let $g = ((f_s), i)$ be an element in the diagonal product Δ . Then under Assumption 2.1, any one abelianized function $\theta_s(f_s)$ is determined by $\theta_0(f_0) = f_0$ and vice-versa. More precisely, for any, s

$$\theta_s^A(f_s(x)) = \theta_0^A(f_0(x)) \text{ and } \theta_s^B(f_s(x)) = \theta_0^B(f_0(x-k_s)).$$

Proof. We proceed by induction on the word length of $g = ((f_s), i)$. Multiplying by a generator α_j , $\theta_s^A(f_s(x))$ is modified exactly at x = i, which also accordingly modifies $\theta_0^A(f_0(i))$. Multiplying by a generator β_j , $\theta_s^B(f_s(x))$ is modified exactly at $x = i + k_s$, which also accordingly modifies $\theta_0^B(f_0(i))$. \Box

2.2.2. Local finiteness of the diagonal product. Denote by $\pi_{\mathbb{Z}} : \mathbf{G} = \mathbb{Z} * A * B \to \mathbb{Z}$ the projection on the first factor.

Definition 2.8. The range $\operatorname{Range}(w)$ of a representative word w of an element in **G** is the collection of all $\pi_{\mathbb{Z}}(w')$, where w' is a prefix of w. It is a finite subinterval of \mathbb{Z} , the set of sites visited by the cursor. We will also denote $\operatorname{Range}(w)$ as its diameter.

For an element g in Δ or in Δ_s , we denote by $\operatorname{Range}(g)$ the diameter of a minimal range interval of a word in **G** representing it and by $s_0(g)$ the maximal integer with $k_{s_0(g)} \leq \operatorname{Range}(g)$.

Denote by $\Delta_{\leq s} = \mathbf{G} / \bigcap_{0 \leq s' \leq s} \operatorname{Ker}(\pi_{s'})$ the diagonal product of the s + 1 first factors and by $\pi_{\leq s} : \Delta \to \Delta_{\leq s}$ the natural projection.

FACT 2.9. Under Assumption 2.1, for any $g \in \mathbf{G}$, the evaluation $\pi(g)$ in Δ is determined by $\pi_{\leq s_0(g)}(g)$.

Proof. If $k_s > \text{Range}(g)$, then f_s takes values in the generating set $A(s) \cup B(s)$. By Assumption 2.1, f_s is determined by $\theta_s(f_s)$ and so by $\theta_0(f_0) = f_0$ using Lemma 2.7. So all $\pi_s(g)$ for $s > s_0(g)$ can be recovered from $\pi_0(g)$. \Box

In particular, as the range is bounded above by the length, the above argument shows that the sequence of marked groups $(\Delta_s, \mathcal{T}_s)$ converges to $(\Delta_0, \mathcal{T}_0)$ and the sequence $(\Delta_{\leq s}, \mathcal{T}_{\leq s})$ to (Δ, \mathcal{T}) in the Chabauty topology.

We also record the following.

FACT 2.10. Assume (k_s) is unbounded.

- (i) If all the groups in the family $\{\Gamma_s\}$ are elementary amenable (e.g., finite), then the diagonal product Δ is also elementary amenable.
- (ii) If all the groups in the family $\{\Gamma_s\}$ are locally embeddable into finite groups (e.g., finite), then the diagonal product Δ is also locally embeddable into finite groups.
- (iii) If all the groups in the family $\{\Gamma_s\}$ are finite, then the diagonal product Δ has asymptotic dimension one.

Proof. We observe that when (k_s) is unbounded, ker $\pi_{\mathbb{Z}}$ is locally included in a finite direct product of copies of the groups $\{\Gamma_s\}$. If G is the subgroup generated by elements g_1, \ldots, g_k in Ker $\pi_{\mathbb{Z}}$, then by the previous fact, G is isomorphic to $\pi_{\leq S}(G)$ where $S = \max_{1 \leq i \leq k} s_0(g_i)$. Moreover, in each copy Δ_s with $s \leq S$, each element g_i is described by a function $f_{i,s} : \mathbb{Z} \to \Gamma_s$ with finite support, and the group law induced is point-wise multiplication. This proves (i) and also (iii) as the asymptotic dimension of an extension is less than that of quotient plus that of kernel by [BD06].

By Fact 2.9, the ball of radius R in the Cayley graph of Δ depends only on the R-balls in Δ_s for s below the maximal s_0 with $k_{s_0} \leq R$. Each R-ball in a group Γ_s coincides with the R-ball of a finite group H_s , so the R-ball in Δ_s coincides with the R-ball in $H_s \wr \mathbb{Z}/(4R+1)\mathbb{Z}$. A fortiori, the R-ball in Δ coincides with the R-ball of the diagonal product of these $s_0 + 1$ finite groups. This proves (ii).

2.2.3. Metric in one copy Δ_s . In order to estimate the metric in the diagonal product, the sequence (k_s) must grow exponentially fast. Therefore we make the

ASSUMPTION 2.11. Throughout the paper, we assume that the sequence (k_s) is a sequence of strictly increasing even numbers such that $k_{s+1} > 2k_s$ for all s.

Exponential growth of the sequence (k_s) is crucial for the second part of Lemma 2.13 below. We choose the factor 2 for simplicity. It could be replaced by any $m_0 > 1$, which would only modify our estimates by some multiplicative constants.

Definition 2.12. For $j \in \mathbb{Z}$, let $I_j^s = [\frac{jk_s}{2}, \frac{(j+1)k_s}{2})$. We call essential contribution of the function $f_s : \mathbb{Z} \to \Gamma_s$ the quantity

$$E_s(f_s) = \sum_{\left\{j: \ I_j^s \cap \operatorname{Range}(f_s, i) \neq \emptyset\right\}} k_s \max_{x \in I_j^s} (|f_s(x)|_{\Gamma_s} - 1)_+,$$

where $(x)_{+} = \max\{x, 0\}$. In words, we partition the range into intervals of width $\frac{k_s}{2}$. Each of these intervals essentially contributes as k_s times the maximal distance $|f_s(x)|_{\Gamma_s} - 1$.

The essential contribution measures the contribution of the terms $f_s(x)$ of Γ_s -length more than 2 to the length of an element (f_s, i) of Δ_s . The range will take care of the contribution of terms of length less than 1.

LEMMA 2.13. Let (f_s, i) belong to Δ_s . Then

$$\max\left\{\frac{1}{8}E_s(f_s), \operatorname{Range}(f_s, i)\right\} \le |(f_s, i)|_{\Delta_s} \le 9\left(E_s(f_s) + \operatorname{Range}(f_s, i)\right).$$

Let Δ be the diagonal product of $\{\Delta_s\}$. Under Assumptions 2.1 and 2.11 and if, moreover, $\theta_s(f_s) = e$, then there is a word $\omega(f_s, i) \in \mathbf{G}$ of length less than the above upper bound such that

$$\pi_{s'}(\omega(f_s, i)) = \begin{cases} (f_s, i) & \text{if } s' = s, \\ (0, i) & \text{if } s' \neq s. \end{cases}$$

Proof. The lower bound by the range is obvious.

Let [x] denote the integer part of x. To get the lower bound by essential contribution, notice that in order to write $f_s(x)$, the cursor has to traverse at least $[|f_s(x)|_{\Gamma_s}/2]$ times between positions in $x - k_s$ and x because a minimal representative word alternates elements of A(s) and B(s). So x contributes at least $k_s[|f_s(x)|_{\Gamma_s}/2]$ to the length of a representative word w of (f_s, i) .

If the intervals $[x - k_s, x]$ and $[x' - k_s, x']$ are disjoint, the contributions of x and x' must add up. Let x_j^s achieve the maximum of $|f_s(x)|_{\Gamma_s}$ on the interval I_j^s . The separation condition is satisfied for a family of x_j^s with the same congruence of j modulo 4. Therefore

$$|w| \ge \max_{0 \le a \le 3} \left(\sum_{j \equiv a \mod 4} k_s \left[|f_s(x)|_{\Gamma_s}/2 \right] \right)$$
$$\ge \frac{1}{4} \sum_{j \in \mathbb{Z}} k_s \max_{I_j^s} \left[|f_s(x)|_{\Gamma_s}/2 \right] \ge \frac{1}{8} E_s(f_s).$$

To get the upper bound, the generic strategy is the following. We partition the convex envelope of $\operatorname{supp}(f_s) \cup \{0, i\}$, of length less than $\operatorname{Range}(f_s, i) + k_s$, into its intersections with the intervals I_j^s for $j \in \mathbb{Z}$. The elements of the partition are still denoted I_j^s . Let J_{\max} (resp. J_{\min}) denote the maximum (resp. minimum) index of this partition, and let $w(f_s(x))$ be a fixed minimal representative word for $f_s(x)$. We produce a representative word for $(f_s, i)^{-1}$ by the following strategy.

First apply a power $p_1 \leq \text{Range}(f_s, i) + k_s$ of the shift τ to move the cursor from *i* to the rightmost point of the interval $I_{J_{\text{max}}}^s$. Then for each integer *j* from J_{max} to J_{min} , produce a word ω_j that, taking the cursor from the rightmost point of I_j^s , erases all the words $w(f_s(x))$ for $x \in I_j^s$ and eventually leaves the cursor at the rightmost point of I_{j-1}^s .

The description of ω_j is the following. The first run takes the cursor to $j\frac{k_s}{2} - k_s$ and then back, so that all the last letters of $f_s(x)$ for $x \in I_j^s$ can be deleted. More precisely, while the cursor is in $[j\frac{k_s}{2}, (j+1)\frac{k_s}{2}) = I_j^s$, multiplying by $\alpha_l(s)$ at appropriate locations removes the last letter to those words $w(f_s(x))$ ending with $a_l(s)$ and while the cursor is in $[j\frac{k_s}{2} - k_s, (j+1)\frac{k_s}{2} - k_s) = I_{j-2}^s$ multiplying by $\beta_l(s)$ at appropriate locations removes the last letter to those words $w(f_s(x))$ ending with $b_l(s)$. One run has length $3k_s$ and cancels one letter in each of the words. The number of runs necessary to erase completely all the words is $\max_{x \in I_j^s} |f_s(x)|_{\Gamma_s}$. A last run takes the cursor from the rightmost point of I_j^s to the rightmost point of I_{j-1}^s , except for $j = J_{\min}$. Thus $|\omega_j| \leq 3k_s \max_{x \in I_j^s} |f_s(x)|_{\Gamma_s} + \max(I_j^s) - \max(I_{j-1}^s)$ without last term for $j = J_{\min}$.

Finally apply a power $p_2 \leq \text{Range}(f_s, i) + k_s$ of the shift τ to move the cursor from $\max(I_{J_{\min}}^s)$ to 0. All in all,

$$\omega(f_s,i) = (\tau^{p_1}\omega_{J_{\max}}\cdots\omega_{J_{\min}}\tau^{p_2})^{-1}$$

is a representative word of (f_s, i) with length

$$\begin{aligned} |\omega(f_s, i)| &\leq \sum_{j \in \mathbb{Z}} 3k_s \max_{I_j^s} |f_s(x)|_{\Gamma_s} + \sum_{j=J_{\min}}^{J_{\max}} \max(I_j^s) - \max(I_{j-1}^s) \\ &+ 2 \operatorname{Range}(f_s, i) + 2k_s \leq 3 \sum_{j \in \mathbb{Z}} k_s \max_{I_j^s} |f_s(x)|_{\Gamma_s} + 3 \operatorname{Range}(f_s, i) + 2k_s \end{aligned}$$

Now the number of indices j such that I_j^s intersects the range of (f_s, i) is less than $[2 \operatorname{Range}(f_s, i)/k_s] + 1$. Therefore

$$\sum_{j \in \mathbb{Z}} k_s \max_{I_j^s} |f_s(x)|_{\Gamma_s} \le E_s(f_s) + 2 \operatorname{Range}(f_s, i) + k_s$$

Generically, $E_s(f_s) \neq 0$ and then $k_s \leq E_s(f_s)$, and the two previous inequalities give the upper bound. In the non-generic case when $E_s(f_s) = 0$, then $|f_s(x)|_{\Gamma_s} \leq 1$ for all x and an obvious bound is $|(f_s, i)|_{\Delta_s} \leq 3 \operatorname{Range}(f_s, i)$.

To get the second part of the lemma, observe first that if $E_s(f_s) = 0$ and $\theta_s(f_s) = \mathbf{e}$, then $(f_s, i) = (\mathbf{e}, i)$ is just a translation, and the conclusion holds trivially. In the generic case, we have to check that for each sub-word ω_j in the above procedure, the lamp function $f_{s'}^{\omega_j}$ of $\pi_{s'}(\omega_j)$ is trivial.

First assume s' > s. By Assumption 2.11, the cursor moves in the interval $I = [j\frac{k_s}{2} - k_s, (j+1)\frac{k_s}{2}]$ of length $\frac{3}{2}k_s < 2k_s < k_{s'}$. In this condition, multiplying by α_l contributes to $f_{s'}^{\omega_j}(x)$ by $a_l(s')$ at positions $x \in I$ and multiplying by β_l contributes to $f_{s'}^{\omega_j}(x)$ by $b_l(s')$ at positions $x \in I + k_{s'}$. These intervals are disjoint, so $f_{s'}^{\omega_j}$ takes values in the generating set of Γ_s . Thus $f_{s'}^{\omega_j}(x) = \theta_{s'}(f_{s'}^{\omega_j})(x) = \theta_{s'}^A(f_{s'}^{\omega_j})(x) \theta_{s'}^B(f_{s'}^{\omega_j})(x) = e_{\Gamma_s}$, because $\theta_{s'}^A(f_{s'}^{\omega_j})(x) = \theta_0^A(f_0^{\omega_j})(x) = e_0^B(f_0^{\omega_j})(x + k_s - k_{s'}) = e$ using Lemma 2.7 and our hypothesis.

Now assume s' < s. The generators $a_l(s')$ were applied only when the cursor i was in I_j^s . On the other hand, the generators β_l were applied only at points $i + k_s \in I_j^s$, that is, when the cursor i was in I_{j-2}^s . Then, as $k_{s'} \in [0, \frac{k_s}{2})$, by Assumption 2.11, the element $b_l(s')$ is applied only at locations $x = i + k_{s'} \in I_{j-2}^s + [0, \frac{k_s}{2}) \subset I_{j-2}^s \cup I_{j-1}^s$. As the latter set is disjoint from I_j^s , the function $f_{s'}^{\omega_j}$ again takes values in the generating set of Γ_s . We conclude as above. \Box

2.2.4. Description of the metric in the diagonal product. Now we are ready to estimate metric in the diagonal product Δ .

PROPOSITION 2.14. Suppose the sequence $\{\Gamma_s\}$ of marked groups satisfies Assumption 2.1 and the sequence (k_s) of integers satisfies Assumption 2.11. For any element $g = ((f_s), i)$ in the diagonal product Δ , the word distance in (Δ, \mathcal{T}) satisfies

$$\max_{s \ge 0} |(f_s, i)|_{\Delta_s} \le |g|_{\Delta} \le 500 \sum_{0 \le s \le s_0(g)} |(f_s, i)|_{\Delta_s}.$$

Proof. The first inequality holds because Δ_s is a marked quotient of Δ .

For the second inequality, let $\omega(f_0, i)$ be a minimal representative word of (f_0, i) . This is realized when the cursor moves across the range and at each site switches appropriately the A and B lamps. In particular, the word $\omega(f_0, i)$ has

length $|(f_0, i)|_{\Delta_0} \leq 3 \operatorname{Range}(f_0, i)$. It represents an element $((h_s), i)$ in Δ with

 $h_0 = f_0 \text{ and } |h_s(x)|_{\Gamma_s} \le 2 \text{ for all } x \text{ and } s.$ Then $g\omega(f_0, i)^{-1} = ((f'_s), 0) \text{ with } f'_s = f_s h_s^{-1} \text{ for all } s.$ In particular, $f'_0 = e$, thus $\theta_s(f'_s) = e$ for all s by Lemma 2.7. Lemma 2.13 applies and furnishes words $\omega(f'_s, 0)$ such that

$$g\omega(f_0,i)^{-1}\omega(f'_1,0)^{-1}\cdots\omega(f'_{s_0(g)},0)^{-1}=e.$$

This is true in $\Delta_{\leq s_0(g)}$ and hence in Δ by Fact 2.9. This shows that

$$|g|_{\Delta} \leq |\omega(f_0, i)| + \sum_{1 \leq s \leq s_0(g)} |\omega(f'_s, 0)|.$$

We claim that $\operatorname{support}(f'_s) \subset \operatorname{support}(f_s)$. This implies that $\operatorname{Range}(f'_s, 0) \leq$ $2 \operatorname{Range}(f_s, i)$. Moreover, $E_s(f'_s) \leq 3E_s(f_s)$ because $|f'_s(x)|_{\Gamma_s} \leq |f_s(x)|_{\Gamma_s} + 2$ for all x and s. Therefore, using Lemma 2.13, we can conclude that

$$\begin{aligned} \left|\omega(f'_s, 0)\right| &\leq 9(E_s(f'_s) + \operatorname{Range}(f'_s, 0)) \leq 27(E_s(f_s) + \operatorname{Range}(f_s, i)) \\ &\leq 54 \max(E_s(f_s), \operatorname{Range}(f_s, i)) \leq 432 \left|(f_s, i)\right|_{\Delta_s}. \end{aligned}$$

The claim follows from Lemma 2.7. Indeed, if $f_s(x) = e_{\Gamma_s}$, then, in particular,

$$e_A = \theta_s^A(f_s(x)) = \theta_0^A(f_0(x)) = \theta_0^A(h_0(x)) = \theta_s^A(h_s(x))$$

and

$$e_B = \theta_s^B(f_s(x)) = \theta_0^B(f_0(x - k_s)) = \theta_0^B(h_0(x - k_s)) = \theta_s^B(h_s(x))$$

so $\theta_s(h_s(x)) = e_{A \times B}$. As $|h_s(x)|_{\Gamma_s} \leq 2$, this implies that $h_s(x) = e_{\Gamma_s}$. Therefore $f'_s(x) = e_{\Gamma_s}$ as well.

2.3. Metric spaces embedded in Δ . In this section, we gather some elementary facts about embeddings of some metric spaces into the diagonal product Δ . It will be used to obtain upper bounds on compression in Sections 6 and 8.

2.3.1. Embedding a lamp group Γ_s .

FACT 2.15. Each group Γ_s embeds homothetically in the diagonal product Δ with ratio $k_s + 1$; i.e., there is group homomorphism $\vartheta_s : \Gamma_s \to \Delta$ satisfying

$$\begin{aligned} |\vartheta_s(\gamma)|_{\Delta} &= (k_s+1) \, |\gamma|_{\Gamma_s} \, . \end{aligned}$$
Proof. Let $w = a_{i_1}(s)b_{i_1}(s)\cdots a_{i_n}(s)b_{i_n}(s)$ belong to $A(s)*B(s)$. Set
 $\vartheta_s(w) = \tau^{\frac{k_s}{2}} \alpha_{i_1} \tau^{-k_s} \beta_{i_1} \tau^{k_s} \cdots \alpha_{i_n} \tau^{-k_s} \beta_{i_n} \tau^{k_s}. \end{aligned}$

The application $\vartheta_s : A(s) * B(s) \to \mathbf{G} = \mathbb{Z} * A * B$ induces an embedding $\vartheta_s: \Gamma_s \to \Delta$. Indeed by Lemma 2.7, we easily check that if w represents an element γ in Γ_s , then $\vartheta_s(w) = ((f_s), 0)$ with

$$f_s(x) = \begin{cases} \gamma & \text{for } x = \frac{k_s}{2}, \\ e_{\Gamma_s} & \text{for } x \neq \frac{k_s}{2} \end{cases} \text{ and } f_{s'}(x) = \begin{cases} \theta_s^A(\gamma) & \text{for } x = \frac{k_s}{2}, \\ \theta_s^B(\gamma) & \text{for } x = \frac{k_s}{2} - k_{s'} \text{ for } s' \neq s. \end{cases}$$

By construction, $|\vartheta_s(\gamma)|_{\Delta} \leq (k_s+1) |\gamma|_{\Gamma_s}$. Moreover, it is clear that if w is a minimal representative of γ , then $\vartheta_s(w)$ is a minimal representative of $\pi_s(\vartheta_s(\gamma))$ in the quotient Δ_s . This proves the other inequality.

2.3.2. Embedding products with ℓ^{∞} -norm. We denote $\Gamma'_{s} = [A(s), B(s)]^{\Gamma_{s}}$. By Assumption 2.1, Γ'_{s} is the same as ker $(\Gamma_{s} \to A(s) \times B(s))$; it is a subgroup of Γ_{s} of finite index |A| |B|.

Given an integer $t \ge 0$, we consider

(2)
$$\Pi_s^t = \left\{ \begin{array}{ccc} f_s(x) \in \Gamma'_s & \text{ for } x \in [0,t) \\ ((f_s),0): & f_s(x) = e_{\Gamma_s} & \text{ for } x \notin [0,t) \\ & f_{s'} = e & \text{ for } s' \neq s \end{array} \right\}.$$

This is a subset of Δ . Indeed, $\theta_s(f_s) = e$ by choice of Γ'_s , so $\theta_0(f_0) = e$ by Lemma 2.7. Thus all such elements $((f_s), 0)$ actually belong to Δ by Lemma 2.13. Clearly, Π^t_s is a subgroup of Δ isomorphic to a direct product of t copies of Γ'_s :

$$\Pi_s^t \simeq \prod_{t \in [0,t)} \Gamma_s'.$$

By abuse of notation, we denote the elements of Π_s^t simply by functions f_s : $[0,t) \to \Gamma_s'$. The metric induced by Δ on Π_s^t can be estimated via Lemma 2.13.

LEMMA 2.16. For any f_s in Π_s^t ,

$$\frac{1}{2}k_s \max_{[0,t)} |f_s(x)|_{\Gamma_s} \le |f_s|_{\Delta} \le 36t \max_{[0,t)} |f_s(x)|_{\Gamma_s}.$$

In particular, diam_{Δ} $\Pi_s^t \leq 36t \operatorname{diam}_{\Gamma_s}(\Gamma_s') \leq 36t l_s$, where $l_s = \operatorname{diam}(\Gamma_s)$. Moreover,

$$\left| \left\{ f_s \in \Pi_s^t : |f_s|_{\Delta} \ge \frac{1}{72} \operatorname{diam}_{\Delta} \left(\Pi_s^t \right) \right\} \right| \ge \frac{1}{2} \left| \Pi_s^t \right|.$$

The last statements imply that Π_s^t satisfies the $(p; \frac{\dim_{\Delta}(\Pi_s^t)}{72}, \frac{1}{2})$ -mass distribution condition (21); see Section 5.

Proof. This follows from Lemma 2.13. To get the lower bound, notice that as $f_s(x)$ belongs to Γ'_s , we necessarily have $|f_s(x)|_{\Gamma_s} \geq 2$ when $f_s(x) \neq e_{\Gamma_s}$. To get the upper bound, observe that there are $[2t/k_s] + 1$ intervals I_i^s intersecting

[0, t), so the essential contribution is at most

$$E_s(f_s) \le k_s \left(\left\lfloor \frac{2t}{k_s} \right\rfloor + 1 \right) \max_{[0,t)} |f_s(x)|_{\Gamma_s} \le (2t+1) \max_{[0,t)} |f_s(x)|_{\Gamma_s}$$

and the range is bounded by t.

To get the second part, observe that for more than half of functions $I_j^s \to \Gamma'_s$, there exists $x \in I_j^s$ with $|f_s(x)|_{\Gamma_s} \ge \operatorname{diam}_{\Gamma_s}(\Gamma'_s)/2$. This holds for each j. Therefore there exists a subset A of Π_s^t of size $|A| \ge |\Pi_s^t|/2$ such that for each $f_s \in A$, more than half of the $[2t/k_s] + 1$ intervals I_j^s intersecting [0,t) satisfy $\max_{I_s^s} |f_s(x)|_{\Gamma_s} \ge \operatorname{diam}_{\Gamma_s}(\Gamma'_s)/2$. This implies that for any $f_s \in A$,

$$|f_s|_{\Delta} \ge E_s(f_s) \ge \frac{1}{4} \left(\left[\frac{2t}{k_s} \right] + 1 \right) k_s \operatorname{diam}_{\Gamma_s}(\Gamma'_s) \\ \ge \frac{t}{2} \operatorname{diam}_{\Gamma_s}(\Gamma'_s) \ge \frac{1}{72} \operatorname{diam}_{\Delta}(\Pi^t_s).$$

Example 2.17. When $\Gamma_s = D_{2l_s}$ is a dihedral group of size $2l_s$, then $\Gamma'_s \simeq \mathbb{Z}_{l_s/2}$ is a cyclic group. Edges of $\mathbb{Z}_{l_s/2}$ have length 4 in the D_{2l_s} metric. For $t = \frac{k_s}{2}$, Lemma 2.16 gives

$$|f_s|_{\Delta} \simeq_{72} k_s \max_{[0,\frac{k_s}{2})} |f_s(x)|_{\mathbb{Z}_{l_s/2}}.$$

In particular, Π_s^t is then a copy of the discrete torus $\mathbb{Z}_{l_s/2}^{k_s/2}$ with l^{∞} -metric rescaled by k_s , embedded with bounded distortion in Δ .

We fix a generating set for Γ'_s using the following classical lemma.

LEMMA 2.18 (Reidemeister-Schreier algorithm). We let (Γ, S) be a group marked with a finite generating set and $\pi : \Gamma \to F$ be a surjective mapping to a finite group F. Then

(1) there exists a set $C = \{a_1, \ldots, a_{|F|}\}$ of coset representatives

$$\Gamma = \bigcup_{i=1}^{|F|} (\operatorname{Ker} \pi) \, a_i$$

of length $|a_i|_S \leq \operatorname{diam}_{\pi(S)}(F);$

(2) the set R = CSC⁻¹ ∩ Ker π is a finite symmetric generating set of Ker π;
(3) for any γ ∈ Ker π,

$$|\gamma|_R \le |\gamma|_S \le \left(2\operatorname{diam}_{\pi(S)}(F) + 1\right)|\gamma|_R.$$

For Γ_s that satisfies Assumption 2.1 and $F = A(s) \times B(s)$, we fix a generating set R(s) for $\Gamma'_s = \ker(\Gamma_s \to A(s) \times B(s))$ provided by the Reidemeister-Schreier algorithm. In this case diam $(A(s) \times B(s)) = 2$. It follows from Lemma 2.18 that the inclusion map from $(\Gamma'_s, R(s))$ into $(\Gamma_s, A(s) \cup B(s))$ is bi-Lipschitz $|\gamma|_{R(s)} \leq |\gamma|_{\Gamma_s} \leq 5|\gamma|_{R(s)}$ for all $\gamma \in \Gamma'_s$ and that $|R(s)| \leq (|A||B|)^5$. Consider the direct product

$$H = \prod_{s \in \mathbb{N}} \left(\Gamma'_s \right)^{k_s/2} = \prod_{s \ge 1} \Pi_s^{k_s/2},$$

and denote elements of H as $\mathbf{h} = (h_s)$, where h_s is a vector

$$h_s = \left(h_s(0), \dots h_s\left(\frac{k_s}{2} - 1\right)\right) \in \left(\Gamma'_s\right)^{k_s/2}.$$

Equip H with a left invariant metric l,

$$l_s(\mathbf{h}) = rac{k_s}{2} \max_{0 \le j \le k_s/2 - 1} |h_s(j)|_{R_s}, \ l(\mathbf{h}) = \sum_{s \in \mathbb{N}} l_s(\mathbf{h}).$$

PROPOSITION 2.19. Suppose $\{\Gamma_s\}$ is a sequence of finite groups satisfying Assumption 2.1 and (k_s) satisfies the growth assumption 2.11. Let Δ be the diagonal product constructed with $\{\Gamma_s\}$ and parameters (k_s) . Then Δ is elementary amenable, and there exists an embedding $\theta : H \to \Delta$ such that for every $\mathbf{h} \in H$,

$$\max_{s \in \mathbb{N}} l_s(\mathbf{h}) \le |\theta(\mathbf{h})|_{\Delta} \le 45000 \ l(\mathbf{h}).$$

Proof. The group Δ is elementary amenable by Fact 2.10. By Proposition 2.14 and Lemma 2.16, we have for each $s \geq 0$,

$$|\theta(\mathbf{h})|_{\Delta} \ge |\pi_s \theta(\mathbf{h})|_{\Delta_s} \ge \frac{k_s}{2} \max_{[0,k_s/2)} |h_s(j)|_{\Gamma_s} \ge \frac{k_s}{2} \max_{[0,k_s/2)} |h_s(j)|_{R_s} = l_s(\mathbf{h})$$

and similarly

$$\begin{aligned} |\theta(\mathbf{h})|_{\Delta} \leq & 500 \sum_{s \geq 0} |\pi_s(\theta(\mathbf{h}))|_{\Delta_s} \leq 500 \sum_{s \geq 0} 10k_s \max_{[0,k_s/2)} |h_s(j)|_{\Gamma_s} \\ \leq & 500 \cdot 10 \cdot 5 \sum_{s \geq 0} k_s \max_{[0,k_s/2)} |h_s(j)|_{R_s} = 25000 \ l(\mathbf{h}). \end{aligned}$$

Since (k_s) satisfies the growth assumption 2.11, it is clear that H has at most exponential volume growth with respect to the length function l. By the general theorem of Olshanskii-Osin [OO13], there exists an elementary amenable group G equipped with a finite generating set S such that H embeds as a subgroup of G, and there exists a constant c > 0 such that $c |h|_S \leq l(h) \leq |h|_S$ for all h. In general the group G provided by the Olshanskii-Osin embedding is rather large compared to H. In the current setting, although the embedding $\theta : H \to \Delta$ is not necessarily bi-Lipschitz, the geometry of group Δ is in some sense controlled by H. In particular, we will show in Sections 6 and 8 that if $\{\Gamma_s\}$ is taken to be an expander family or finite dihedral groups and if the sequences (k_s) , $(\dim_{R(s)}(\Gamma'_s))$ satisfy certain growth conditions, then the Hilbert compression exponent of (Δ, d_{Δ}) is the same as (H, l),

$$\alpha_2^*\left((\Delta, d_\Delta)\right) = \alpha_2^*((H, l)).$$

3. Speed and entropy of random walk on Δ

Recall that Δ denotes the diagonal product of the sequence of marked groups $\{\Delta_s\}$. It is marked with generating tuple

$$\mathcal{T} = (au, lpha_1, \dots, lpha_{|A|}, eta_1, \dots, eta_{|B|})$$

Let U_{α} and U_{β} denote the uniform measure on the subgroups $A = \{\alpha_1, \ldots, \alpha_{|A|}\}$ and $B = \{\beta_1, \ldots, \beta_{|B|}\}$ respectively. Let μ denote the uniform measure on $\{\tau^{\pm 1}\}$. For the convenience of speed calculation, we take the following specific "switch-walk-switch" step distribution on Δ :

$$q = (U_{\alpha} * U_{\beta}) * \mu * (U_{\alpha} * U_{\beta}).$$

Note that in the construction of Δ , since $\Gamma_0 = A(0) \times B(0)$ and $k_s > 0$ for all s > 0, it follows that α_i commutes with β_j , and therefore $U_{\alpha} * U_{\beta}$ is a symmetric probability measure on Δ . Let $A_{1,s}, A_{2,s}, \ldots$ (resp. $B_{1,s}, B_{2,s}, \ldots$) be a sequence of independent random variables with uniform distribution on A(s) (resp. B(s)). We will refer to $A_{1,s}B_{1,s}\cdots A_{t,s}B_{t,s}$ as a random alternating word of length t in A(s) and B(s) starting with A.

We first describe what the random walk trajectory with step distribution q looks like. Let X_1, X_2, \ldots be a sequence of independent and identically distributed random variables uniform on $\{\pm 1\}$, $S_n = X_1 + \cdots + X_n$; $\mathcal{A}_1, \mathcal{A}_2, \ldots$ (resp. $\mathcal{B}_1, \mathcal{B}_2, \ldots$) a sequence of independent and identically distributed random variables with distribution U_{α} (resp. U_{β}). Let W_n denote the random variable on Δ given by

$$W_n = \mathcal{A}_1 \mathcal{B}_1 \tau^{X_1} \mathcal{A}_2 \mathcal{B}_2 \cdots \mathcal{A}_{2n-1} \mathcal{B}_{2n-1} \tau^{X_n} \mathcal{A}_{2n} \mathcal{B}_{2n}.$$

Then W_n has distribution q^n . Each letter \mathcal{A} can be written as $\mathcal{A} = ((f_s^{\mathcal{A}}), 0)$, where $f_s^{\mathcal{A}}(0) = \mathcal{A}_s$, with $\mathcal{A}_s = a_j(s)$ if $\mathcal{A} = \alpha_j$, and $f_s^{\mathcal{A}}(x) = e_{\Gamma_s}$ for all $x \neq 0$; similarly, $\mathcal{B} = ((f_s^{\mathcal{B}}), 0)$ where $f_s^{\mathcal{B}}(k_s) = \mathcal{B}_s$, with $\mathcal{B}_s = b_j(s)$ if $\mathcal{B} = \beta_j$, and $f_s^{\mathcal{B}}(x) = e_{\Gamma_s}$ for all $x \neq k_s$.

Now we rewrite W_n into the standard form $((f_s), z)$. Consider the projection to the copy Δ_s , from the definition of generators $\alpha_i(s)$ and $\beta_i(s)$,

$$f_{s}^{W_{n}}(y) = \mathcal{A}_{1,s}^{\mathbf{1}_{\{S_{0}=y\}}} \mathcal{B}_{1,s}^{\mathbf{1}_{\{S_{0}=y-k_{s}\}}} \\ \cdot \left(\prod_{j=1}^{n-1} \left(\mathcal{A}_{2j,s} \mathcal{A}_{2j+1,s} \right)^{\mathbf{1}_{\{S_{j}=y\}}} \left(\mathcal{B}_{2j,s} \mathcal{B}_{2j+1,s} \right)^{\mathbf{1}_{\{S_{j}=y-k_{s}\}}} \right) \\ \cdot \mathcal{A}_{2n,s}^{\mathbf{1}_{\{S_{n}=y\}}} \mathcal{B}_{2n}^{\mathbf{1}_{\{S_{n}=y-k_{s}\}}}.$$

For $x \in \mathbb{Z}$, let T(k, x, m) be the number of excursions of the simple random walk $\{S_n\}$ away from x that cross x - k and are completed before time m. Then conditioned on $\{S_k\}_{0 \le k \le n}$, the distribution of $f_s^{W_n}(y)$ is the same as a random alternating word in A(s) and B(s) of length $T(k_s, y, n)$ with an appropriate random letter added as prefix/suffix.

3.1. The case with linear speed in $\{\Gamma_s\}$. In this subsection we consider the case where speed of a simple random walk on Γ_s grows linearly up to some time comparable to the diameter l_s .

Definition 3.1. Let $\{\Gamma_s\}$ be a sequence of finite groups where each Γ_s is marked with a generating set $A(s) \cup B(s)$. Let $\eta_s = U_{A(s)} * U_{B(s)} * U_{A(s)}$, where $U_{A(s)}, U_{B(s)}$ are uniform distributions on A(s), B(s). We say $\{\Gamma_s\}$ satisfies the (σ, T_s) -linear speed assumption if in each Γ_s ,

$$L_{\eta_s}(t) = \mathbf{E} \left| X_t^{(s)} \right|_{\Gamma_s} \ge \sigma t \text{ for all } t \le T_s,$$

where $X_t^{(s)}$ has distribution η_s^{*t} .

Recall that since $X_t^{(s)}$ is a random walk on a transitive graph, by [AV17, Prop. 8],

$$\max\left\{H(X_t^{(s)}), 1\right\} \ge \frac{1}{t} \left(\frac{1}{4}\mathbf{E} \left|X_t^{(s)}\right|_{\Gamma_s}\right)^2.$$

Note that if $\{\Gamma_s\}$ satisfies Assumption 2.1, η_s projects onto the uniform distribution on $A(s) \times B(s)$, and thus $H(X_t^{(s)}) \ge H(X_1^{(s)}) \ge \log(|A|||B||) \ge \log 4$. Therefore, in this case, the (σ, T_s) -linear speed assumption implies that

(3)
$$H(X_t^{(s)}) \ge \sigma' t \text{ for all } t \le T_s, \text{ where } \sigma' = \left(\frac{\sigma}{4}\right)^2.$$

One important class of examples that satisfies the linear speed assumption consists of expander families.

Example 3.2. On $\Gamma_s = \langle A(s), B(s) \rangle$, $A(s) \simeq A, B(s) \simeq B$, let d = |A(s)| + |B(s)| - 2, v_s be the uniform probability measure on $A(s) \cup B(s)$. Suppose there exists $\delta > 0$ such that the spectral gap $\lambda(\Gamma_s, v_s)$ satisfies

$$\lambda(\Gamma_s, v_s) = \inf_{f:\Gamma_s \to \mathbb{R}, f \neq c} \left\{ \frac{\sum_{u, v \in \Gamma_s} |f(u) - f(uv)|^2 \nu_s(v)}{\frac{1}{|\Gamma_s|} \sum_{u, v \in \Gamma_s} |f(u) - f(v)|^2} \right\} \ge \delta$$

for all s; that is, $\{\Gamma_s\}$ forms a family of d-regular δ -expanders in ℓ^2 . Then $\{\Gamma_s\}$ satisfies the $(\sigma, c_0 \log |\Gamma_s|)$ -linear speed assumption with constants $\sigma, c_0 > 0$ only depending on δ and |A|, |B|. We reproduce the proof of this fact for completeness; see [HLW06, Th. 3.6].

By standard comparison of Dirichlet forms,

$$\lambda(\Gamma_s, \eta_s) = \hat{\delta} \ge \frac{\delta}{|A||B|}.$$

From the spectral gap we have

$$\left|\mathbf{P}(X_t^{(s)} = x) - \frac{1}{|\Gamma_s|}\right| \le e^{-\hat{\delta}t}.$$

Then for $t < \frac{1}{\hat{\delta}} \log |\Gamma_s|, \gamma = \frac{\hat{\delta}}{2 \log d}$,

$$\mathbf{P}\left(X_t^{(s)} \in B(e,\gamma t)\right) \le d^{\gamma t} \left(e^{-\hat{\delta}t} + \frac{1}{|\Gamma_s|}\right) \le 2\exp\left(\left(\gamma \log d - \hat{\delta}\right)t\right) = 2e^{-\hat{\delta}t/2}.$$

Therefore $\mathbf{E} \left| X_t^{(s)} \right|_{\Gamma_s} \geq \gamma t \left(1 - 2e^{-\hat{\delta}t/2} \right)$. We conclude that $\{\Gamma_s\}$ satisfies the $(\sigma, c_0 \log |\Gamma_s|)$ -linear speed assumption with $\sigma = \min \left\{ \frac{\hat{\delta}}{4 \log d}, \frac{\hat{\delta}}{2 \log 4} \right\}, c_0 = 1/\hat{\delta}$.

The lamplighter groups over \mathbb{Z}^d , $d \geq 3$, are the first examples of solvable groups where simple random walk has linear speed; see Kaimanovich-Vershik [KV83]. The following examples satisfying the linear speed assumption are analogues of finite quotients of $\mathbb{Z}_2 \wr \mathbb{Z}^d$.

Example 3.3. Let $\Gamma = \mathbb{Z}_2 \wr D_{\infty}^d$, $d \geq 3$ as in the second item of Example 2.4, marked with generating subgroups $A = \mathbb{Z}_2 \wr \langle a_j, 1 \leq j \leq d \rangle$, $B = \mathbb{Z}_2 \wr \langle b_j, 1 \leq j \leq d \rangle$. Fix an increasing sequence $n_s \in \mathbb{N}$, and let $\Gamma_s = \mathbb{Z}_2 \wr D_{2n_s}^d$. Then Γ_s is a finite quotient of Γ . Let A(s), B(s) denote the projection of A and B to Γ_s . There exists constant $\sigma_d > 0$ only depending on d such that $\{\Gamma_s\}$ satisfies the $(\sigma_d, (2n_s)^d)$ -linear speed assumption. A proof of this fact is included in Lemma C.4 in the appendix.

3.1.1. Bounds on speed and entropy in one copy. In the upper bound direction, we will use the trivial bound that in each lamp group,

(4)
$$\left| f_s^{W_n}(x) \right|_{\Gamma_s} \leq \min \left\{ 2T(k_s, x, n) + \mathbf{1}_{\{L(x,n)>0\}} + \mathbf{1}_{\{L(x-k_s,n)>0\}}, \operatorname{diam}(\Gamma_s) \right\},$$

where $L(x,n) = \#\{0 < k \le n : S_k = x\}$ is the local time at x. Recall that we set the parameter $l_s = \text{diam}(\Gamma_s)$.

LEMMA 3.4. There exists an absolute constant C > 0 such that for all $s \ge 0$,

$$\mathbf{E}\left[\left|\left(f_{s}^{W_{n}}, S_{n}\right)\right|_{\Delta_{s}} \mathbf{1}_{\{s \leq s_{0}(W_{n})\}}\right] \leq \begin{cases} Cn^{\frac{1}{2}} \min\left\{\frac{n^{\frac{1}{2}}}{k_{s}}, l_{s}\right\} & \text{if } k_{s}^{2} \leq n, \\ C\left(n^{\frac{1}{2}} + k_{s}\right)e^{-\frac{k_{s}^{2}}{8n}} & \text{if } k_{s}^{2} > n. \end{cases}$$

Proof. From the metric upper estimate in Lemma 2.13,

$$\left| \left(f_s^{W_n}, S_n \right) \right|_{\Delta_s} \le 9 \left(\sum_{j \in \mathbb{Z}} k_s \max_{x \in I_j^s} \left| f_s^{W_n}(x) \right|_{\Gamma_s} + \mathcal{R}_n \right),$$

where $\mathcal{R}_n = \# \{S_k : 0 \le k \le n\}$ is the size of the range of simple random walk on \mathbb{Z} and $I_j^s = \left[j\frac{k_s}{2}, (j+1)\frac{k_s}{2}\right)$. Observe that for each $x \in I_j^s$,

(5)
$$T(k_s, x, n) \le T\left(\frac{k_s}{2}, j\frac{k_s}{2}, n\right),$$

because each excursion from x to the left that crosses $x - k_s$ must contain an excursion from $j\frac{k_s}{2}$ to the left that crosses $(j-1)\frac{k_s}{2}$. Apply (4):

$$\left| \left(f_s^{W_n}, S_n \right) \right|_{\Delta_s} \le 9 \sum_{j \in \mathbb{Z}} k_s \max_{x \in I_j^s} \left\{ 2T(k_s, x, n) \right\} + 11(\mathcal{R}_n)$$
$$\le 11 \left(\sum_{j \in \mathbb{Z}} k_s T\left(\frac{k_s}{2}, j\frac{k_s}{2}, n\right) + \mathcal{R}_n \right).$$

By Lemma A.1,

$$\mathbf{E}T\left(\frac{k_s}{2}, j\frac{k_s}{2}, n\right) \le \frac{2Cn^{\frac{1}{2}}}{k_s} \exp\left(-\frac{(jk_s/2)^2}{2n}\right).$$

The size of the range of the simple random walk on \mathbb{Z} satisfies

$$\mathbf{P}\left(\mathcal{R}_n \ge x\right) \le \mathbf{P}\left(\max_{0 \le k \le n} |S_k| \ge \frac{x}{2}\right) \le 4\exp\left(-\frac{x^2}{8n}\right).$$

Recall that by definition of $s_0(g)$ in Section 2.2.2,

$$\{s \le s_0(W_n)\} \subseteq \{\mathcal{R}_n \ge k_s\}.$$

Summing up,

$$\begin{split} & \mathbf{E}\left[\left|\left(f_{s}^{W_{n}}, S_{n}\right)\right|_{\Delta_{s}} \mathbf{1}_{\{s \leq s_{0}(W_{n})\}}\right] \\ & \leq 11k_{s}\sum_{j \in \mathbb{Z}} \mathbf{E}T\left(\frac{k_{s}}{2}, j\frac{k_{s}}{2}, n\right) + 11\mathbf{E}\left[\left(\mathcal{R}_{n}\right)\mathbf{1}_{\{\mathcal{R}_{n} \geq k_{s}\}}\right] \\ & \leq 11k_{s}\sum_{j \in \mathbb{Z}} \frac{Cn^{\frac{1}{2}}}{k_{s}} \exp\left(-\frac{\left(jk_{s}/2\right)^{2}}{2n}\right) + C'n^{\frac{1}{2}}e^{-\frac{k_{s}^{2}}{8n}} \\ & \leq \begin{cases} C''\left(\frac{n}{k_{s}} + n^{\frac{1}{2}}\right) & \text{if } k_{s}^{2} \leq n, \\ C''\left(n^{\frac{1}{2}} + k_{s}\right)e^{-\frac{k_{s}^{2}}{8n}} & \text{if } k_{s}^{2} > n. \end{cases} \end{split}$$

For $k_s^2 \leq n$, since $|f_s^{W_n}(y)|_{\Gamma_s}$ cannot exceed the diameter of Γ_s , together with Lemma 2.13, we have a second upper bound

$$\mathbf{E}|(f_s^{W_n}, S_n)|_{\Delta_s} \le 10l_s \mathbf{E}(\mathcal{R}_n + k_s) \le Cl_s n^{\frac{1}{2}}.$$

Combining these bounds, we obtain the statement.

Now we turn to the lower bound direction.

LEMMA 3.5. Suppose that $\{\Gamma_s\}$ satisfies the $(\sigma, c_0 l_s)$ -linear speed assumption. Then there exists an absolute constant C > 0 such that for s with $k_s^2 \leq n$,

$$\mathbf{E}\left(\left|\left(f_s^{W_n}, S_n\right)\right|_{\Delta_s}\right) \ge \frac{\sigma}{C}\min\left\{\frac{n}{k_s}, c_0 n^{\frac{1}{2}} l_s\right\}$$

and

$$H(f_s^{W_n}) \ge \frac{\sigma'}{C} \min\left\{\frac{n}{k_s}, c_0 n^{\frac{1}{2}} l_s\right\}.$$

Proof. We use a weaker lower bound for the metric,

$$|(f_s, z)|_{\Delta_s} \ge \sum_{y \in \mathbb{Z}} |f_s(y)|_{\Gamma_s}$$
.

Applying the $(\sigma, c_0 l_s)$ -linear speed assumption of Definition 3.1, we have

$$\mathbf{E}\left(\left|\left(f_{s}^{W_{n}}, S_{n}\right)\right|_{\Delta_{s}}\right) \geq \mathbf{E}\left(\sum_{y \in \mathbb{Z}}\left|f_{s}^{W_{n}}(y)\right|\right) \geq \sum_{y \in \mathbb{Z}}\mathbf{E}\left[\sigma \min\left\{T(k_{s}, y, n), c_{0}l_{s}\right\}\right].$$

Then by Lemma A.2, there exists a constant c > 0 for $k_s \leq c^2 n^{\frac{1}{2}}$:

$$\mathbf{E}\left[\min\left\{T(k_s, y, n), c_0 l_s\right\}\right] \ge \frac{1}{2}\min\left\{\frac{c\sqrt{n}}{4k}, c_0 l_s\right\} \mathbf{P}\left(L\left(y, \frac{n}{2}\right) > 0\right).$$

Summing up over y,

$$\sum_{y \in \mathbb{Z}} \mathbf{E} \left[\min \left\{ T(k_s, y, n), c_0 l_s \right\} \right] \ge \frac{1}{2} \min \left\{ \frac{c\sqrt{n}}{4k_s}, c_0 l_s \right\} \mathbf{E} \mathcal{R}_{n/2},$$

where $\mathcal{R}_{n/2}$ is the size of the range of simple random walk on \mathbb{Z} up to n/2. Since $\mathbf{E}\mathcal{R}_n \simeq n^{\frac{1}{2}}$, it follows that there exists a constant C > 1 such that for s with $k_s^2 \leq n$,

$$\mathbf{E}\left(\left|\left(f_s^{W_n}, S_n\right)\right|_{\Delta_s}\right) \ge \frac{\sigma}{C} \min\left\{\frac{n}{k_s}, c_0 n^{\frac{1}{2}} l_s\right\}.$$

Concerning entropy, we condition by the traverse time function (see, for instance, [AV17] for the basic properties of entropy),

$$H(f_s^{W_n}) \ge H(f_s^{W_n} | T(k_s, \cdot, n)) = \sum_{z \in \mathbb{Z}} \mathbf{E} H(X_{T(k_s, z, n)}^{(s)})$$
$$\ge \sum_{z \in \mathbb{Z}} \mathbf{E} \left[\sigma' \min\{T(k_s, z, b), c_0 l_s\} \right] \ge \frac{\sigma'}{C} \min\left\{ \frac{n}{k_s}, c_0 n^{\frac{1}{2}} l_s \right\},$$

using (3) and the same computation.

3.1.2. Speed and entropy estimates in the diagonal product Δ . Recall from Assumption 2.11 that (k_s) grows exponentially. In the diagonal product Δ , by the metric upper estimate in Proposition 2.14 and speed upper estimates in Lemma 3.4, we have

(6)
$$\mathbf{E}(|W_n|_{\Delta}) \leq \mathbf{E}\left(500\sum_{s\leq s_0(W_n)} \left|\left(f_s^{W_n}, S_n\right)\right|_{\Delta_s}\right)$$
$$\leq \sum_{s\leq s_0(n)} C\left(\min\left\{\frac{n}{k_s}, n^{\frac{1}{2}}l_s\right\} + n^{\frac{1}{2}}\right),$$

where

$$s_0(n) = \min\{s: k_s^2 \ge n\}.$$

Indeed, denote $x_s = \frac{k_s}{\sqrt{n}}$ growing exponentially. Then

$$\sum_{s>s_0(n)} (n^{\frac{1}{2}} + k_s) e^{-\frac{k_s^2}{8n}} \le n^{\frac{1}{2}} \sum_{x_s \ge 1} (1 + x_s) e^{-\frac{x_s^2}{8}} \le Cn^{\frac{1}{2}}.$$

In the lower bound direction, suppose $\{\Gamma_s\}$ satisfies the (σ, c_0) -linear speed Assumption 3.1. Then by the metric lower estimate in Proposition 2.14 and Lemma 3.5,

(7)
$$\mathbf{E}\left(|W_n|_{\Delta}\right) \ge \max_{s} \mathbf{E}\left(\left|\left(f_s^{W_n}, S_n\right)\right|_{\Delta_s}\right) \ge \frac{\sigma}{C} \max_{s \le s_0(n)} \min\left\{\frac{n}{k_s}, n^{\frac{1}{2}} c_0 l_s\right\}.$$

To understand the upper bound (6), divide the collection of Δ_s with $s \leq s_0(n)$ into two subsets:

(i) Let $s_1(n)$ denote the index

$$s_1(n) = \max\left\{s \ge 0: n^{\frac{1}{2}} \ge k_s l_s\right\}.$$

Then the contribution of these $s \leq s_1(n)$ to the sum is bounded by

$$\sum_{s \le s_1(n)} \mathbf{E} \left| \left(f_s^{W_n}, S_n \right) \right|_{\Delta_s} \le C n^{\frac{1}{2}} \sum_{s \le s_1(n)} \left(l_s + 1 \right)$$

(ii) The contribution of $s \in (s_1(n) + 1, s_0(n)]$ to the sum is bounded by

$$\sum_{s=s_1(n)+1}^{s_0(n)} \mathbf{E} \left| \left(f_s^{W_n}, S_n \right) \right|_{\Delta_s} \le C \sum_{s=s_1(n)+1}^{s_0(n)} \left(\frac{n}{k_s} + n^{\frac{1}{2}} \right).$$

Combining these parts, we have

s=

(8)
$$\mathbf{E}(|W_n|_{\Delta}) \le 2C \left(n^{\frac{1}{2}} \sum_{s=0}^{s_1(n)} l_s + \sum_{s=s_1(n)+1}^{s_0(n)} \frac{n}{k_s} \right).$$

PROPOSITION 3.6. Suppose that $\{\Gamma_s\}$ satisfies the $(\sigma, c_0 l_s)$ -linear speed assumption and diam $(\Gamma_s) \leq C_0 l_s$. Suppose there exists a constant $m_0 > 1$ such that

 $k_{s+1} > 2k_s, l_{s+1} \ge m_0 l_s$ for all s.

Let

$$s_0(n) = \min\{s: k_s^2 \ge n\}, \ s_1(n) = \max\{s \ge 0: n^{\frac{1}{2}} \ge k_s l_s\}.$$

Then

$$\frac{\sigma c_0}{2C} \left(n^{\frac{1}{2}} l_{s_1(n)} + \frac{n}{k_{s_1(n)+1}} \right) \le \mathbf{E} |W_n|_{\Delta} \le \frac{4C}{1 - 1/m_0} \left(n^{\frac{1}{2}} l_{s_1(n)} + \frac{n}{k_{s_1(n)+1}} \right).$$

The same bounds hold for the entropy $H(W_n^{\Delta})$ with σ replaced by $\sigma' = \left(\frac{\sigma}{4}\right)^2$ and C replaced by a constant C' > 0 that only depends on the size of generating sets |A| + |B|.

Proof. The lower bound is a direct consequence of (7). For the upper bound, apply (8) and note that because of the assumption on growth of k_s, l_s , the sums satisfy

$$\sum_{\substack{s \le s_1(n) \\ s \ge s_1(n)+1}} l_s \le l_{s_1(n)} \frac{1}{1 - 1/m_0}$$

For the entropy, by [Ers03, Lemma 6], there is C' depending only on the exponential rate of volume growth in Γ , which cannot exceed $\log(|A| + |B|)$, such that $H(W_n^{\Delta}) \leq C' \mathbf{E} |W_n|_{\Delta}$, giving the upper bound. Lemma 3.5 gives the lower bound (7).

3.1.3. Possible speed and entropy functions.

Example 3.7. If Γ_s is an family as in Examples 3.2 or 3.3 and $k_s = 2^{\beta s + o(s)}$ and $l_s = 2^{\iota s + o(s)}$, with $\beta, \iota \in [1, \infty)$, a direct application of Proposition 3.6 shows that the speed and entropy exponents are

$$\lim \frac{\log \mathbf{E}|W_n|_{\Delta}}{\log n} = \lim \frac{\log H(W_n^{\Delta})}{\log n} = \frac{\beta + 2\iota}{2\beta + 2\iota} = 1 - \frac{1}{2(1 + \frac{\iota}{\beta})}$$

which can take any value in $(\frac{1}{2}, 1)$.

THEOREM 3.8. There exist universal constants c, C > 0 such that the following holds. For any function $\varrho : [1, \infty) \to [1, \infty)$ such that $\frac{\varrho(x)}{\sqrt{x}}$ and $\frac{x}{\varrho(x)}$ are non-decreasing, there exist a group Δ and a symmetric probability measure q of finite support on Δ such that

$$c\varrho(n) \leq \mathbf{E} |W_n|_{\Delta} \leq C\varrho(n) \text{ and } c\varrho(n) \leq H(W_n^{\Delta}) \leq C\varrho(n).$$

Moreover, the group Δ can be chosen to be 4-step solvable.

Proof. The choice of a family of Γ_s as in Examples 3.2 or 3.3 guarantees the existence of $C_1 > 1$ such that for all $x \ge 1$, there is Γ_s of diameter l_s with $\frac{x}{C_1} \le l_s \le C_1 x$.

As ρ belongs to $C_{\frac{1}{2},1}$, Corollary B.3 provides two sequences (k_s) of integers and (l_s) among diameters of Γ_s with $k_{s+1} \ge m_0 k_s$ and $l_{s+1} \ge m_0 l_s$ for all ssuch that the function

$$\bar{\varrho}(x) = x^{\frac{1}{2}}l_s + \frac{x}{k_{s+1}}$$
 for $(k_s l_s)^2 \le x \le (k_{s+1} l_{s+1})^2$

satisfies $\bar{\varrho}(x) \simeq_{2m_0 C_1^5} \varrho(x)$ for all x.

Combining with Proposition 3.6, the diagonal product Δ associated to these sequences has a speed and entropy satisfying

$$\frac{\sigma c_0}{4m_0 C C_1^5} \rho(x) \le \mathbf{E} |W_n|_{\Delta} \le \frac{4m_0 C C_1^5}{1 - \frac{1}{m_0}} \rho(x).$$

For $\{\Gamma_s\}$ as in Example 3.3, the group Δ is 4-step solvable.

Note that when the speed is linear, the last term of the sequence (l_s) is infinite, and thus the last quotient Γ_s is in fact the whole group Γ . In our examples, Γ is either $\mathbb{Z}_2 \wr D^d_{\infty}$ for $d \geq 3$ or a lattice in SL(3, F). In the latter case, the finite diagonal product Δ is non-amenable. When the speed is diffusive, the last term of the sequence (k_s) is infinite, and the group Δ is a diagonal product of finitely many groups Δ_s where the lamp groups Γ_s are finite.

3.2. The case of Δ with dihedral groups. In this subsection we focus on the case where $\{\Gamma_s\}$ is taken to be a sequence of finite dihedral groups. Since the unlabelled Cayley graph of D_{2l} is the same as a cycle of size 2l, the simple random walk on D_{2l} can be identified with the simple random walk on the cycle of size 2l. Consider a simple random walk on the cycle as the projection of the simple random walk on \mathbb{Z} . Then the classical Gaussian bounds on the simple random walk on \mathbb{Z} imply that there exist constants c, C > 0 such that for all $1 \le t \le l_s^2, 1 \le x \le t^{\frac{1}{2}}$,

(9)
$$c \exp\left(-Cx^{2}\right) \leq \mathbf{P}\left(\left|X_{t}^{(s)}\right|_{D_{2l_{s}}} \geq xt^{\frac{1}{2}}\right) \leq C \exp\left(-cx^{2}\right),$$

where $X_t^{(s)}$ is a random alternating word in $\{e, a(s)\}$ and $\{e, b(s)\}$.

LEMMA 3.9. Suppose $\Gamma_s = D_{2l_s}$. There exists an absolute constant C > 0 such that in each Δ_s ,

$$\begin{split} \mathbf{E} \left[\left| \left(f_s^{W_n}, S_n \right) \right|_{\Delta_s} \mathbf{1}_{\{s \le s_0(W_n)\}} \right] \\ \le \begin{cases} C \min \left\{ n^{\frac{3}{4}} k_s^{-\frac{1}{2}} \log^{\frac{1}{2}} k_s, nk_s^{-1}, n^{\frac{1}{2}} l_s \right\} & \text{ if } k_s^2 < n, \\ C \left(n^{\frac{1}{2}} + k_s \right) e^{-\frac{k_s^2}{8n}} & \text{ if } k_s^2 \ge n. \end{cases} \end{split}$$

Proof. From the upper bounds in Lemma 3.4 that are valid for any choice of $\{\Gamma_s\}$, the only bound we need to show here is that if $k_s \leq n^{\frac{1}{2}}$, then

(10)
$$\mathbf{E}\left[\left|\left(f_{s}^{W_{n}}, S_{n}\right)\right|_{\Delta_{s}} \mathbf{1}_{\{s \leq s_{0}(W_{n})\}}\right] \leq Cn^{\frac{3}{4}} k_{s}^{-\frac{1}{2}} \log^{\frac{1}{2}} k_{s}.$$

To prove this, note that the collection $(|f_s^{W_n}(z)|_{D_{\infty}})_{z \in I_j^s}$ as a vector is stochastically dominated by the random vector

$$\left(\left|X_{T(k_s/2,jk_s/2,n)}(z)\right|_{D_{\infty}} + \mathbf{1}_{\{L(n,z)>0\}} + \mathbf{1}_{\{L(n,z-k_s)>0\}}\right)_{z\in I_j^s}\right)$$

where $\{X_t(z)\}\$ is a sequence of independent random alternating words in A(s)and B(s) of length t. Then $\max_{z \in I_j^s} |f_s^{W_n}(z)|_{D_{\infty}}$ is stochastically dominated by

$$\max_{z \in I_j^s} \left| X_{T(k_s/2, jk_s/2, n)}(z) \right|_{D_{\infty}} + \mathbf{1}_{\{L(n, I_j^s) > 0\}} + \mathbf{1}_{\{L(n, I_{j-2}^s) > 0\}}.$$

Plug in the metric estimate in Lemma 2.13,

(11)
$$\mathbf{E} \left| \left(f_s^{W_n}, S_n \right) \right| \mathbf{1}_{\{s \le s_0(W_n)\}}$$
$$\leq 9k_s \left(\sum_{j \in \mathbb{Z}} \mathbf{E} \max_{z \in I_j^s} \left| X_{T(k_s/2, jk_s/2, n)}(z) \right|_{D_{\infty}} \right) + 11 \mathbf{E} \left[\mathcal{R}_n \mathbf{1}_{\{\mathcal{R}_n \ge k_s\}} \right].$$

From the upper bound in (9), since $\max_{z \in I_j^s} |X_t(z)|$ is maximum of $k_s/2$ independent and identically distributed random variables,

$$\mathbf{P}\left(\max_{z\in I_{j}^{s}}|X_{t}(z)|\leq xt^{\frac{1}{2}}\right)\geq\left(1-c_{1}\exp\left(-c_{2}x^{2}\right)\right)^{\left|I_{j}^{s}\right|}$$

Then

$$\mathbf{E}\left[\max_{z\in I_{j}^{s}}\left|X_{T(k_{s}/2,jk_{s}/2,n)}(z)\right|\left|\left|T(k_{s}/2,jk_{s}/2,n)\right|\right] \le C_{1}T(k_{s}/2,jk_{s}/2,n)^{\frac{1}{2}}\log^{\frac{1}{2}}k_{s},$$

where C_1 depends on c_1, c_2 . Applying Lemma A.1,

$$\mathbf{E}\left[\max_{z\in I_j^s} \left| X_{T(k_s/2, jk_s/2, n)}(z) \right| \right] \le C_2 \left(\frac{n^{\frac{1}{2}}}{k_s} \exp\left(-\frac{(jk_s/2)^2}{2n} \right) \right)^{\frac{1}{2}} \log^{\frac{1}{2}} k_s.$$

Plugging in the estimates in (11) and summing up over j, we obtain (10), because of the Gaussian tail. The main contribution comes from $1 \leq j \leq n^{\frac{1}{2}}/k_s$.

LEMMA 3.10. Suppose $\Gamma_s = D_{2l_s}$. There exists an absolute constant c > 0 such that in each Δ_s , for $n \ge ck_s^2$,

$$\mathbf{E}\left| \left(f_{s}^{W_{n}}, S_{n} \right) \right|_{\Delta_{s}} \geq c \min\left\{ n^{\frac{3}{4}} k_{s}^{-\frac{1}{2}} \log^{\frac{1}{2}} k_{s}, n k_{s}^{-1}, n^{\frac{1}{2}} l_{s} \right\}.$$

Proof. For each $z \in I_j^s = [\frac{j}{2}k_s, \frac{j+1}{2}k_s)$, the traverse time satisfies $T(k_s, z, n) \ge T(2k_s, \frac{j+1}{2}k_s, n)$. So

$$\mathbf{E}\max_{z\in I_{j}^{s}}\left|f_{s}^{W_{n}}(z)\right|_{D_{2l_{s}}} \geq \mathbf{E}\max_{z\in I_{j}^{s}}\left|X_{T(2k_{s},\frac{j+1}{2}k_{s},n)}^{s}(z)\right|_{D_{2l_{s}}},$$

where $\{X_t(z)\}\$ is a sequence of independent random alternating words in A(s) and B(s) of length t.

For any $t \ge 1$ and $1 \le x \le t^{\frac{1}{2}}$,

$$\mathbf{P}\left(|X_t(z)|_{D_{2l_s}} \ge \min\left\{xt^{\frac{1}{2}}, \frac{l_s}{2}\right\}\right) \ge c_1 e^{-c_2 x^2}$$

By independence, this implies the existence of c > 0 with

$$\mathbf{P}\left(\max_{z\in I_{j}^{s}}|X_{t}(z)|_{D_{2l_{s}}}\geq \min\left\{c\log^{\frac{1}{2}}k_{s}t^{\frac{1}{2}},ct,\frac{l_{s}}{2}\right\}\right)\geq c.$$

Using Lemmas 2.13 and A.2, for some c > 0 and for all $n \ge ck_s^2$,

$$\begin{split} \mathbf{E}|(f_s^{W_n}, S_n)|_{\Delta_s} &\geq \frac{k_s}{4} \sum_{j \in \mathbb{Z}} \mathbf{E} \max_{z \in I_j^s} \left| X_{T(2k_s, \frac{j+1}{2}k_s, n)}^s(z) \right|_{D_{2l_s}} \\ &\geq ck_s \sum_{j \in \mathbb{Z}} \min\left\{ n^{\frac{1}{4}} k_s^{-\frac{1}{2}} \log^{\frac{1}{2}} k_s, n^{\frac{1}{2}} k_s^{-1}, \frac{l_s}{2} \right\} \\ &\quad \cdot P_0\left(L\left(\frac{j+1}{2}k_s, \frac{n}{2}\right) > 0 \right). \end{split}$$

Finally $\sum_{j\in\mathbb{Z}} P_0\left(L(\frac{j+1}{2}k_s, \frac{n}{2}) > 0\right) \ge \mathbf{E}\mathcal{R}_{\frac{n}{2}}/k_s \ge c\frac{n^{\frac{1}{2}}}{k_s}.$

PROPOSITION 3.11. Suppose that $\Gamma_s = D_{2l_s}$ and that there exists $m_0 > 1$ such that $k_{s+1} > 2k_s$, $l_{s+1} > m_0 l_s$ for all s. Define

$$t_1(n) = \max\left\{s: \ \frac{l_s^2 k_s}{\log k_s} < n^{\frac{1}{2}} \ and \ l_s k_s < n^{\frac{1}{2}}\right\}.$$

Then in the diagonal product Δ ,

$$\frac{n^{\frac{1}{2}}}{C} \left(l_{t_1(n)} + \min\left\{ n^{\frac{1}{4}} \left(\frac{\log k_{t_1(n)+1}}{k_{t_1(n)+1}} \right)^{\frac{1}{2}}, \frac{n^{\frac{1}{2}}}{k_{t_1(n)+1}} \right\} \right) \leq \mathbf{E}(|W_n|_{\Delta}) \\
\leq \frac{2Cn^{\frac{1}{2}}}{1 - 1/m_0} \left(l_{t_1(n)} + \min\left\{ n^{\frac{1}{4}} \left(\frac{\log k_{t_1(n)+1}}{k_{t_1(n)+1}} \right)^{\frac{1}{2}}, \frac{n^{\frac{1}{2}}}{k_{t_1(n)+1}} \right\} \right).$$

Remark 3.12. The bounds here are more complicated than the linear case because in Lemmas 3.9 and 3.10 we have to consider minimum over three

quantities. If we further assume that $l_s \ge \log k_s$ for all s, then the bounds simplify to

$$\mathbf{E}(|W_n|_{\Delta}) \simeq_{Cm_0} n^{\frac{1}{2}} l_{t_1(n)} + n^{\frac{3}{4}} \left(\frac{\log k_{t_1(n)+1}}{k_{t_1(n)+1}} \right)^{\frac{1}{2}}.$$

Proof. By Proposition 2.14 and the third line of Lemma 3.9,

$$c \max_{s \le s_0(n)} \mathbf{E} |W_n|_{\Delta_s} \le \mathbf{E} |W_n|_{\Delta} \le C \sum_{s \le s_0(n)} \mathbf{E} |W_n|_{\Delta_s}.$$

The choice of $t_1(n)$ implies that

$$\min\left\{n^{\frac{3}{4}}k_{s}^{-\frac{1}{2}}\log^{\frac{1}{2}}k_{s}, nk_{s}^{-1}, n^{\frac{1}{2}}l_{s}\right\} = \begin{cases}n^{\frac{1}{2}}l_{s} & \forall s \leq t_{1}(n),\\ \min\left\{n^{\frac{3}{4}}k_{s}^{-\frac{1}{2}}\log^{\frac{1}{2}}k_{s}, nk_{s}^{-1}\right\} & \forall s > t_{1}(n).\end{cases}$$

Using Lemmas 3.9 and 3.10 and the exponential growth of $(k_s), (l_s)$ gives the proposition.

Example 3.13. Let $k_s = 2^{\beta s}$ and $l_s = 2^{\iota s}$, with $\beta > 1$, $\iota > 0$, and $\Gamma_s = D_{2l_s}$. Proposition 3.11 implies with this choice of parameters that

$$\mathbf{E}\left(\left|W_{n}\right|_{\Delta}\right) \simeq n^{\frac{3\iota+\beta}{4\iota+2\beta}} \left(\log n\right)^{\frac{\iota}{2\iota+\beta}}.$$

THEOREM 3.14. There exist universal constants c, C > 0 such that the following holds. For any continuous function $\varrho : [1, \infty) \to [1, \infty)$ satisfying $\varrho(1) = 1$ and $\frac{\varrho(x)}{x^{\frac{1}{2}} \log^{1+\epsilon} x}, \frac{x^{\frac{3}{4}}}{\varrho(x)}$ non-decreasing for some $\epsilon > 0$, there exists a 3-step solvable group Δ with dihedral groups Γ_s and a symmetric probability measure q of finite support on Δ such that

$$c\varrho(n) \leq \mathbf{E} |W_n|_{\Delta} \leq C\varrho(n).$$

Remark 3.15. The lower condition on $\rho(x)$ is only technical. There is no gap that would isolate diffusive behaviors among 3-step solvable groups. Indeed, it easily follows from Lemmas 3.9 and 3.10 that for any $\rho(x)$ such that $\frac{\rho(x)}{\sqrt{x}}$ tends to infinity, there is a group Δ with dihedral Γ_s such that $cn^{\frac{1}{2}} \leq \mathbf{E}|W_n|_{\Delta} \leq C\rho(n)$ for all n.

Proof. By Corollary B.3, we can find two sequences $(\kappa_s), (l_s)$ satisfying $\log \kappa_s \leq l_s$ such that $\bar{\varrho}(x)$ and $\varrho(x)$ agree up to multiplicative constants.

Let us set $\kappa_s = \left(\frac{k_s}{\log k_s}\right)^{\frac{1}{2}}$. Then $\log k_s \simeq \log \kappa_s \le l_s^{\frac{1}{1+\epsilon}}$. By Corollary 3.11, the diagonal product Δ with dihedral groups $\Gamma_s = D_{2l_s}$ and sequence (k_s) satisfies, for any $(l_s \kappa_s)^4 \le n \le (l_{s+1} \kappa_{s+1})^4$,

$$c\bar{\varrho}(x) = c\left(n^{\frac{1}{2}}l_s + \frac{n^{\frac{3}{4}}}{\kappa_{s+1}}\right) \leq \mathbf{E} |W_n|_{\Delta} \leq C\left(n^{\frac{1}{2}}l_s + \frac{n^{\frac{3}{4}}}{\kappa_{s+1}}\right) = C\bar{\varrho}(x).$$

The group Δ clearly has a trivial third derived subgroup.

PROPOSITION 3.16. There exist two constants c, C > 0 such that on any diagonal product Δ with dihedral Γ_s satisfying Assumption 2.11, the entropy of the switch-walk-switch random walk satisfies

$$c\sqrt{n} \le H(W_n^{\Delta}) \le C\sqrt{n}\log^2 n.$$

Proof. By Proposition 2.14, the entropy of the random walk on Δ is related to the entropy on the factors Δ_s by the following:

$$\max_{s \ge 0} H(f_s^{W_n}, S_n) \le H((f_s^{W_n}), S_n) \le H(S_n) + \sum_{s \le s_2(n)} H(f_s^{W_n}),$$

where $s_0(W_n) \leq s_2(n) = \max\{s : k_s \leq n\}$ and $H(S_n) \simeq \log n$. The lower bound comes from the first factor $H(f_0^{W_n}, S_n)$, which is the usual random walk on the lamplighter group with finite lamps.

Denote by $T_s^n = T(k_s, \cdot, n)$ the traverse time function and by $[\operatorname{supp} T_s^n]$ the convex envelope of its support. Using conditional entropy (see, for instance, [KV83] or [AV17] for standard properties of conditional entropy), we have

(12)
$$H(f_s^{W_n}) \le H(f_s^{W_n}|T_s^n) + H(T_s^n) \\ \le H(f_s^{W_n}|T_s^n) + H(T_s^n|[\operatorname{supp} T_s^n]) + H([\operatorname{supp} T_s^n]).$$

The convex envelope $[\operatorname{supp} T_s^n]$ is included in $\operatorname{Range}(W_n)$, and therefore $H([\operatorname{supp} T_s^n]) \simeq \log n$ and $\mathbf{E}|[\operatorname{supp} T_s^n]| \le C\sqrt{n}$. As for all $z, 0 \le T_s^n(z) \le n$, we deduce that

$$H(T_s^n | [\operatorname{supp} T_s^n]) \le \mathbf{E} [| [\operatorname{supp} T_s^n] |] \log n \le C\sqrt{n} \log n.$$

As each $f_s^{W_n}(z)$ is distributed as an independent sample of a random walk on Γ_s of length $T(k_s, z, n)$, we have

$$H(f_s^{W_n}|T_s^n) = \sum_{z \in \mathbb{Z}} \mathbf{E} H(X_{T(k_s,z,n)}^{(s)}).$$

The groups Γ_s being dihedral $H(X_t^{(s)}) \leq \log t$, we therefore have

$$H(f_s^{W_n}|T_s^n) \le \sum_{z \in \mathbb{Z}} \mathbf{E} \log T(k_s, z, n) \le C\sqrt{n} \log n.$$

Obviously $n \ge k_{s_0(W_n)}$, so finally piling up the inequalities,

$$H((f_s^{W_n}), S_n) \le C \log(n)(1 + 2\sqrt{n}\log n + \log n).$$

3.3. Joint evaluation of speed and entropy. Using the idea of Amir [Ami17] of taking the direct product of two groups, we can combine the speed and entropy estimates on Δ together with the results of Amir-Virag [AV17] to show the following result concerning the joint behavior of growth of entropy and speed.

Recall that for symmetric probability measure μ on G with finite support, entropy and speed satisfy

$$\frac{1}{n}\left(\frac{1}{4}L_{\mu}(n)\right)^{2} - 1 \le H_{\mu}(n) \le (v+\varepsilon)L_{\mu}(n) + \log n + C,$$

where v is the volume growth rate of $(G, \operatorname{supp}\mu)$ and C > 0 is an absolute constant ([Ers03], [AV17]).

PROPOSITION 3.17. Let $f, h : \mathbb{N} \to \mathbb{N}$ be two functions such that h(1) = 1and

• either $\frac{f(n)}{n^{\frac{3}{4}}}$ and $\frac{n^{1-\epsilon}}{f(n)}$ are non-decreasing for some $\epsilon > 0$ and

$$h(n) \le f(n) \le \sqrt{\frac{nh(n)}{\log n}},$$

• or $\frac{h(n)}{n^{\frac{1}{2}}\log^2 n}$ and $\frac{n^{\frac{3}{4}}}{f(n)}$ are non-decreasing and

$$h(n) \le f(n) \le \sqrt{nh(n)}.$$

Then there exist a constant C > 0 depending only on $\epsilon > 0$, a finitely generated group G and a symmetric probability measure μ of finite support on G such that

 $L_{\mu}(n) \simeq_C f(n)$ and $H_{\mu}(n) \simeq_C h(n)$.

As a corollary, we derive the Corollary 1.3 regarding joint entropy and speed exponents (Conjecture 3 in [Ami17]).

Proof of Corollary 1.3. If $\theta \in \left[\frac{1}{2}, 1\right)$, then take functions

$$h(n) = \max\left\{n^{\frac{1}{2}}\log^2 n, n^{\theta}\right\}, \quad f(n) = \max\left\{n^{\gamma}, h(n)\right\},$$

Note that in the case $\gamma < 1$, the pair of functions f and h is covered by one of the cases in Proposition 3.17; the statement follows.

If $\theta = 1$, in this case $\gamma = 1$ as well. We can take G to be any finitely generated group that admits a symmetric probability measure μ of finite support such that (G, μ) has linear entropy growth.

Proof. We follow Amir's approach in [Ami17] to take the direct product of two appropriate groups such that one would control the speed function and the other would control the entropy function.

In the first case where both $f(n)/n^{\frac{3}{4}}$ and $n^{1-\epsilon}/f(n)$ are non-decreasing, consider the direct product of the following two groups. By [AV17], there exist a group $G_1 = \mathbb{Z} \wr_{\mathcal{S}} \mathcal{M}_m$ and step distribution q_1 on G_1 such that

$$L_{q_1}(n) \simeq f(n)$$
 and $H_{q_1}(n) \simeq \frac{f(n)^2}{n} \log(n+1)$.

36
By Theorem 3.8, there exist a group Δ and step distribution q_2 on Δ such that

$$L_{q_2}(n) \simeq H_{q_2}(n) \simeq h(n).$$

Then on the direct product $G_1 \times \Delta$ with step distribution $q_1 \otimes q_2$,

$$L_{q_1 \otimes q_2}(n) \simeq \max \{ L_{q_1}(n), L_{q_2}(n) \} \simeq f(n),$$

$$H_{q_1 \otimes q_2}(n) \simeq \max \{ H_{q_1}(n), H_{q_2}(n) \} \simeq \max \left\{ \frac{f(n)^2}{n} \log(n+1), h(n) \right\} = h(n).$$

In the second case, where $f(n)/(n^{\frac{1}{2}}\log^2 n)$ and $n^{\frac{3}{4}}/f(n)$ are non-decreasing, by Theorem 3.14 and Proposition 3.16, there exist a group Δ_1 and step distribution q'_1 on Δ_1 such that

$$L_{q'_1}(n) \simeq f(n) \text{ and } \frac{1}{C} n^{\frac{1}{2}} \le H_{q'_1}(n) \le C n^{\frac{1}{2}} \log^2 n.$$

By Theorem 3.8, there exist a group Δ_2 and step distribution q'_2 such that

$$L_{q_2'}(n) \simeq H_{q_2'}(n) \simeq h(n)$$

Then on the direct product $\Delta_1 \times \Delta_2$ with step distribution $q'_1 \otimes q'_2$,

$$L_{q'_1 \otimes q'_2}(n) \simeq \max \left\{ L_{q'_1}(n), L_{q'_2}(n) \right\} \simeq f(n),$$

$$H_{q'_1 \otimes q'_2}(n) \simeq \max \left\{ H_{q'_1}(n), H_{q'_2}(n) \right\} \simeq h(n).$$

Remark 3.18. This method permits us to prove Corollary 1.3 but cannot handle functions that oscillate cross the $n^{\frac{3}{4}}$ borderline. The problem can be reduced to finding extremal examples where the speed follows a prescribed function while entropy growth is as slow as possible. The Amir-Virag result covers the case where speed grows at least like $n^{\frac{3}{4}}$. The construction of the diagonal product Δ is not designed to achieve such a goal.

4. Isoperimetric profiles and return probabilities

In this section we consider the isoperimetric profiles and return probabilities of Δ when $\{\Gamma_s\}$ are taken to be expanders or dihedral groups. For convenience of calculation, we take the switch-or-walk measure $\mathfrak{q} = \frac{1}{2}(\mu + \nu)$ on Δ where μ is the simple random walk measure on the base \mathbb{Z} , $\mu(\tau^{\pm 1}) = \frac{1}{2}$, and ν is uniform on $\{\alpha_i, \beta_j : 1 \leq i \leq |A|, 1 \leq j \leq |B|\}$. Let \mathfrak{q}_s be the projection of \mathfrak{q} to the quotient Δ_s .

4.1. Isoperimetric profiles. We first recall some background information. Given a symmetric probability measure ϕ on G, the p-Dirichlet form associated with (G, ϕ) is

$$\mathcal{E}_{p,\phi}(f) = \frac{1}{2} \sum_{x,y \in G} |f(xy) - f(x)|^p \phi(y)$$

and the ℓ^p -isoperimetric profile $\Lambda_{p,G,\phi}: [1,\infty) \to \mathbb{R}$ is defined as

(13)
$$\Lambda_{p,G,\phi}(v) = \inf \left\{ \mathcal{E}_{p,\phi}(f) : |\mathrm{support}(f)| \le v, ||f||_p = 1 \right\}.$$

The most important ones are the ℓ^1 and ℓ^2 -isoperimetric profiles. Using an appropriate discrete co-area formula, $\Lambda_{1,\phi}$ can equivalently be defined by

$$\Lambda_{1,G,\phi}(v) = \inf \left\{ |\Omega|^{-1} \sum_{x,y} \mathbf{1}_{\Omega}(x) \mathbf{1}_{G \setminus \Omega}(xy) \phi(y) : |\Omega| \le v \right\}.$$

If we define the boundary of Ω to be the set

$$\partial \Omega = \{(x,y) \in G \times G : x \in \Omega, y \in G \setminus \Omega\}$$

and set $\phi(\partial \Omega) = \sum_{x \in \Omega, xy \in G \setminus \Omega} \phi(y)$, then

$$\Lambda_{1,G,\phi}(v) = \inf\{\phi(\partial\Omega)/|\Omega| : |\Omega| \le v\}$$

When ϕ is a symmetric measure supported by a finite generating set S, then $\Lambda_{1,G,\phi}(v)$ is closely related to the Følner function : $(0,\infty) \to \mathbb{N}$ defined as

$$\operatorname{Fol}_{G,S}(r) = \min\left\{ |\Omega|: \ \Omega \subset G, \ \frac{|\partial_S \Omega|}{|\Omega|} < \frac{1}{r} \right\},$$

where $|\partial_S \Omega| = \{x \in \Omega : \exists u \in S, xu \notin \Omega\}$. Namely, let $p_* = \min\{\phi(u) : u \in \Omega\}$ $S, u \neq id$; then

$$\Lambda_{1,G,\phi}^{-1}(1/r) \le \operatorname{Fol}_{G,S}(r) \le \Lambda_{1,G,\phi}^{-1}(p_*/r),$$

where $\Lambda_{1,G,\phi}^{-1}$ is the generalized inverse of $\Lambda_{1,G,\phi}$. We will repeatedly use the following two facts.

For any $1 \leq p \leq q \leq 2$, the isoperimetric profiles $\Lambda_{p,G,\phi}$ and $\Lambda_{q,G,\phi}$ are related by the Cheeger type inequality

(14)
$$c_0 \Lambda_{p,G,\phi}^{q/p} \le \Lambda_{q,G,\phi} \le C_0 \Lambda_{p,G,\phi},$$

where c_0, C_0 are absolute constants; see [LS88], [SCZ16, Prop. 2.8].

Let H be a quotient group of G, and let ϕ be the projection of ϕ on H. Then by [Tes13, Prop. 4.5], for all $1 \le p < \infty$,

$$\Lambda_{p,G,\phi} \ge \Lambda_{p,H,\bar{\phi}}.$$

4.1.1. Isoperimetric profiles of one factor. Let $\{\Gamma_s\}$ be a family of expanders, as in Example 2.3. Let $\nu = \nu_s$ be the uniform distribution on $A(s) \cup B(s)$ in Γ_s . Denote by $h(\Gamma_s, \nu)$ the Cheeger constant

$$h(\Gamma_s, \nu) = \inf \left\{ \Lambda_{1, \Gamma_s, \nu}(v) : v \leq \frac{|\Gamma_s|}{2}, v < \infty \right\}$$

In one copy $\Delta_s = \Gamma_s \wr \mathbb{Z}$, we establish the following lower bound of ℓ^p -isoperimetric profile of Δ_s .

LEMMA 4.1. Let $p \in [1,2]$, Γ_s be a finite group marked with generating subgroups A(s), B(s). Let \mathfrak{q}_s be the uniform distribution on $\{\tau^{\pm 1}\} \cup A(s) \cup B(s)$ in Δ_s . Then there exists an absolute constant C > 1 such that the following is true:

• (Slow phase) for $1 \le v \le 2^{k_s/2}$,

$$\Lambda_{p,\Delta_s,\mathfrak{q}_s}(v) \ge \frac{\left(\log_2 v\right)^{-p}}{C\left(|A(s)| + |B(s)|\right)}$$

• (Fast, then slow down) for $|\Gamma_s|^r \le v \le |\Gamma_s|^{r+1}, r \ge k_s$,

$$\Lambda_{p,\Delta_s,\mathfrak{q}_s}(v) \ge \frac{h(\Gamma_s,\nu)^p}{C\left(|A(s)| + |B(s)|\right)r^p}.$$

Remark 4.2. By monotonicity of the profile function Λ_{p,Δ_s,q_s} , we have that

$$\Lambda_{p,\Delta_s,\mathfrak{q}_s}(v) \ge \frac{h(\Gamma_s,\nu)^p}{C\left(|A(s)| + |B(s)|\right)k_s^p} \quad \text{for } v \in \left(2^{k_s/2}, |\Gamma_s|^{k_s}\right).$$

Proof. We prove a lower bound for the ℓ^1 -isoperimetric profile $\Lambda_{1,\Delta_s,\mathfrak{q}_s}$ and use the Cheeger inequality (14) to derive a lower bound on $\Lambda_{p,\Delta_s,\mathfrak{q}_s}$. Given $r \geq 1$, consider the product of Γ_s in the zero-section over the segment [0, r], namely,

$$\Pi_r = \prod_{x \in [0,r]} \left(\Gamma_s \right)_x.$$

We now construct a product kernel ζ_r on Π_r and discuss its ℓ^1 -isoperimetry.

Phase I. In the first phase, $r < k_s/2$. Let η_x denote the uniform measure on the finite subgroup $A_s \simeq \mathbb{Z}/2\mathbb{Z}$ in the copy $(\Gamma_s)_x$. Let ζ_r denote the product kernel

$$\zeta_r = \eta_0 \otimes \cdots \otimes \eta_r.$$

Then ζ_r is indeed the uniform measure on the subgroup $\prod_{x \in [0,r]} (A_s)_x$ in the zero section of Δ_s , it follows that

$$\Lambda_{1,\Delta_s,\zeta_r}(v) \ge \frac{1}{2} \text{ for all } v \le \frac{1}{2} \left(2^{r+1} \right).$$

Phase II. In the second phase, $r > k_s$. Let ν_x denote the uniform measure on the generating set $A(s) \cup B(s)$ in the copy $(\Gamma_s)_x$. Let ζ_r denote the product kernel

$$\zeta_r = \nu_0 \otimes \cdots \otimes \nu_r.$$

As a transition kernel, ζ_r changes every copy $(\Gamma_s)_x$ independently according to ν_x . By [BH97, Th. 1.1], the Cheeger constant of ζ_r on Π_r satisfies

$$h(\Pi_r, \zeta_r) \ge \frac{1}{2\sqrt{6}}h(\Gamma_s, \nu).$$

In other words, for $r > k_s$,

$$\Lambda_{1,\Delta_s,\zeta_r}(v) \geq \frac{1}{2\sqrt{6}} h(\Gamma_s,\nu) \ \text{ for all } v \leq \frac{1}{2} \left| \Gamma_s \right|^{r+1}.$$

By the Cheeger inequality (14), for $p \in [1, 2]$,

$$\Lambda_{p,\Delta_s,\zeta_r}(v) \ge c_0 \Lambda_{1,\Delta_s,\zeta_r}(v)^p.$$

Now we go back to the simple random walk kernel \mathfrak{q} . By construction of the transition kernel ζ_r in both cases, the metric estimate in Lemma 2.13 implies every element g in the support of ζ_r satisfies

$$|g|_{\Delta_s} \leq 40r.$$

By the standard path length argument (see [PSC00, Lemma 2.1]), we have by comparison of Dirichlet forms on Δ_s that

$$2\left(|A|+|B|\right)\mathcal{E}_{p,\Delta_s,\mathfrak{q}_s} \geq \frac{1}{(40r)^p}\mathcal{E}_{p,\Delta_s,\zeta_r}.$$

It follows that

- for $r < \frac{k_s}{2}$, $\Lambda_{p,\Delta_s,\mathfrak{q}_s}(v) \ge \frac{c_0}{2\left(|A| + |B|\right)(40r)^p} \text{ for all } v \le 2^r;$
- for $r > k_s$,

$$\Lambda_{p,\Delta_s,\mathfrak{q}_s}\left(v\right) \ge \frac{c_0 h(\Gamma_s,\nu)^p}{2\left(|A|+|B|\right) \left(80\sqrt{6}r\right)^p} \quad \text{for all } v \le \frac{1}{2} \left|\Gamma_s\right|^{r+1}.$$

When $\Gamma_s = \Gamma$ is an infinite group, we have the following bound.

LEMMA 4.3. Let $p \in [1, 2]$, $\Gamma_s = \Gamma$ be an infinite group marked with generating subgroups A, B. Let ν be the uniform distribution on $A \cup B$. Then there exists an absolute constant C > 1 such that the following is true:

• for $1 \le v \le 2^{k_s/2}$,

$$\Lambda_{p,\Delta_s,\mathfrak{q}_s}(v) \ge \frac{(\log_2 v)^{-p}}{C\left(|A(s)| + |B(s)|\right)};$$

• for $v > 2^{k_s/2}$,

$$\Lambda_{p,\Delta_s,\mathfrak{q}_s}(v) \ge \frac{h(\Gamma,\nu)^p}{C\left(|A(s)| + |B(s)|\right)k_s^p}.$$

Proof. The first item is the same as in proof of Lemma 4.1. For the second item, consider the copy of Γ at 0 and regard ν as a measure supported on the subgroup $(\Gamma)_0$. Then

$$C\left(|A(s)| + |B(s)|\right)k_s^p \mathcal{E}_{p,\Delta_s,\mathfrak{q}_s} \ge \mathcal{E}_{p,\Gamma,\nu},$$

and the result follows.

4.1.2. Isoperimetric profile of the diagonal product. First we put together isoperimetric estimates on the copies Δ_s to describe the isoperimetric profile of the diagonal product Δ . Let us denote $\ell_s = \log |\Gamma_s|$. Mind the difference with the diameter l_s of Γ_s . For a family of expanders, these two quantities differ only by multiplicative constants depending only on the volume growth and the spectral gap of $\{\Gamma_s\}$.

PROPOSITION 4.4. Suppose $\{(\Gamma_s, A(s) \cup B(s))\}$ with $A(s) \simeq A, B(s) \simeq B$ is a family of groups with the Cheeger constant $h(\Gamma_s, \nu_s) \ge \delta > 0$, where ν_s is uniform on $A(s) \cup B(s)$. Suppose $\{k_s\}$ satisfies the growth assumption 2.11. Let Δ be the diagonal product constructed with $\{\Gamma_s\}$ and parameters $\{k_s\}$, and let \mathfrak{q} be the uniform measure on $\{\tau^{\pm 1}\} \cup A \cup B$ in Δ .

There exists an absolute constant C > 1 such that the following estimates hold for any $s \ge 0$, $p \in [1, 2]$:

(1) for volume $v \in [e^{k_s \ell_s}, e^{k_{s+1} \ell_s}),$

$$\begin{split} \Lambda_{p,\Delta,\mathfrak{q}}(v) &\geq \frac{1}{|A| + |B|} \left(\frac{\delta \ell_s}{C \log v}\right)^p, \\ \Lambda_{p,\Delta,\mathfrak{q}}\left(v^{\frac{\sum_{j \leq s} \ell_j}{\ell_s}}\right) &\leq \left(\frac{C \ell_s}{\log v}\right)^p; \end{split}$$

(2) for volume $v \in [e^{k_{s+1}\ell_s}, e^{k_{s+1}\ell_{s+1}}],$

$$\begin{split} \Lambda_{p,\Delta,\mathfrak{q}}(v) &\geq \frac{1}{|A| + |B|} \left(\frac{\delta}{Ck_{s+1}}\right)^p, \\ \Lambda_{p,\Delta,\mathfrak{q}}\left(v\right) &\leq \left(\frac{C}{k_{s+1}}\right)^p \;\; if \; v \geq \exp{\left(\left(\sum_{j \leq s} \ell_j\right) k_{s+1}\right)}. \end{split}$$

The upper bounds are valid without the requirement of a positive Cheeger constant.

Proof. Let $U_r^{\Delta} = \{((f_s), z) : \operatorname{Range}(f_s, z) \subset [-r, r]\}$, and take a function supported on the subset U_r^{Δ} ,

$$\varphi_r\left((f_s), z\right) = \left(1 - \frac{|z|}{r}\right) \mathbf{1}_{U_r^{\Delta}}\left(\left((f_s), z\right)\right).$$

Let $U_r^{\Delta}(0) = \{((f_s), 0) : \operatorname{Range}(f_s, 0) \subset [-r, r]\}$. Then U_r^{Δ} can be viewed as the product of $U_r^{\Delta}(0)$ and the interval [-r, r]. To compute the Rayleigh quotient of the function φ_r , first note that $\varphi_r(Z\alpha_i) = \varphi_r(Z\beta_j) = \varphi_r(Z)$ for all $Z \in \Delta$ and $\alpha_i \in A, \beta_j \in B$. For the generator τ ,

$$\sum_{((f_s),z)\in U_r^{\Delta}} \left|\varphi_r\left((f_s), z+1\right) - \varphi_r\left((f_s), z\right)\right|^p = \frac{1}{r^p} (2r) \left|U_r^{\Delta}(0)\right|,$$
$$\sum_{((f_s),z)\in U_r^{\Delta}} \varphi_r\left((f_s), z\right)^p = \sum_{z\in[-r,r]} \left(1 - \frac{|z|}{r}\right)^p \left|U_r^{\Delta}(0)\right|$$

Therefore

$$\frac{\mathcal{E}_{p,\Delta,\mathfrak{q}}(\varphi_r)}{\|\varphi_r\|_p^p} \sim \frac{1+p}{2r^p}.$$

For the size of support of φ_r ,

$$|\operatorname{supp}\varphi_r| \leq \prod_{k_s \leq 2r} |\operatorname{supp}\varphi_r^s| \leq \prod_{k_s \leq 2r} |\Gamma_s|^r = e^{r \sum_{k_s \leq 2r} \ell_s}$$

In the first interval $v \in [e^{k_s \ell_s}, e^{k_{s+1} \ell_s})$, let

$$r = \frac{\log v}{\ell_s}.$$

The test function φ_r^{Δ} gives the upper bound on $\Lambda_{p,\Delta,\mathfrak{q}}$ stated. For the lower bound on $\Lambda_{p,\Delta,\mathfrak{q}}$, consider the projection to the quotient Δ_s . Then from the first item in Lemma 4.1, for $v \in [e^{k_s \ell_s}, e^{k_{s+1} \ell_s})$, we have

$$\Lambda_{p,\Delta,\mathfrak{q}}(v) \ge \Lambda_{p,\Delta_s,\mathfrak{q}_s}(v) \ge \frac{1}{|A| + |B|} \left(\frac{\delta}{Ck_s}\right)^p$$

In the second interval $v \in [e^{k_{s+1}\ell_s}, e^{k_{s+1}\ell_{s+1}}]$, first consider the projection to the quotient Δ_{s+1} . The second item in Lemma 4.1 provides

$$\Lambda_{p,\Delta,\mathfrak{q}}\left(\frac{1}{2}\left|\Gamma_{s+1}\right|^{k_{s+1}}\right) \geq \Lambda_{p,\Delta_s,\mathfrak{q}_s}\left(\frac{1}{2}\left|\Gamma_{s+1}\right|^{k_{s+1}}\right) \geq \frac{1}{|A|+|B|}\left(\frac{\delta}{Ck_{s+1}}\right)^p.$$

In the upper bound direction, note that the right end point in the first interval gives

$$\Lambda_{p,\Delta,\mathfrak{q}}\left(\exp\left(\left(\sum_{j\leq s}\ell_j\right)k_{s+1}\right)\right) \leq \left(\frac{C}{k_{s+1}}\right)^p.$$

The statement follows from monotonicity of $\Lambda_{p,\Delta,\mathfrak{q}}$.

Example 4.5. A direct application of Proposition 4.4 shows that when $k_s = 2^{\beta s}$ and $\ell_s = 2^{\iota s}$ with $\beta, \iota > 0$, then for $p \in [1, 2]$,

$$\Lambda_{p,\Delta,\mathfrak{q}}(v) \simeq \left(\log v\right)^{-\frac{p}{1+\frac{\nu}{\beta}}}$$

and the exponent $\frac{p}{1+\frac{\iota}{\beta}}$ can take any value in (0, p).

We allow the sequence (k_s) , (l_s) to take the value ∞ ; the bounds are still valid. In our convention, $k_{s+1} = \infty$ means Δ_{s+1} is trivial. In this case we only use the first item in Proposition 4.4, which covers $v \in [e^{k_s \ell_s}, \infty)$. The bounds in Proposition 4.4 are good when $\{\ell_s\}$ grows at least exponentially. In particular, from these estimates of isoperimetric profiles we deduce that $\Lambda_{p,\Delta,\mathfrak{q}} \circ \exp$ can follow a prescribed function satisfying some log-Lipschitz condition.

THEOREM 4.6. There exist universal constants c, C > 0 such that for any $p \in [1,2]$ and for any non-decreasing function $\varrho(x)$ such that $\frac{x^p}{\varrho(n)}$ is nondecreasing, there is a group Δ such that

$$\forall v \ge 3, \ \frac{c}{\varrho(\log v)} \le \Lambda_{p,\Delta,\mathfrak{q}}(v) \le \frac{C}{\varrho(\log v)}.$$

Proof. We write $\varrho(x) = \left(\frac{x}{f(x)}\right)^p$ with f(x) between 1 and x. The sets $K = \mathbb{Z}_+ \cup \{\infty\}$ and $L = \{\log |\Gamma_m|, m \ge 1\} \cup \{\infty\}$, where $\{\Gamma_s\}$ are groups in the family of Examples 2.3, satisfy the assumptions of Proposition B.2. So we can find sequences $(k_s), (l_s)$ taking values in K and L such that the function defined by $\tilde{f}(x) = l_s$ on $[k_s l_s, k_{s+1} l_s]$ and $\tilde{f}(x) = \frac{x}{k_{s+1}}$ on $[k_{s+1} l_s, k_{s+1} l_{s+1}]$ satisfies $\tilde{f}(x) \simeq_{m_0 C_1^5} f(x)$. Since the infinite group Γ in Example 2.3 has Property (T), there exists a constant $\delta > 0$ such that the Cheeger constants satisfy $h(\Gamma_s, \nu_s) \ge \delta$ for all $s \ge 1$.

We use Proposition 4.4 to evaluate the profile of the group Δ associated to these sequences. The lower bounds show that for all $x \ge 1$,

$$\Lambda_{p,\Delta,\mathfrak{q}} \circ \exp(x) \ge \left(\frac{\delta \tilde{f}(x)}{Cx}\right)^p \ge \frac{c\delta^p}{\varrho(x)}$$

As $\sum_{j \leq s} \ell_j \leq \frac{1}{1 - \frac{1}{m_0}} \ell_s$, making the change of variable $x = (1 - \frac{1}{m_0}) y = \log v$, the first upper bound shows that

$$\Lambda_{p,\Delta,\mathfrak{q}} \circ \exp(y) \le \left(\frac{Cl_s}{\log v}\right)^p = \left(\frac{C\tilde{f}\left((1-\frac{1}{m_0})y\right)}{(1-\frac{1}{m_0})y}\right)^p \le \left(\frac{C\tilde{f}(y)}{y}\right)^p \le \frac{C'}{\varrho(x)}$$

for $\frac{1}{1-\frac{1}{m_0}}k_s\ell_s \leq y \leq \frac{1}{1-\frac{1}{m_0}}k_{s+1}\ell_s$. The second upper bound shows that

$$\Lambda_{p,\Delta,\mathfrak{q}} \circ \exp(y) \le \left(\frac{C}{k_{s+1}}\right)^p = \left(\frac{C\tilde{f}(k_{s+1}l_s)}{k_{s+1}l_s}\right)^p = \left(\frac{C\tilde{f}\left((1-\frac{1}{m_0})y\right)}{(1-\frac{1}{m_0})y}\right)^p \le \frac{C''}{\varrho(y)}$$

for $\frac{1}{1-\frac{1}{m_0}}k_{s+1}\ell_s \leq y \leq \frac{1}{1-\frac{1}{m_0}}k_{s+1}\ell_{s+1}$. We used the fact that $\tilde{\varrho}(x)$ is constant on the interval $[k_{s+1}\ell_s, k_{s+1}\ell_{s+1}]$.

We derive the following corollary regarding Følner functions from Theorem 1.1. The definition of a Følner function is recalled in the beginning of Section 4. We use the convention that on a non-amenable group G, if $1/r \leq \inf \frac{|\partial S|}{|S|}$, then $\operatorname{Fol}_{G,S}(r) = \infty$.

COROLLARY 4.7. There exists an universal constant C > 1. Let $g : [1,\infty) \to [1,\infty]$ be any non-decreasing function with g(1) = 1 and $\frac{\log(g(x))}{x}$ non-decreasing. Then there exists a group Δ marked with finite generating set T such that

$$g(r/C) \le \operatorname{Fol}_{\Delta,T}(r) \le g(Cr).$$

Further, when range of g is contained in $[1, \infty)$, the group Δ constructed is elementary amenable and there exists a symmetric probability measure q with finite generating support on Δ such that (Δ, q) is Liouville.

Proof. Let $p_* = \min\{q(u) : u \in \mathcal{T}, u \neq id\}$; then $p_* \ge \frac{1}{2(|A|+|B|)}$. Recall that by definition of the Følner function

$$\Lambda_{1,\Delta,q}^{-1}(1/r) \le \operatorname{Fol}_{\Delta,\mathcal{T}}(r) \le \Lambda_{1,\Delta,q}^{-1}(p_*/r).$$

By Proposition 4.6, there exist universal constants C > 0 such that for any function $\rho(x)$ between 1 and x, there is a group Δ such that

$$\forall v \ge 3, \ \frac{1}{C\varrho(\log v)} \le \Lambda_{1,\Delta,\mathfrak{q}}(v) \le \frac{C}{\varrho(\log v)}.$$

In particular, in the construction of Δ we can choose $\{\Gamma_s\}$ from Lafforgue's expanders as in Example 2.3, where |A| = 2, $|B| = r_0$ for some fixed r_0 . Therefore

$$\exp\left(\varrho^{-1}(r/C)\right) \leq \operatorname{Fol}_{\Delta,\mathcal{T}}(r) \leq \exp(\varrho^{-1}(Cr_0r)).$$

Since ρ is any function between 1 and x, the statement about the Følner function follows.

When the range of g is in $[1, \infty)$, the group Δ in the proof of Proposition 4.6 is constructed with an infinite sequence of finite groups $\{\Gamma_s\}$ and $\{k_s\}$ satisfying the growth assumption (2.11). By Fact 2.10, Δ is elementary amenable. Apply Theorem 3.6, we have that $L_q(\mu)$ and $H_q(\mu)$ have sub-linear growth, thus (Δ, q) is Liouville.

4.2. Return probabilities of simple random walk on Δ . By Theorem 4.6 with p = 2, we have that $\Lambda_{2,\Delta,\mathfrak{q}} \circ \exp$ can follow a prescribed function satisfying some log-Lipschitz condition. Now we turn the ℓ^2 -isoperimetric profile estimates into return probability bounds using the Coulhon-Grigor'yan theory. Let μ be a symmetric probability measure on a group G. Between discrete time random walk and continuous time random walk, we have (see [PSC00, §3.2])

$$\mu^{(2n+2)}(e) \le 2h_{2n}^{\mu}(e)$$
 and $h_{4n}^{\mu}(e) \le e^{-2n} + \mu^{(2n)}(e)$,

where

(15)
$$h_t^{\phi} = e^{-t} \sum_{0}^{\infty} \frac{t^k}{k!} \phi^{(k)}.$$

Define the function $\psi : [0, +\infty) \to [1, +\infty)$ implicitly by

(16)
$$t = \int_{1}^{\psi(t)} \frac{dv}{v\Lambda_{2,G,\mu}(v)}$$

Then by [Cou96, Prop. II.1], we have

$$\mu^{(2n+2)}(e) \le \frac{8}{\psi(8n)}.$$

In the current context, it is convenient to do a change of variable in (16). Setting $v = \exp(s)$,

(17)
$$t = \int_{1}^{w(t)} \frac{ds}{\Lambda_{2,\Delta,\mathfrak{q}} \circ \exp(s)}.$$

If, in addition, $\Lambda_{2,G,\mu} \circ \exp$ is doubling, namely, $\Lambda_{2,G,\mu} \circ \exp(2s) \geq c\Lambda_{2,G,\mu} \circ \exp(s)$ for all s > 1, then by [BPS12, Prop. 2.3], w'(t) is doubling with the same constant. Applying [CG97, Th. 3.2],

$$\mu^{(2n)}(e) \ge \frac{1}{\exp \circ \psi(8n/c)} - e^{-2n}.$$

Combining the upper and lower bounds, if $\Lambda_{2,G,\mu} \circ \exp(2s) \ge c\Lambda_{2,G,\mu} \circ \exp(s)$, we have

(18)
$$-\log \mu^{(2n)}(e) \simeq_C w(2n)$$

with constant C > 0 only depending on the doubling constant c.

THEOREM 4.8. There exist universal constants c, C > 0 such that the following is true. Let $\gamma : [1, \infty) \to [1, \infty)$ be any function such that $\frac{\gamma(n)}{n^{\frac{1}{3}}}$ and $\frac{n}{\gamma(n)}$ are non-decreasing. Then there is a group Δ such that

$$\forall t \ge 1, \ c\gamma(t) \le -\log\left(\mathfrak{q}^{(2t)}\left(e_{\Delta}\right)\right) \le C\gamma(t).$$

Proof. Given such a function $\gamma : [1, \infty) \to [1, \infty)$, which is strictly increasing and continuous, define $\varrho : [1, \infty) \to [1, \infty)$ by

$$\varrho(x) = \frac{1}{x}\gamma^{-1}(x).$$

From the assumption on γ we have $\gamma(1) = 1$ and $a^{\frac{1}{3}}\gamma(x) \leq \gamma(ax) \leq a\gamma(x)$ for any $a, x \geq 1$. Thus

$$a\gamma^{-1}(x) \le \gamma^{-1}(ax) \le a^3\gamma^{-1}(x),$$

and therefore

$$\varrho(x) \le \varrho(ax) \le a^2 \varrho(x),$$

which satisfies the assumption of Proposition 4.6 with p = 2.

By Proposition 4.6, there exist universal constants c, C > 0 such that there is a group Δ , for all $v \geq 3$:

$$\frac{c}{\varrho(\log v)} \le \Lambda_{\Delta,\mathfrak{q}}(v) \le \frac{C}{\varrho(\log v)}.$$

Note that since $\varrho(2x) \leq 4\varrho(x)$, it follows that for all s > 0,

$$\Lambda_{\Delta,\mathfrak{q}} \circ \exp(2s) \ge \frac{c}{4C} \Lambda_{\Delta,\mathfrak{q}} \circ \exp(s).$$

In particular, the function $\Lambda_{\Delta,\mathfrak{q}} \circ \exp: (0,\infty) \to \mathbb{R}$ is doubling at infinity. Then by [BPS12, Lemma 2.5], the solution w(t) to (17) satisfies

$$\Lambda_{\Delta,\mathfrak{q}} \circ \exp \circ w(t) \leq \frac{w(t)}{t} \leq D\Lambda_{\Delta,\mathfrak{q}} \circ \exp \circ w(t),$$

where D is a constant that only depends on the doubling constant c/4C. Plugging in the estimate of $\Lambda_{\Delta,\mathfrak{q}}$, we have

$$\frac{c}{\varrho(w(t))} \le \frac{w(t)}{t} \le \frac{DC}{\varrho(w(t))}.$$

By the definition of ρ ,

$$\gamma(t)\varrho\left(\gamma(t)\right) = t,$$

note that $x \rho(x)$ is strictly increasing, and therefore

$$c^{\frac{1}{3}}\gamma(t) \leq \gamma(ct) \leq w(t) \leq \gamma(DCt) \leq DC\gamma(t).$$

Since the constants c, C, D are universal, from (18) provided by the Coulhon-Grigor'yan theory, we conclude that

$$\forall t \ge 1, \ c'\gamma(t) \le -\log\left(\mathfrak{q}^{(2t)}\left(e_{\Delta}\right)\right) \le C'\gamma(t),$$

where c', C' > 0 are universal constants.

Example 4.9. When $k_s = 2^{\beta s}$ and $l_s = 2^{\iota s}$ with $\beta > 1$, $\iota > 0$, the L^2 -profile given in Example 4.5 turns to return probability

$$-\log\left(\mathfrak{q}^{(2t)}\left(e_{\Delta}\right)\right)\simeq t^{\frac{\beta+\iota}{3\beta+\iota}},$$

where the exponent can take any value in $(\frac{1}{3}, 1)$.

46

4.3. The case of dihedral groups. In this subsection we estimate decay of return probability of simple random walk on Δ where $\Gamma_s = D_{2l_s}$ are dihedral groups. We show that in this case the return exponent of simple random walks is $\frac{1}{3}$. Obtaining more precise estimates requires further work.

PROPOSITION 4.10. There exists an absolute constant C > 0 such that the following holds. Let Δ be the diagonal product constructed with $\Gamma_s = D_{2l_s}$ and parameters $\{k_s\}$ satisfying Assumption (2.11). Then

$$\frac{1}{C}n^{\frac{1}{3}} \le -\log \mathfrak{q}^{(2n)}(e_{\Delta}) \le Cn^{\frac{1}{3}}\log^{\frac{4}{3}}n.$$

Proof. Since Δ projects onto $\Delta_0 \simeq (\mathbb{Z}/2\mathbb{Z} \times \mathbb{Z}/2\mathbb{Z}) \wr \mathbb{Z}$, we have

$$\mathfrak{q}^{(2n)}\left(e_{\Delta}\right) \leq \mathfrak{q}_{0}^{(2n)}\left(e_{\Delta_{0}}\right).$$

The lower bound on $-\log \mathfrak{q}^{(2n)}(e_{\Delta})$ follows from the decay of return probability on Δ_0 (see [PSC02]),

$$\mathfrak{q}_{0}^{(2n)}\left(e_{\Delta_{0}}\right) \leq \exp\left(-\frac{1}{C}n^{\frac{1}{3}}\right)$$

In the other direction we construct a test function on Δ . First take a test function ψ_r on D_{2l_s} :

$$\psi_r(g) = \max\left\{1 - \frac{|g|_{D_{2l_s}}}{r}, 0\right\} \text{ for } 1 \le r \le l_s.$$

Recall that the set U_r^{Δ} is defined as $U_r^{\Delta} = \{((f_s), z) : \operatorname{Range}(f_s, z) \subset [-r, r]\}$. Let $S(r) = \{s : k_s \leq r, l_s \geq r^2\}$, and take

$$\Psi_r^{\Delta}\left((f_s), z\right) = \left(1 - \frac{|z|}{r}\right) \mathbf{1}_{U_r^{\Delta}}\left(\left((f_s), z\right)\right) \prod_{s \in \mathcal{S}(r)} \prod_{x \in [-r+k_s, r]} \psi_{r^2}(f_s(x)).$$

Depending on the sequences $(k_s), (l_s)$, the set S(r) might be empty, in which case we recover the test function of Proposition 4.4. (Recall the notation $\ell_s = \log |D_{2l_s}| = \log 2l_s$.) As in the proof of Proposition 4.4, we have

$$\frac{\sum_{Z \in \Delta} \left(\Psi_r^{\Delta}(Z) - \Psi_r^{\Delta}(Z\tau) \right)^2}{\|\Psi_r^{\Delta}\|_2^2} \le \frac{C_1}{r^2}.$$

By Cauchy-Schwarz inequality, we have

$$\frac{\sum_{Z \in \Delta} \left(\Psi_r^{\Delta}(Z) - \Psi_r^{\Delta}(Z\alpha) \right)^2}{\|\Psi_r^{\Delta}\|_2^2} \le |\mathcal{S}(r)| \sum_{s \in \mathcal{S}(r)} \frac{\sum_{g \in D_{2l_s}} \left(\psi_{r^2}(ga(s)) - \psi_{r^2}(g) \right)^2}{\|\psi_{r^2}\|_{l^2(D_{2l_s})}^2} \le C_1 |\mathcal{S}(r)|^2 r^{-4}.$$

The same estimates holds for β with a(s) replaced by b(s). Since $\{k_s\}$ satisfies the growth assumption (2.11), we have $|\mathcal{S}(r)| \leq \log_2 r$, and therefore

$$\frac{\mathcal{E}_{\Delta,\mathfrak{q}}\left(\Psi_{r}^{\Delta}\right)}{\left\|\Psi_{r}^{\Delta}\right\|_{2}^{2}} \leq C_{1}r^{-2}.$$

The support of function Ψ_r^{Δ} is bounded by

$$\left|\operatorname{supp}\Psi_{r}^{\Delta}\right| \leq (2r+1) \left(\prod_{s:k_{s} \leq r, l_{s} < r^{2}} (2l_{s})^{4r}\right) \left(\prod_{s:k_{s} \leq r, l_{s} > r^{2}} (2r^{2})^{4r}\right)$$
$$\leq (2r+1) (2r^{2})^{4r \log_{2} r}.$$

From these test functions we have

(19)
$$\Lambda_{2,\Delta,\mathfrak{q}}(v) \le C_1' \frac{(\log \log v)^4}{\log^2 v}$$

By the Coulhon-Grigor'yan theory, we conclude that there exists an absolute constant C,

$$\mathfrak{q}^{(2n)}\left(e_{\Delta}\right) \ge \exp\left(-Cn^{\frac{1}{3}}\log^{\frac{4}{3}}(2n)\right).$$

Remark 4.11. To get an estimate for $\Lambda_{p,\Delta,\mathfrak{q}}$, $p \in [1, 2]$, note that by projecting onto Δ_0 , we have

$$\Lambda_{1,\Delta,\mathfrak{q}}(v) \geq \Lambda_{1,\Delta,\mathfrak{q}_0}(v) \geq \frac{1}{C \log v},$$

and in the proof above we have an upper bound (19) for $\Lambda_{2,\Delta,\mathfrak{q}}$. By the Cheeger inequality (14), we have

$$\frac{1}{C_p} \frac{1}{\log^p v} \le \Lambda_{p,\Delta,\mathfrak{q}}(v) \le C_p \frac{(\log \log v)^{2p}}{\log^p v}.$$

5. Review: distortions of metric embeddings

We first recall the standard definition of distortion of a map between metric spaces. Given an injective map $f: X \to Y$ between two metric spaces (X, d_X) and (Y, d_Y) , the distortion of f measures quantitatively how far away f is from being a homothety,

distortion(f) =
$$\left(\sup_{u,v\in X, u\neq v} \frac{d_Y(f(u), f(v))}{d_X(u, v)}\right) \left(\sup_{u,v\in X, u\neq v} \frac{d_X(u, v)}{d_Y(f(u), f(v))}\right).$$

When f is C-Lipschitz, the first sup is bounded by C. In this case, the distortion is comparable to the second factor. It is the inverse of the expansion ratio, defined as

$$\operatorname{ratio}(f) = \inf_{u,v \in X, u \neq v} \frac{d_Y(f(u), f(v))}{d_X(u, v)}.$$

The smallest distortion with which X can be embedded in Y is denoted by $c_Y(X)$,

$$c_Y(X) = \inf \{ \operatorname{distortion}(f) : f : X \hookrightarrow Y \}.$$

To connect with uniform embedding of an infinite group G, it is well known that a sequence of finite metric spaces (X_k, d_k) embedded in the group G can provide obstruction for good embedding of the whole space; see, e.g., Arzhantseva-Drutu-Sapir [ADS09] and Austin [Aus11]. We quote a special case of a lemma in [Aus11].

LEMMA 5.1 (The Austin Lemma [Aus11]). Let \mathfrak{X} be a metric space. Let Γ be a finitely generated infinite group equipped with a finite generating set S, and let d denote the word distance on the Cayley graph (Γ, S) . Suppose that we can find a sequence of finite graphs (X_n, σ_n) , where σ_n is a 1-discrete metric on X_n , and embeddings $\vartheta_n : X_n \hookrightarrow \Gamma$ such that there are constants $C, L \ge 1$, $\delta > 0$ that are independent of n:

- diam $(X_n, \sigma_n) \to \infty$ as $n \to \infty$;
- there exists a sequence of positive reals $(r_n)_{n\geq 1}$ such that

$$\frac{1}{L}r_n\sigma_n(u,v) \le d\left(\vartheta_n(u),\vartheta_n(v)\right) \le Lr_n\sigma_n(u,v) \text{ for all } u,v \in X_n, n \ge 1$$

and moreover, $r_n \leq C \operatorname{diam}(X_n, \sigma_n)^{\beta}$ for all $n \geq 1$;

• distortion of (X_n, σ_n) into \mathfrak{X} is large in the sense that

$$c_{\mathfrak{X}}(X_n, \sigma_n) \ge \delta \operatorname{diam}(X_n, \sigma_n)^{\eta},$$

and then

$$\alpha_{\mathfrak{X}}^*(\Gamma, d) \le 1 - \frac{\eta}{1+\beta}.$$

The second assumption in Lemma 5.1 requires that under the embedding ϑ_k , the induced metric $d_{(G,S)}$ only dilates d_k with uniformly bounded distortion. This point-wise assumption is rather restrictive. In what follows we will present some bounds that are more flexible.

The term "Poincaré inequalities" in the context of metric embeddings was first systematically used in Linial-Magen-Naor [LMN02]. It is a key ingredient for many existing lower bounds for distortion of finite metric spaces. We review the basic idea now. Let $(\mathcal{M}, d_{\mathcal{M}})$ be a finite metric space, $\mathbf{a} = (a_{u,v}), \mathbf{b} = (b_{u,v}),$ where $u, v \in \mathcal{M}$ are two non-zero arrays of non-negative real numbers. A *p*-Poincaré type inequality for $f : \mathcal{M} \to \mathfrak{X}$ is an inequality of the form

(20)
$$\sum_{u,v\in\mathcal{M}} a_{u,v} d_{\mathfrak{X}}(f(u), f(v))^p \leq C \sum_{u,v\in\mathcal{M}} b_{u,v} d_{\mathfrak{X}}(f(u), f(v))^p.$$

The infimum of the constant C such that the inequality holds for all non-trivial $f: \mathcal{M} \to \mathfrak{X}$ is known as the \mathfrak{X} -valued Poincaré constant associated with \mathbf{a}, \mathbf{b} ,

$$P_{\mathbf{a},\mathbf{b},p}(\mathcal{M},\mathfrak{X}) = \sup \frac{\sum_{u,v} a_{u,v} d_{\mathfrak{X}}(f(u), f(v))^p}{\sum_{u,v} b_{u,v} d_{\mathfrak{X}}(f(u), f(v))^p},$$

where the sup is taken over all $f : \mathcal{M} \to \mathfrak{X}$ such that $\sum_{u,v} a_{u,v} d_{\mathfrak{X}}(f(u), f(v))^p \neq 0$. It follows from definition of the Poincaré constant that

$$\left(\inf_{u,v\in\mathcal{M},u\neq v}\frac{d_{\mathfrak{X}}(f(u),f(v))^{p}}{d_{\mathcal{M}}(u,v)^{p}}\right)\left(\sum_{u,v\in\mathcal{M}}a_{u,v}d_{\mathcal{M}}(u,v)^{p}\right)$$
$$\leq P_{\mathbf{a},\mathbf{b},p}(\mathcal{M},\mathfrak{X})\left(\sum_{u,v}b_{u,v}d_{\mathfrak{X}}(f(u),f(v))^{p}\right);$$

that is, the expansion ratio of f satisfies

$$\inf_{u,v\in\mathcal{M}, u\neq v} \frac{d_{\mathfrak{X}}(f(u), f(v))^p}{d_{\mathcal{M}}(u, v)^p} \le P_{\mathbf{a}, \mathbf{b}, p}(\mathcal{M}, \mathfrak{X}) \left(\frac{\sum_{u,v} b_{u,v} d_{\mathfrak{X}}(f(u), f(v))^p}{\sum_{u,v} a_{u,v} d_{\mathcal{M}}(u, v)^p}\right).$$

To relate to compression function, we need an extra ingredient that resembles a mass distribution assumption. We say that the array **a** satisfies the (p; l, c)-mass distribution condition if

(21)
$$\frac{\sum_{d_{\mathcal{M}}(u,v)\geq l} a_{u,v} d_{\mathcal{M}}(u,v)^{p}}{\sum_{d_{\mathcal{M}}(u,v)} a_{u,v} d_{\mathcal{M}}(u,v)^{p}} \geq c;$$

in other words, the *c*-fraction of the total **a** array sum is from vertices at least l apart. Under this additional assumption, for any $f : \mathcal{M} \to \mathfrak{X}$, there exists $u, v \in \mathcal{M}$ with $d_{\mathcal{M}}(u, v) \geq l$ such that

This compression upper bound is very useful. In practice, to apply this we need to choose the arrays \mathbf{a} , \mathbf{b} and obtain a good Poincaré inequality of the form (20). This is not an easy task in general. In what follows we review some special cases. These settings have been investigated extensively in the literature, thus established results are available for application to metric embeddings.

5.1. Poincaré inequalities in the classical form. Pioneered by work of Enflo [Enf69], it is well known that the spectral gap of certain Markov operators on a finite metric space (X, d) can be used to show the lower bound for distortion of the embedding of (X, d) into Hilbert spaces. This method appeared in Linial-Magen [LM00], Newman-Rabinovich [NR03] and was extended in Grigorchuk-Nowak [GN12], Jolissaint-Valette [JV14] and Mimura [Mim15]. Interested readers may also consult Chapter 13.5 in the book [LP16] for a nice introduction to this topic.

Let $(\mathcal{M}, d_{\mathcal{M}})$ be a finite metric space, and let $K : \mathcal{M} \times \mathcal{M} \to [0, 1]$ be a Markov transition kernel kernel on \mathcal{M} . Suppose K is reversible with respect to stationary distribution π . The most familiar Poincaré inequality for such a finite Markov chain takes the following form: for $f : \mathcal{M} \to \mathbb{R}$,

$$\sum_{u,v} |f(u) - f(v)|^2 \pi(u)\pi(v) \le C \sum_{u,v} |f(u) - f(v)|^2 K(u,v)\pi(u).$$

The reciprocal of the Poincaré constant is known as the spectral gap,

(23)
$$\lambda(K) = \inf_{f:\mathcal{M}\to\mathbb{R}, f\neq c} \left\{ \frac{\sum_{u,v\in\mathcal{M}} |f(u) - f(v)|^2 K(u,v)\pi(v)}{\sum_{u,v\in\mathcal{M}} |f(u) - f(v)|^2 \pi(u)\pi(v)} \right\}.$$

In this case the Poincaré constant is often referred to as the relaxation time of K. Mixing times of finite Markov chains have been a very active research area in the past decades. For a great variety of Markov chains, good estimates of their spectral gaps are known; examples can be found in [SC97], [LPW09]. Note that the same Poincaré inequality holds for Hilbert space valued functions $f : \mathcal{M} \to \mathcal{H}$. This fact can be checked by eigenbasis expansion. In some examples, based on the ℓ^2 -Poincaré inequality, one can apply Matoušek extrapolation (see [Mat97] and the version in [NS11]) to obtain useful Poincaré inequalities for ℓ^p -valued functions.

In the setting of inequality (20), having variance of f on the left side of the inequality and Dirichlet form on the right side corresponds to taking

(24)
$$a_{u,v} = \pi(u)\pi(v) \text{ and } b_{u,v} = \pi(u)K(u,v).$$

Define $\lambda_p(\mathcal{M}, K, \mathfrak{X})$ of the Markov operator K on Y to be

(25)
$$\lambda_p(\mathcal{M}, K, \mathfrak{X}) = \frac{1}{P_{\mathbf{a}, \mathbf{b}, p}(\mathcal{M}, \mathfrak{X})},$$

where \mathbf{a}, \mathbf{b} are specified by (24). When \mathfrak{X} is a Hilbert space and p = 2, this definition agrees with the standard variational formula of the spectral gap.

We now formulate an analogue of Lemma 5.1. Since the bound relies crucially on the \mathfrak{X} -valued Poincaré constants $1/\lambda_p(X_n, K_n, \mathfrak{X})$ of the Markov operator K_n on X_n , we refer to it as the spectral method for bounding compression functions.

LEMMA 5.2. Let G be an infinite group equipped with a metric d and $p \in [1, \infty)$. Let X_n be a sequence of finite subsets in G and K_n be reversible Markov kernels on X_n with stationary distribution π_n . Suppose there exist a constant $c \in (0, 1)$ and an increasing sequence $\{l_n\}$ such that the array \mathbf{a}_n defined as $\mathbf{a}_n(u, v) = \pi(u)\pi(v)$ satisfies the $(p; l_n, c)$ -mass distribution condition (21).

Let $f: G \to \mathfrak{X}$ be a 1-Lipschitz uniform embedding. Then the compression function of f satisfies

$$\rho_f(l_n) \le \operatorname{diam}_d(X_n) \left(\frac{1}{\lambda_p(X_n, K_n, \mathfrak{X})} \left(\frac{\sum_{u, v \in X_n} d_{\mathfrak{X}}(f(u), f(v))^p K_n(u, v) \pi_n(v)}{c \sum_{u, v \in X_n} d(u, v)^p \pi(u) \pi(v)} \right) \right)^{\frac{1}{p}}.$$

Proof. Equip X_n with the metric induced by the metric d on G; the inequality follows from (22).

Example 5.3. Consider the special case where X_n is a sequence of finite subgroups in G and d is a left invariant metric on G, e.g., the word metric. Take μ_n to be a symmetric probability measure on X_n and $K_n(u, v) = \mu_n (u^{-1}v)$. Then the Markov chain with transition kernel K_n is the random walk on X_n with step distribution μ_n . It is reversible with respect to the uniform distribution U_n on X_n . In this case, because of transitivity, the mass distribution condition is easily satisfied, namely,

$$\sum_{v: \ d(u,v) \ge \frac{1}{2} \ \text{diam}_d(X_n)} U_n(v) \ge \frac{1}{2} \ \text{ for every } u \in X_n.$$

It follows that $\mathbf{a}_n = (U_n(u)U_n(v))$ satisfies the $(p, \frac{1}{2} \operatorname{diam}_d(X_n), \frac{1}{2})$ -mass distribution condition, and the bound in Lemma 5.2 simplifies to

$$\rho_f\left(\frac{\operatorname{diam}_d(X_n)}{2}\right) \le \left(\frac{2^{p+2}\sum_{u,v\in X_n} d_{\mathfrak{X}}(f(u),f(v))^p K_n(u,v)\pi_n(v)}{\lambda_p(X_n,K_n,\mathfrak{X})}\right)^{\frac{1}{p}}.$$

5.2. Markov type inequalities . The notion of the Markov type of a metric space was introduced by K. Ball in [Bal92]. It has found important applications in metric geometry. In [LMN02], Linal, Magen and Naor pointed out that the basic assumption of this concept can be viewed as Poincaré inequalities. The Markov type method for bounding a compression exponent was first introduced by Naor and Peres in [NP08] and later significantly extended in [NP11].

Definition 5.4 (K. Ball [Bal92]). Given a metric space $(\mathfrak{X}, d_{\mathfrak{X}})$ and $p \in [1, \infty)$, we say that \mathfrak{X} has Markov-type p if there exists a constant C > 0 such that for every stationary reversible Markov chain $\{Z_t\}_{t=0}^{\infty}$ on $\{1, \ldots, n\}$, every mapping $f : \{1, \ldots, n\} \to \mathfrak{X}$ and every time $t \in \mathbb{N}$,

(26)
$$\mathbf{E}d_{\mathfrak{X}}(f(Z_t), f(Z_0))^p \le C^p t \mathbf{E}d_{\mathfrak{X}}(f(Z_1), f(Z_0))^p.$$

The least such constant C is called the Markov-type p constant of \mathfrak{X} and is denoted by $M_p(\mathfrak{X})$.

Theorem 2.3 in Naor-Peres-Sheffield-Schramm [NPSS06] implies the following results for the classical Lebesgue spaces L_p . For $p \in (1, 2]$, the space L_p has Markov type p and $M_p(L_p) \leq \frac{8}{(2^{p+1}-4)^{1/p}}$; and for every $p \in [2, \infty)$, L_p has Markov type 2 and $M_2(L_p) \leq 4(p-1)^{\frac{1}{2}}$. See [NPSS06] for more examples of metric spaces of known Markov type.

In the setting of (20), the inequality (26) in the definition of Markov type p can be viewed as a Poincaré inequality with

$$a_{u,v} = K^t(u, v)\pi(u)$$
 and $b_{u,v} = K(u, v)\pi(u)$

where K is the transition kernel of a reversible Markov chain on state space \mathcal{M} of n points, and π is its stationary distribution. The Poincaré inequality provided by (26) reads

$$\sum_{u,v\in\mathcal{M}} d_{\mathfrak{X}}(f(u),f(v))^p K^t(u,v)\pi(u) \le M_p^p\left(\mathfrak{X}\right) t \sum_{u,v\in\mathcal{M}} d_{\mathfrak{X}}(f(u),f(v))K(u,v)\pi(u)$$

for all functions $f : \mathcal{M} \to \mathfrak{X}$. Note that the notion of Markov type is very powerful. If \mathfrak{X} has Markov type p, then the inequality above is valid for any finite state space \mathcal{M} and any reversible Markov transition kernel K on \mathcal{M} .

Now we examine the mass distribution condition. Let $(\mathcal{M}, d_{\mathcal{M}})$ be a finite metric space, and let K be a reversible Markov kernel on \mathcal{M} with stationary distribution π . Let $\{Z_t\}_{t=0}^{\infty}$ be a stationary Markov chain on \mathcal{M} with transition kernel K. At time t, set

$$\gamma(t)^p = \frac{1}{2} \mathbf{E}_{\pi} \left[d_{\mathcal{M}}(Z_t, Z_0)^p \right].$$

Then

$$\begin{aligned} \mathbf{E}_{\pi} [d_{\mathcal{M}}(Z_{t}, Z_{0})^{p} \mathbf{1}_{\{d_{\mathcal{M}}(Z_{t}, Z_{0}) > \gamma(t)\}}] \\ &= \mathbf{E}_{\pi} \left[d_{\mathcal{M}}(Z_{t}, Z_{0})^{p} \right] - \mathbf{E}_{\pi} \left[d_{\mathcal{M}}(Z_{t}, Z_{0})^{p} \mathbf{1}_{\{d_{Y}(Z_{t}, Z_{0}) \leq \gamma(t)\}} \right] \\ &\geq \mathbf{E}_{\pi} \left[d_{\mathcal{M}}(Z_{t}, Z_{0})^{p} \right] - \gamma(t)^{p} = \frac{1}{2} \mathbf{E}_{\pi} \left[d_{\mathcal{M}}(Z_{t}, Z_{0})^{p} \right]. \end{aligned}$$

That is, the array **a** satisfies the $\left(p; \left(\frac{1}{2}\mathbf{E}_{\pi} \left[d_{\mathcal{M}}(Z_t, Z_0)^p\right]\right)^{\frac{1}{p}}, \frac{1}{2}\right)$ -mass distribution condition, where **a** is defined by $a_{u,v} = K^t(u, v)\pi(u)$. From the inequality (22) we derive the following upper bound on compression function.

LEMMA 5.5. Let G be an infinite group equipped with a metric d. Let $f: G \to \mathfrak{X}$ be a 1-Lipschitz uniform embedding. Assume that \mathfrak{X} has Markov type p.

Let X_n be a sequence of finite sets of G, and let K_n be a reversible Markov kernel on X_n with stationary distribution π_n . Let $\{Z_t^{(n)}\}_{t=0}^{\infty}$ be a stationary Markov chain on X_n with transition kernel K_n . Then for any $t_n \in \mathbb{N}$, the compression function of f satisfies

$$\rho_f\left(\left(\frac{1}{2}\mathbf{E}_{\pi_n}\left[d\left(Z_{t_n}^{(n)}, Z_0^{(n)}\right)^p\right]\right)^{\frac{1}{p}}\right)$$

$$\leq \left(2M_p^p(\mathfrak{X})t_n \operatorname{diam}_{(G,d)}(X_n)^p \frac{\mathbf{E}_{\pi_n}\left[d_{\mathfrak{X}}\left(f\left(Z_1^{(n)}\right), f\left(Z_0^{(n)}\right)\right)^p\right]}{\mathbf{E}_{\pi_n}\left[d\left(Z_{t_n}^{(n)}, Z_0^{(n)}\right)^p\right]}\right)^{\frac{1}{p}}.$$

Remark 5.6. This upper bound on the compression function is in the same spirit as the argument of Naor and Peres in Section 5 of [NP11]. The difference is that in [NP11] the authors considered random walks on the infinite group Gstarting at identity and f is taken to be a 1-cocycle on G. Then the Markov type inequality for 1-cocycles was applied to bound the compression function. One restriction for such an approach is that the step distribution of the random walk needs to have finite p-moment. While in the finite subsets, in principle one can experiment with any reversible transition kernel and choose the best one available. Examples that illustrate this point can be found in Section 7.1.

5.3. Comparing spectral and Markov type methods. It is interesting to compare the classical Poincaré inequalities and the ones from Markov type method. Suppose in the infinite group G that we have chosen a sequence of subsets $\{X_n\}$ and reversible Markov kernels K_n on X_n . With this sequence $\{(X_n, K_n)\}$ we compare the results given by the two methods. Let \mathfrak{X} be a metric space of Markov type p and $f : G \to \mathfrak{X}$ be a 1-Lipschitz embedding from (G, d) to $(\mathfrak{X}, d_{\mathfrak{X}})$. To compare terms in the bounds of Lemmas 5.2 and 5.5, first note that

$$\mathbf{E}d_{\mathfrak{X}}\left(f\left(Z_{1}^{(n)}\right), f\left(Z_{0}^{(n)}\right)\right)^{p} = \sum_{u,v \in X_{n}} d_{\mathfrak{X}}(f(u), f(v))^{p} K_{n}(u, v) \pi_{n}(u).$$

Now we choose t_n to be comparable to the Poincaré constant $P_p(X_n, K_n, \mathfrak{X})$. (It corresponds to relaxation time when \mathfrak{X} is a Hilbert space and p = 2.) Suppose in addition that π_n satisfies the $(p; \theta \operatorname{diam}(X_n), c)$ -mass distribution condition. Then essentially the difference in the two bounds comes from the ratio

$$\frac{\operatorname{diam}_{(G,d)}(X_n)^p}{\operatorname{\mathbf{E}} d\left(Z_{t_n}^{(n)}, Z_0^{(n)}\right)^p}.$$

Thus if there is a constant $c_1 > 0$ such that for $t_n \simeq P_p(X_n, K_n, \mathfrak{X})$,

$$\mathbf{E}d_{X_n}\left(Z_{t_n}^{(n)}, Z_0^{(n)}\right)^p \ge c_1^p \operatorname{diam}_{(G,d)}(X_n)^p,$$

then up to some multiplicative constants, the two methods give the same compression upper bound.

It is important in applications that the choice of the sequence of finite subsets X_n and Markov kernels K_n is flexible. For example, in order to use Poincaré inequalities to obtain an upper bound on the compression function of uniform embedding f from G into a Hilbert space, the subsets X_n should be chosen to capture some worst distorted elements in the group under f, and the Markov kernel K_n on X_n should be chosen so that

$$\frac{1}{\lambda(K_n)} \left(\sum_{u,v \in X_n} d_{\mathcal{H}}(f(u), f(v))^2 K_n(u, v) \pi_n(v) \right)$$

is as small as possible. That is, K_n needs to achieve a balance between spectral gap and Dirichlet form $\mathcal{E}_{K_n}(f)$. This point will be the guideline for the choice of (X_n, K_n) in the examples we treat.

5.4. Metric cotype inequalities. The notion of type and cotype plays a central role in the local theory of Banach spaces. The classical linear notion of type and cotype is defined as follows. A Banach space \mathfrak{X} is said to have (Rademacher) type p > 0 if there exists a constant T > 0 such that for every n and every $x_1, \ldots, x_n \in \mathfrak{X}$,

$$\mathbb{E} \left\| \sum_{j=1}^{n} \varepsilon_j x_j \right\|_{\mathfrak{X}}^p \leq T^p \sum_{j=1}^{n} \|x_j\|_{\mathfrak{X}}^p,$$

where \mathbb{E} is the expectation with respect to uniform distribution on $(\varepsilon_1, \ldots, \varepsilon_n) \in \{-1, 1\}^n$. A Banach space \mathfrak{X} is said to have (Rademacher) cotype q > 0 if there exists a constant C > 0 such that for every n and every $x_1, \ldots, x_n \in \mathfrak{X}$,

$$\mathbb{E} \left\| \sum_{j=1}^{n} \varepsilon_{j} x_{j} \right\|_{\mathfrak{X}}^{q} \geq \frac{1}{C^{q}} \sum_{j=1}^{n} \|x_{j}\|_{\mathfrak{X}}^{q}.$$

Given a Banach space \mathfrak{X} , define

 $p_{\mathfrak{X}} = \sup \{ p : \mathfrak{X} \text{ has type } p \}, \ q_{\mathfrak{X}} = \inf \{ q : \mathfrak{X} \text{ has cotype } q \}.$

The space \mathfrak{X} is said to be of non-trivial type if $p_{\mathfrak{X}} > 1$, and it is of non-trivial cotype if $q_{\mathfrak{X}} < \infty$.

Mendel and Naor [MN08] introduced the non-linear notion of metric cotype. By [MN08, Def. 1.1], $(\mathfrak{X}, d_{\mathfrak{X}})$ has *metric cotype* q with constant Γ if for every integer $n \in \mathbb{N}$, there exists an even integer m, such that for every $f: \mathbb{Z}_m^n \to \mathfrak{X}$,

$$\sum_{j=1}^{n} \sum_{u \in \mathbb{Z}_{m}^{n}} d_{\mathfrak{X}} \left(f\left(u + \frac{m}{2} \mathbf{e}_{j}\right), f(u) \right)^{q} \pi(u) \\ \leq \Gamma^{q} m^{q} \sum_{u \in \mathbb{Z}_{m}^{n}} \mathbb{E} \left[d_{\mathfrak{X}} \left(f\left(u + \varepsilon\right), f(u) \right)^{q} \right] \pi(u),$$

where π is the uniform distribution on \mathbb{Z}_m^n and \mathbb{E} is the expectation taken with respect to uniform distribution on $\boldsymbol{\varepsilon} = (\varepsilon_1, \ldots, \varepsilon_n) \in \{-1, 0, 1\}^n$, and $\{\mathbf{e}_j\}$ is the standard basis of \mathbb{R}^n . Mendel and Naor proved in [MN08] that for a Banach space \mathfrak{X} and $q \in [2, \infty)$, \mathfrak{X} has metric cotype q if and only if it has Rademacher cotype q. As a key step, they established the following sharp estimate, which we will refer to as the metric cotype inequality.

THEOREM 5.7 ([MN08, Th. 4.2]). Let π be the uniform distribution on \mathbb{Z}_m^n and σ be the uniform distribution on $\{-1, 0, 1\}^n$. Let \mathfrak{X} be a Banach space of Rademacher type p > 1 and cotype $q \in [2, \infty)$. Then for every $f : \mathbb{Z}_m^n \to \mathfrak{X}$,

$$\sum_{u \in \mathbb{Z}_m^n} \sum_{j=1}^n d_{\mathfrak{X}} \left(f\left(u + \frac{m}{2} \mathbf{e}_j\right), f(u) \right)^q \pi(u) \\ \leq \left(5 \max\left\{ C(\mathfrak{X})m, n^{\frac{1}{q}} \right\} \right)^q \sum_{u \in \mathbb{Z}_m^n} \sum_{\boldsymbol{\varepsilon} \in \{-1, 0, 1\}^n} d_{\mathfrak{X}} \left(f(u + \boldsymbol{\varepsilon}), f(u) \right)^q \sigma(\boldsymbol{\varepsilon}) \pi(u),$$

where $C(\mathfrak{X}) > 0$ is a constant that only depends on the cotype constant and K_q -convexity constant of \mathfrak{X} .

For our purposes, the Mendal-Naor metric cotype inequality can be viewed as a Poincaré inequality with a rather unusual choice of arrays \mathbf{a}, \mathbf{b} on \mathbb{Z}_m^n , namely,

$$a_{u,v} = \sum_{j=1}^{n} \pi(u) \mathbf{1}_{\left\{v=u+\frac{m}{2}\mathbf{e}_{j}\right\}} \text{ and } b_{u,v} = \sum_{\boldsymbol{\varepsilon} \in \{-1,0,1\}^{n}} \pi(u) \mathbf{1}_{\left\{v=u+\boldsymbol{\varepsilon}\right\}} \sigma(\boldsymbol{\varepsilon}).$$

Then the Poincaré constant is bounded by

$$P_{\mathbf{a},\mathbf{b},2}\left(\mathbb{Z}_{m}^{n},\mathfrak{X}\right) \leq \left(5 \max\left\{C(\mathfrak{X})mn^{\frac{1}{2}-\frac{1}{q}},n^{\frac{1}{2}}\right\}\right)^{2}.$$

It captures a subtle comparison between a transition kernel that moves far in one fiber and another kernel that moves by ± 1 across the whole product.

Among many other applications of metric cotype, such inequalities provide sharp lower bound for the distortion of embeddings of the ℓ^{∞} -integer lattice $[m]_{\infty}^{n}$ into Banach spaces of non-trivial type and cotype q; see Theorem 1.12 in [MN08]. In Section 8 we will apply these metric cotype inequalities in the study of compression of diagonal product Δ constructed with dihedral groups, exactly because of the presence of l^{∞} -lattices of growing side length in the group.

6. Compression of Δ with embedded expanders

In this section we consider compression of the diagonal product Δ constructed with $\{\Gamma_s\}$ chosen to be certain families of expanders. Let \mathfrak{X} be a Banach space. A map $\Psi : G \to \mathfrak{X}$ is called *G*-equivariant if there exists an action τ of *G* on \mathfrak{X} by affine isometries and a vector $v \in \mathfrak{X}$ such that $\Psi(g) = \tau(g)v$ for all $x \in G$. Such a map is called a 1-cocycle; see [dCTV07].

A couple of functions (g_1, g_2) is an equivariant- \mathfrak{X} -compression gap of G if any 1-Lipschitz G-equivariant embedding $\varphi: G \to \mathfrak{X}$ satisfies $\rho_{\varphi}(t) \leq g_2(t)$ for all $t \geq 1$ and there exists a 1-Lipschitz *G*-equivariant embedding $\Psi : G \to \mathfrak{X}$ such that $\rho_{\Psi}(t) \geq g_1(t)$ for all $t \geq 1$.

We address the question regarding possible L_p -compression exponents of finitely generated amenable groups.

PROPOSITION 6.1. For any $\gamma \in [0,1]$, there exists a finitely generated elementary amenable group Δ such that for all $p \geq 1$,

$$\alpha_p^{\#}(\Delta) = \gamma.$$

This result follows from a more precise result about the equivariant compression gap of the diagonal product group Δ ; see Theorem 6.11. We will see that when the lamp groups $\{\Gamma_s\}$ are chosen to be expanders, single copies of these lamp groups provide sufficient obstruction for embedding. In some sense this case can be viewed as an amenable analogue of [ADS09].

6.1. An upper bound in any uniformly convex Banach spaces. In this subsection we take $\{\Gamma_s\}$ as a subsequence in the Lafforgue super expanders $\{\Gamma_m\}$ described in Example 2.3. By Fact 2.15, each group Γ_s embeds homothetically in the diagonal product Δ with ratio $k_s + 1$, i.e., there is group homomorphism $\vartheta_s: \Gamma_s \to \Delta$ satisfying

$$\left|\vartheta_s(\gamma)\right|_{\Delta} = \left(k_s + 1\right) \left|\gamma\right|_{\Gamma_s}.$$

From these distortion estimates and the embeddings $\vartheta_s : \Gamma_s \hookrightarrow \Delta$, we immediately derive an upper bound on the compression function of Δ into \mathfrak{X} by Lemma 5.2.

LEMMA 6.2. Let Δ be the diagonal product with parameters (k_s) and lamp groups $\{\Gamma_s\}$ chosen as a subsequence of Lafforgue super expanders in Example 2.3, diam $(\Gamma_s) = l_s$. Then for any uniformly convex Banach space \mathfrak{X} , there exists a constant $\delta = \delta(\Gamma, \mathfrak{X}, |A| + |B|) > 0$ such that the compression function of any 1-Lipschitz embedding $\Psi : \Delta \to \mathfrak{X}$ satisfies

$$\rho_{\Psi}\left(\frac{1}{2}(k_s+1)l_s\right) \le 4\delta^{-\frac{1}{2}}(k_s+1).$$

Proof. Take $X_s = \vartheta_s(\Gamma_s)$ and $K_s(u, v) = \nu_s(\vartheta_s^{-1}(u^{-1}v))$, where ν_s is uniform on the generating set $A(s) \cup B(s)$. To apply Lemma 5.2, note that $\operatorname{diam}_{d_{\Delta}}(X_s) = (k_s + 1)l_s$, the Poincaré constant $P_2(X_s, K_s, \mathfrak{X}) \leq 1/\delta$ by Lafforgue's result (1), where δ is a constant only depending on Γ, \mathfrak{X} and |A| + |B|. Since Ψ is 1-Lipschitz with respect to $|\cdot|_{\Delta}$,

$$\sum_{u,v \in X_s} d_{\mathfrak{X}}(\Psi(u), \Psi(v))^2 K_s(u, v) \pi_s(u) \le (k_s + 1)^2.$$

Since π_s is the uniform distribution on the subgroup X_s , the upper bound on ρ_{Ψ} then follows from the Poincaré inequalities (1) in Example 5.3 with p = 2. \Box

6.2. Compression gaps of embeddings of Δ into L_p . In this subsection we focus on the case where L_p , $p \geq 1$, are target spaces for embedding.

6.2.1. Upper bound when $\{\Gamma_s\}$ are expanders. When the target space is $L_p, p \geq 1$, a more precise piecewise upper bound of the compression gap can be obtained. Recall that a symmetric probability measure μ on a group G defines a Markov transition kernel $K(u, v) = \mu(u^{-1}v)$ that is reversible with respect to the uniform distribution on G. Its ℓ^2 -spectral gap $\lambda(G, \mu) = \lambda(G, K)$ is defined as in (23).

PROPOSITION 6.3. Let Δ be the diagonal product with parameters (k_s) and lamp groups $\{\Gamma_s\}$ expanders where diam $(\Gamma_s) = l_s < \infty$. Suppose $\{\Gamma_s\}$ satisfies Assumption 2.1 and

 $\lambda(\Gamma_s,\nu_s) \geq \delta > 0$ for all s with $l_s < \infty$,

where ν_s is uniform on $A(s) \cup B(s)$. Then there exists a constant C_0 depending only on |A|, |B| such that for any 1-Lipschitz embedding $\Psi : \Delta \to L_p$, the compression function of Ψ satisfies, for all $s \ge 1$ with $k_s, l_s < \infty$,

(27)
$$\rho_{\Psi}\left(\frac{1}{2}x\right) \leq C(\delta,p)\frac{x}{l_s} \text{ if } x \in [k_s l_s, k_{s+1} l_s],$$

$$where \ C(\delta,p) = \begin{cases} C_0 \delta^{-\frac{1}{p}} & \text{if } 1 \leq p \leq 2, \\ C_0 p \delta^{-\frac{1}{2}} & \text{if } p > 2. \end{cases}$$

Remark 6.4. If $k_{s+1}, l_{s+1} < \infty$, then by monotonicity of the compression function, the bound extends to the interval $[k_{s+1}l_s, k_{s+1}l_{s+1}]$; namely, for $x \in [k_{s+1}l_s, k_{s+1}l_{s+1}]$,

$$\rho_{\Psi}\left(\frac{1}{2}x\right) \le \rho_{\Psi}\left(\frac{1}{2}k_{s+1}l_{s+1}\right) \le C(\delta, p)k_{s+1}.$$

If $l_{s+1} = \infty$, the situation is different. We need to have information regarding compression of the infinite group Γ ; see Lemma 6.6.

Proof. Consider the subgroup

$$\Gamma'_s = [A(s), B(s)]^{\Gamma_s} = \ker \left(\Gamma_s \to A(s) \times B(s) \right)$$

Take the symmetric generating set R(s) for Γ'_s using the Reidemeister-Schreier algorithm in Lemma 2.18, where $F = A(s) \times B(s)$, $S = A(s) \cup B(s)$. Then the inclusion map from $(\Gamma'_s, R(s))$ into $(\Gamma_s, A(s) \cup B(s))$ is bi-Lipschitz, $|\gamma|_{R(s)} \leq$ $|\gamma|_{\Gamma_s} \leq 5|\gamma|_R$ for all $\gamma \in \Gamma'_s$. Let μ_s be the uniform distribution on R(s). It is known that if there is a (C, C)-quasi isometric map $\psi : (G, S) \to (H, T)$ and if the image $\psi(G)$ is R dense in H, then the Poincaré constant of (H, ν) is comparable to the Poincaré constant of (G, μ) with constants only depending on C, R, |S|, |T|, where μ (resp. ν) is the uniform distribution on $S \cup S^{-1}$ (resp. $T \cup T^{-1}$); see the proof of [CSC95, Prop. 4.2] or [PSC00, Th. 1.2]. In the current situation, since the inclusion map $(\Gamma'_s, R(s))$ into $(\Gamma_s, A(s) \cup B(s))$ is a (5,5)-quasi-isometry and Γ'_s is 2-dense in Γ_s , there exists a constant c_0 only depending on |A| and |B| such that the spectral gap of μ_s satisfies

$$\lambda(\Gamma'_s, \mu_s) = \tilde{\delta} \ge c_0 \delta.$$

Let $t \in [k_s, k_{s+1}]$. Consider the direct product Π_s^t of t copies of Γ_s' in the factor Δ_s at site $0, 1, \ldots, t-1$. By Section 2.3, Π_s^t is an embedded subgroup of Δ . Denote such an embedding by $\theta_s : \Pi_s^t \hookrightarrow \Delta$. On Π_s^t , take the product kernel $\zeta_t = (\mu_s)_0 \otimes \cdots \otimes (\mu_s)_{t-1}$. By the tensorizing property of classical Poincaré inequalities, we have that for any function $f : \Pi_t \to \mathbb{R}$,

$$\sum_{u,v\in\Pi_t} |f(u) - f(v)|^2 \pi_{\Pi_s^t}(u) \pi_{\Pi_s^t}(v) \le \tilde{\delta}^{-1} \sum_{u,v\in\Pi_t} |f(u) - f(v)|^2 \pi_{\Pi_t}(u) \zeta_t \left(u^{-1}v\right).$$

By Matoušek's extrapolation lemma for Poincaré inequalities [Mat97] (see the version in [NS11, Lemma 4.4]), it follows that for any $f : \Pi_t \to \ell^p$,

• if $1 \le p \le 2$, then

$$\sum_{u,v\in\Pi_{s}^{t}} \|f(u) - f(v)\|_{p}^{p} \pi_{\Pi_{s}^{t}}(u) \pi_{\Pi_{s}^{t}}(v) \\ \leq \tilde{\delta}^{-1} \sum_{u,v\in\Pi^{t}} \|f(u) - f(v)\|_{p}^{p} \pi_{\Pi_{s}^{t}}(u) \zeta_{t}\left(u^{-1}v\right);$$

• if p > 2, then

$$\sum_{u,v\in\Pi_s^t} \|f(u) - f(v)\|_p^p \pi_{\Pi_s^t}(u) \pi_{\Pi_s^t}(v) \\ \leq (2p)^p \tilde{\delta}^{-p/2} \sum_{u,v\in\Pi_s^t} \|f(u) - f(v)\|_p^p \pi_{\Pi_s^t}(u) \zeta_t \left(u^{-1}v\right).$$

Let $\varphi : \Delta \to \ell^p$ be a 1-Lipschitz uniform embedding of Δ with respect to the word metric $|\cdot|_{\Delta}$. Apply Lemma 5.2 to the subset $\theta_s(\Pi_t)$ equipped with the kernel $\zeta_t \circ \theta_s^{-1}$. With the mass distribution condition satisfied by Lemma 2.16, we have

$$\rho_{\varphi}\left(\frac{1}{2}tl_s\right) \le c(\tilde{\delta}, p)t_s$$

where the constant $c(\tilde{\delta}, p)$ is given by

(28)
$$c(\tilde{\delta}, p) = \begin{cases} 4(2/\tilde{\delta})^{\frac{1}{p}} & \text{if } 1 \le p \le 2, \\ 4 \cdot 2^{1+\frac{1}{p}} \tilde{\delta}^{-\frac{1}{2}} p & \text{if } p > 2. \end{cases}$$

From the standard fact that L_p is $(1 + \varepsilon)$ -finitely presentable in l_p (see, for example, the proof of [JV14, Th. 1.1]), we conclude that for for any 1-Lipschitz

uniform embedding $\Psi : \Delta \to L_p$, the same bound holds:

$$\rho_{\Psi}\left(\frac{1}{2}tl_s\right) \le c(\tilde{\delta}, p)t.$$

6.2.2. Upper bound with an infinite group Γ_s having strong property (T). Next we consider the case where $\Gamma_s = \Gamma$ is an infinite group. (It corresponds to $l_s = \infty$.) Let Γ be a discrete group equipped with a finite generating set S, and let \mathfrak{X} be a Banach space. A linear isometric Γ -representation on \mathfrak{X} is a homomorphism $\varrho: \Gamma \to O(\mathfrak{X})$, where $O(\mathfrak{X})$ denotes the groups of all invertible linear isometries of \mathfrak{X} . Denote by $\mathfrak{X}^{\varrho(\Gamma)}$ the closed subspace of Γ -fixed vectors. When \mathfrak{X} is uniformly convex, by [BFGM07, Prop. 2.6] the subspace of $\mathfrak{X}^{\varrho(\Gamma)}$ is complemented in $\mathfrak{X}, \mathfrak{X} = \mathfrak{X}^{\varrho(\Gamma)} \oplus \mathfrak{X}'(\varrho)$, and the decomposition is canonical.

Definition 6.5. Let Γ be a discrete group equipped with a finite generating set S, and let \mathfrak{X} be a uniformly convex Banach space.

- Following [BFGM07], we say that Γ has Property $(F_{\mathfrak{X}})$ if any action of Γ on \mathfrak{X} by affine isometries has a Γ -fixed point.
- We say Γ has Property $(T_{\mathfrak{X}})$ if there exists a constant $\varepsilon > 0$ such that for any representation $\varrho : \Gamma \to O(\mathfrak{X})$,

$$\max_{s \in S} \|\varrho(s)v - v\|_{\mathfrak{X}} \ge \varepsilon \, \|v\|_{\mathfrak{X}} \quad \text{for all } v \in \mathfrak{X}'(\varrho).$$

The maximal ε with this property is called the \mathfrak{X} -Kazhdan constant of Γ with respect to S and is denoted by $\kappa_{\mathfrak{X}}(\Gamma, S)$.

By [BFGM07, Th. 1.3 and Rem. 2.28], Property $(F_{\mathfrak{X}})$ implies Property $(T_{\mathfrak{X}})$ in any uniformly convex Banach space \mathfrak{X} with uniformly convex dual.

LEMMA 6.6. Let Δ be the diagonal product with parameters $(k_s)_{s \leq \mathfrak{s}_0}$ and lamp groups $\{\Gamma_s\}_{s \leq \mathfrak{s}_0}$, where $\Gamma_{\mathfrak{s}_0} = \Gamma$ is an infinite group marked with generating subgroups A, B. Suppose \mathfrak{X} is a uniformly convex Banach space and Γ has Property $(F_{\mathfrak{X}})$. Then for any equivariant 1-Lipschitz embedding $\Psi : \Delta \to \mathfrak{X}$, the compression function of Ψ satisfies

$$\rho_{\Psi}\left(x\right) \leq \frac{2}{\kappa_{\mathfrak{X}}(\Gamma, A \cup B)} \left(k_{\mathfrak{s}_{0}} + 1\right) \text{ for all } x \in \left[k_{\mathfrak{s}_{0}} + 1, \infty\right].$$

Proof. Since the embedding $\vartheta_{\mathfrak{s}_0}: \Gamma \hookrightarrow \Delta$ is a homothety with $|\vartheta_{\mathfrak{s}_0}(\gamma)|_{\Delta} = (k_{\mathfrak{s}_0}+1) |\gamma|_{\Gamma}, \psi = \Psi \circ \vartheta_{\mathfrak{s}_0}: \Gamma \to \mathfrak{X}$ is a $(k_{\mathfrak{s}_0}+1)$ -Lipschitz equivariant embedding. Consider $\tilde{\psi} = \frac{\psi}{k_{\mathfrak{s}_0}+1}$. Since $\tilde{\psi}$ equivariant, it is a 1-cocycle with respect to some representation $\varrho: \Gamma \to O(\mathfrak{X})$. Since Γ has Property $(F_{\mathfrak{X}}), H^1(\Gamma, \varrho) = Z^1(\Gamma, \varrho)/B^1(\Gamma, \varrho)$ vanishes. It follows that $\tilde{\psi}$ is a 1-coboundary; that is, there exists $v \in \mathfrak{X}$ such that

$$\tilde{\psi}(g) = \varrho(g)v - v.$$

We may take v in the complement $\mathfrak{X}'(\varrho)$. Then by Property $(T_{\mathfrak{X}})$,

$$\max_{s \in A \cup B} \|\varrho(s)v - v\|_{\mathfrak{X}} \ge \kappa \|v\|_{\mathfrak{X}}, \text{ where } \kappa = \kappa_{\mathfrak{X}}(\Gamma, A \cup B).$$

Since $\tilde{\psi}$ is 1-Lipschitz, we have $\kappa \|v\|_{\mathfrak{X}} \leq \max_{s \in A \cup B} \|\tilde{\psi}(s)\|_{\mathfrak{X}} \leq 1$. It follows that for any $g \in \Gamma$,

$$\left\|\tilde{\psi}(g)\right\|_{\mathfrak{X}} = \|\varrho(g)v - v\|_{\mathfrak{X}} \le \|\varrho(g)v\|_{\mathfrak{X}} + \|v\|_{\mathfrak{X}} = 2 \|v\|_{\mathfrak{X}} \le 2/\kappa.$$

Now we get back to Ψ . Since $\Psi \circ \vartheta_{\mathfrak{s}_0} = (k_{\mathfrak{s}_0} + 1)\tilde{\psi}$ and $|\vartheta_{\mathfrak{s}_0}(\gamma)|_{\Delta} = (k_{\mathfrak{s}_0} + 1)|\gamma|_{\Gamma}$, we deduce from $\|\tilde{\psi}(g)\|_{\mathfrak{r}} \leq 2/\kappa$ that

$$\rho_{\Psi}(x) \le \frac{2}{\kappa_{\mathfrak{X}}(\Gamma, A \cup B)} \left(k_{\mathfrak{s}_0} + 1\right) \text{ for all } x \in [k_{\mathfrak{s}_0} + 1, \infty].$$

Remark 6.7. In practice, we use the bound in Lemma 6.6 for the interval $[(k_{\mathfrak{s}_0}+1) l_{\mathfrak{s}_0-1},\infty]$, because for smaller length x, the copies Γ_s with $s \leq \mathfrak{s}_0 - 1$ provide better upper bounds.

Property $(F_{\mathfrak{X}})$ is very strong. By Bader-Furman-Gelander-Monod (see [BFGM07, Th. B]) and standard Hereditary properties ([BdlHV08, §2.5]), the lattice Γ in Example 2.3 has Property F_{L_p} for all 1 .

When we specialize to Lebesgue spaces L_p , $p \in (1, \infty)$, the *p*-Kazhdan constant can be estimated in terms of the Kazhdan constant in Hilbert space, via the explicit Mazur map.

FACT 6.8 (Follows from [BFGM07], [Maz29]). Let Γ be a discrete group equipped with finite generating set S, and suppose that Γ has the Kazhdan property (T_{L_2}) . Then for p > 2, $\kappa_{L_p}(\Gamma, S) \ge \frac{1}{p2^{p/2}}\kappa_{L_2}(\Gamma, S)$; for 1 , $<math>\kappa_{L_p}(\Gamma, S) \ge 2^{-\frac{p+2}{p}}\kappa_{L_2}^{2/p}(\Gamma, S)$.

Proof. Let $\rho: \Gamma \to O(L_p)$ be a Γ -representation in L_p , and take any unit vector f in the complement $\mathfrak{X}'(\rho)$. Let $M_{p,q}: L_p \to L_q$ be the Mazur map

$$M_{p,q}(f) = \operatorname{sign}(f)|f|^{p/q}.$$

By [BFGM07, Lemma 4.2], the conjugation $U \mapsto M_{p,2} \circ U \circ M_{2,p}$ sends $O(L_p)$ to $O(L_2)$. Define $\pi : \Gamma \to O(L_2)$ by $\pi(g) = M_{p,2} \circ \varrho(g) \circ M_{2,p}$. By definition of the Mazur map, we have $\|M_{p,2}(f)\|_2^2 = \|f\|_p^p = 1$.

Consider first the case p > 2. From [Maz29],

(29)
$$||a|^{\frac{2}{p}} \operatorname{sign}(a) - |b|^{\frac{2}{p}} \operatorname{sign}(b)| \le 2|a-b|^{\frac{2}{p}},$$

and so we have $\|M_{2,p}(u) - M_{2,p}(v)\|_p^p \leq 2^p \|u - v\|_2^2$ for $u, v \in L_2$. Note that $M_{p,2}$ maps $L_p^{\varrho(\Gamma)}$ onto $L_2^{\pi(\Gamma)}$, and therefore for any unit vector $f \in \mathfrak{X}'(\varrho)$, we

have

$$\inf_{v \in L_2^{\pi(\Gamma)}} \|M_{p,2}(f) - v\|_2^2 \ge \frac{1}{2^p} \inf_{h \in L_p^{\varrho(\Gamma)}} \|f - h\|_p^p \ge \frac{1}{2^p};$$

that is, the projection of $M_{p,2}(f)$ to $(L_2^{\pi(\Gamma)})^{\perp}$ has L_2 -norm at least $2^{-p/2}$. By [Maz29],

(30)
$$\left| |a|^{p/2} \operatorname{sign}(a) - |b|^{p/2} \operatorname{sign}(b) \right| \leq \frac{p}{2} |a - b| \left(|a|^{\frac{p}{2} - 1} + |b|^{\frac{p}{2} - 1} \right),$$

and so we have (31)

$$\begin{aligned} \|\pi(s)u - u\|_{2}^{2} &= \int \left| |\varrho(s)f|^{p/2} \operatorname{sign}(\varrho(s)f) - |f|^{p/2} \operatorname{sign}(f) \right|^{2} dm \\ &\leq \int \left(\frac{p}{2}\right)^{2} |\varrho(s)f - f|^{2} \left(|\varrho(s)f|^{\frac{p}{2}-1} + |f|^{\frac{p}{2}-1} \right)^{2} dm \\ &\leq \left(\frac{p}{2}\right)^{2} \left(\int |\varrho(s)f - f|^{p} dm \right)^{2/p} \left(\int \left(|\varrho(s)f|^{\frac{p}{2}-1} + |f|^{\frac{p}{2}-1} \right)^{\frac{2p}{p-2}} dm \right)^{\frac{p-2}{p}} \end{aligned}$$

The last step uses Hölder inequality. By triangle inequality in $L_{\frac{p}{p-2}}$ and $\varrho(s) \in O(L_p)$,

$$\left(\int \left(|\varrho(s)f|^{\frac{p}{2}-1} + |f|^{\frac{p}{2}-1}\right)^{\frac{2p}{p-2}} dm\right)^{\frac{p-2}{p}} \le 4 \, \|f\|_p^{p-2} = 4.$$

Let u' be the projection of u to $L'_2(\pi)$. Then

$$\max_{s \in S} \|\pi(s)u - u\|_2 = \max_{s \in S} \|\pi(s)u' - u'\|_2 \ge \kappa_{L_2}(\Gamma, S) \|u'\|_2 \ge 2^{-p/2} \kappa_{L_2}(\Gamma, S).$$

Combining with (31), we have

$$1 \le \frac{2^p}{\kappa_{L_2}(\Gamma, S)^2} \max_{s \in S} \|\pi(s)u - u\|_2^2 \le \left(\frac{p2^{p/2}}{\kappa_{L_2}(\Gamma, S)}\right)^2 \max_{s \in S} \|\varrho(s)f - f\|_p^2.$$

We conclude that $\kappa_{L_p}(\Gamma, S) \geq \frac{1}{p2^{p/2}}\kappa_{L_2}(\Gamma, S)$. In the case 1 , rewrite (29) as

$$|a-b|^p \le 2^p \left| |a|^{p/2} \operatorname{sign}(a) - |b|^{p/2} \operatorname{sign}(b) \right|^2$$

We deduce that the projection of $M_{p,2}(f)$ to $(L_2^{\pi(\Gamma)})^{\perp}$ has L_2 -norm at least $2^{-p/2}$. Applying (29), we get

$$\begin{aligned} \|\pi(s)u - u\|_2^2 &= \int \left| |\varrho(s)f|^{p/2} \operatorname{sign}(\varrho(s)f) - |f|^{p/2} \operatorname{sign}(f) \right|^2 dm \\ &\leq \int 4|\varrho(s)f - f|^p dm = 4 \left\| \varrho(s)f - f \right\|_p^p. \end{aligned}$$

It follows that $\kappa_{L_p}(\Gamma, S) \geq 2^{-\frac{p+2}{p}} \kappa_{L_2}^{2/p}(\Gamma, S).$

62

6.2.3. Basic test functions and 1-cocycles on Δ . The discussion in this subsection is valid for any choice of $\{\Gamma_s\}$ that satisfies Assumption 2.1. The 1-cocycle constructed using the basic test functions will be useful in later sections as well.

First recall some basic test functions on Δ constructed in Section 4.1.2. They capture the feature that in each copy of Γ_s in Δ_s , the generators $a_i(s)$ and $b_i(s)$ are kept distance k_s apart. In the group Δ , for $r \geq 2$, define the subset U_r as

$$U_r = \{ Z \in \Delta : \operatorname{Range}(Z) \subseteq [-r, r] \}.$$

Recall that $\operatorname{Range}(Z)$ is defined in Section 2.2.2; it is the minimal interval of \mathbb{Z} visited by the cursor of a path representing Z. Take a function supported on the subset U_r ,

(32)
$$\varphi_r\left((f_s), z\right) = \max\left\{0, 1 - \frac{|z|}{r}\right\} \mathbf{1}_{U_r}\left((f_s), z\right).$$

Let \mathfrak{q} be the switch-or-walk measure $\mathfrak{q} = \frac{1}{2}(\mu + \nu)$ on Δ , where ν is the uniform measure on $\{\alpha_i, \beta_j : 1 \leq i \leq |A|, 1 \leq j \leq |B|\}$ and μ is the simple random walk measure on the base \mathbb{Z} , $\mu(\tau^{\pm 1}) = \frac{1}{2}$. We have seen in the proof of Proposition 4.4 that for p = 2,

$$\frac{\mathcal{E}_{\Delta,\mathfrak{q}}\left(\varphi_{r}\right)}{\|\varphi_{r}\|_{2}^{2}} \sim \frac{3}{2r^{2}}.$$

Define φ_1 to be the indicator function of the identity e_{Δ} ,

$$\varphi_1 = \mathbf{1}_{e_\Delta}.$$

Motivated by Tessera's embedding in [Tes11, §3], given a non-decreasing function $\gamma : \mathbb{Z}_{\geq 0} \to \mathbb{R}_+$ with $\gamma(0) = 1$ and

(33)
$$C(\gamma) = \sum_{t=1}^{\infty} \left(\frac{1}{\gamma(t)}\right)^2 < \infty,$$

we define a 1-cocycle $b_{\gamma} : \Delta \to \bigoplus_{j=0}^{\infty} \ell^2(\Delta) \subset L_2$,

(34)
$$b_{\gamma}(Z) = \bigoplus_{j=0}^{\infty} \left(\frac{1}{\gamma(j)} \frac{\varphi_{2^j} - \tau_Z \varphi_{2^j}}{\mathcal{E}_{\Delta,\mathfrak{q}} (\varphi_{2^j})^{\frac{1}{2}}} \right),$$

where τ_g denote right translation of functions, $\tau_g \varphi(h) = \varphi(hg^{-1})$.

LEMMA 6.9. The 1-cocycle $b_{\gamma} : \Delta \to L_2$ defined in (34) is $\sqrt{2C(\gamma)}$ -Lipschitz. Suppose in addition that there exists $m_0 \ge 2$ such that $k_{s+1} \ge m_0 k_s$, $l_{s+1} \ge m_0 l_s$ for any $s \ge 1$. Then there is a constant $C(m_0) > 0$ depending only on m_0 such that

$$\rho_{b_{\gamma}}(x) \ge C(m_0) \frac{x/l_s}{\gamma(\log_2(x/l_s))} \quad \text{if } x \in [k_s l_s, k_{s+1} l_s),$$

$$\rho_{b_{\gamma}}(x) \ge C(m_0) \frac{k_{s+1}}{\gamma(\log_2 k_{s+1})} \quad \text{if } x \in [k_{s+1} l_s, k_{s+1} l_{s+1}).$$

Proof. Clearly $||b_{\gamma}(Z)|| = 1$ when Z is a generator in $A \cup B$ and $||b_{\gamma}(Z)|| = \sqrt{2C(\gamma)}$ when $Z = \tau$ is the generator of the cyclic base \mathbb{Z} .

The 1-cocycle b_{γ} captures the size of $\operatorname{Range}(Z)$. Denote $Z = ((f_s), i)$ and observe that if $\operatorname{Range}(Z) > 2^{j+2}$, then

$$(\operatorname{supp}\varphi_{2^j}) \cap (\operatorname{supp}\tau_Z\varphi_{2^j}) = \emptyset.$$

Indeed, either there exist $\iota \notin [-2^{j+1}, 2^{j+1}]$ and $s \leq s_0(Z)$ with $f_s(\iota) \neq e_{\Delta_s}$ and then this also holds for all elements of $\operatorname{supp} \tau_Z \varphi_{2^j}$ and none of $\operatorname{supp} \varphi_{2^j}$, or $|i| > 2^{j+1}$ and then the projections on the base of the two supports are disjoint. Therefore

$$\frac{\|\varphi_{2^j} - \tau_Z \varphi_{2^j}\|_2^2}{\mathcal{E}_{\Delta,\mathfrak{q}}\left(\varphi_{2^j}\right)} = \frac{2 \|\varphi_{2^j}\|_2^2}{\mathcal{E}_{\Delta,\mathfrak{q}}\left(\varphi_{2^j}\right)} \sim \frac{4 \cdot 2^{2j}}{3}.$$

By construction of b_{γ} , this implies that if $\operatorname{Range}(Z) > 2^{j+2}$, then

$$\left\|b_{\gamma}(Z)\right\|_{2} \ge \frac{2^{j}}{\sqrt{3}\gamma(j)}.$$

By Definition 2.8, for Z with Range $(Z) = r \in [k_s, k_{s+1})$, we have $s_0(Z) \leq s$. Denote $Z = ((f_s), i)$. Then by Lemma 2.13,

$$|(f_s, i)|_{\Delta_s} = |\pi_s(Z)|_{\Delta_s} \le 18(r+1)l_s$$

because at most $2r/k_s + 1$ intervals contribute to the essential contribution. By Proposition 2.14, the word distance of Z to e_{Δ} is bounded:

$$|Z|_{\Delta} \le 500 \cdot 18(r+1) \left(l_0 + \dots + l_s \right) \le \frac{9000(r+1)l_s}{1 - 1/m_0}.$$

It follows that

$$\rho_{b_{\gamma}}\left(\frac{9000(r+1)l_s}{1-1/m_0}\right) \ge \frac{r}{8\gamma(\log_2 r)}.$$

To write it into the first inequality stated, note that since b_{γ} is equivariant, $\rho_{b_{\gamma}}$ is subadditive.

The second bound follows from the first bound evaluated at $x = k_{s+1}l_s$ and the monotonicity of the compression function $\rho_{b_{\gamma}}$.

Remark 6.10. The function $t/\gamma \circ \log(t)$ with γ satisfying (33) does not satisfy any a priori majoration by a sublinear function. More precisely, [Tes11, Prop. 8] implies that for any increasing sublinear function $h : \mathbb{R}_+ \to \mathbb{R}_+$, there exist a non-decreasing function γ satisfying (33), a constant c > 0 and an increasing subsequence of integers n_i , such that

$$\frac{k_{n_i}}{\gamma \circ \log_2\left(k_{n_i}\right)} \ge ch\left(k_{n_i}\right) \text{ for all } i.$$

It follows that the upper bound in Proposition 6.3 cannot be improved.

6.2.4. Possible compression gaps of embeddings into L_p . The 1-cocycle b_{γ} defined in (34) is almost optimal in the sense that the lower bounds of Lemma 6.9 match with the upper bounds of Proposition 6.3 up to the factor sequence $1/\gamma \circ \log_2(k_s)$. The following result is an analogue of [ADS09, Th. 5.5(II)] in our setting.

THEOREM 6.11. There exist absolute constants $\delta > 0, C > 0$ such that the following holds. Let $\rho(x)$ be any non-decreasing function such that $\frac{x}{\rho(x)}$ is non-decreasing. Then there exists a finitely generated group Δ such that $\left(\frac{1}{C\varepsilon}\frac{\rho}{\log(1+\rho)^{(1+\varepsilon)/2}}, Cp2^{p/2}\rho\right)$ is an equivariant L_p -compression gap of Δ for any p > 1.

Furthermore, if $\lim_{x\to\infty} \rho(x) = \infty$, then the group Δ constructed is elementary amenable, and $\left(\frac{1}{C\varepsilon} \frac{\rho}{\log(1+\rho)^{(1+\varepsilon)/2}}, C\rho\right)$ is an equivariant L_1 -compression gap of Δ .

Proof. We write $\rho(x) = \frac{x}{f(x)}$ with f(x) between 1 and x. The sets $K = \mathbb{Z}_+ \cup \{\infty\}$ and $L = \{\operatorname{diam} \Gamma_m, m \ge 1\} \cup \{\infty\}$ of diameters of Lafforgue expanders from Example 2.3 satisfy the assumptions of Proposition B.2. So we can find sequences $(k_s), (l_s)$ taking values in K and L such that the function defined by $\tilde{f}(x) = l_s$ on $[k_s l_s, k_{s+1} l_s]$ and $\tilde{f}(x) = \frac{x}{k_{s+1}}$ on $[k_{s+1} l_s, k_{s+1} l_{s+1}]$ satisfies $\tilde{f}(x) \simeq_{m_0 C_1^5} f(x)$. Then $\tilde{\rho}(x) = \frac{x}{\tilde{f}(x)} \simeq_{m_0 C_1^5} \rho(x)$. Let Δ be the diagonal product associated to these sequences.

By Proposition 6.3 and Lemma 6.6, any 1-Lipschitz equivariant embedding $\Psi : \Delta \to L_p$ satisfies, for all x,

$$\rho_{\Psi}(x) \le C\tilde{\rho}(2x) \le 2m_0 C_1^5 C\rho(x),$$

where by Fact 6.8,

$$C \le \max\{C(\delta, p), \frac{2}{\kappa_{L_p}(\Gamma)})\} \le \max\{C'p, C''p2^{p/2}\}.$$

This gives the upper bound of the compression gap. For p = 1, Lemma 6.6 does not hold, so the upper bound is valid only on the condition that all the diameters l_s are finite, which is satisfied when ρ is unbounded, or equivalently, f is not asymptotically linear.

The lower bound is given by the 1-cocyle $b_{\gamma} : \Delta \to \ell^2$ of Lemma 6.9 with $\gamma(x) = C_{\epsilon} x^{\frac{1+\epsilon}{2}}$, where $C_{\epsilon} \sim \frac{1}{\epsilon}$ is such that $\sqrt{2C(\gamma)} = 1$. For all x,

$$\rho_{b_{\gamma}}(x) \ge C_{\epsilon}C(m_0)\frac{\tilde{\rho}(x)}{\log_2(\tilde{\rho}(x))^{\frac{1+\epsilon}{2}}} \ge \frac{1}{C\epsilon}\frac{\rho(x)}{\log(1+\rho(x))^{\frac{1+\epsilon}{2}}}$$

Since ℓ^2 embeds isometrically in L_p for all $p \ge 1$ (see Lemma 2.3 in [NP08]), it is also an L_p -compression lower bound.

7. L_p -compression of the wreath product $H \wr \mathbb{Z}$

In general, for the diagonal products constructed with $\{\Gamma_s\}$ chosen to be finite groups other than expanders, the analysis of L_p -compression is more involved. Since our main object Δ is a diagonal product of a sequence of wreath products, in this section we first understand compression of uniform embedding of a single copy of the wreath product $H \wr \mathbb{Z}$.

In [NP08], Naor and Peres prove that if $\alpha_2^{\#}(H) = \frac{1}{2\beta^*(H)}$, where $\beta^*(H)$ is the supremum of upper speed exponent of symmetric random walk of bounded support on H, then ([NP08, Cor. 1.3])

$$\alpha_2^{\#}(H \wr \mathbb{Z}) = \frac{2\alpha_2^{\#}(H)}{2\alpha_2^{\#}(H) + 1}.$$

In their subsequent work [NP11], which significantly extends the method in [NP08], the L_p -compression exponent of $\mathbb{Z} \wr \mathbb{Z}$ is determined ([NP11, Th. 1.2]): for every $p \in [1, \infty)$,

$$\alpha_p^{\#}\left(\mathbb{Z}\wr\mathbb{Z}\right) = \max\left\{\frac{p}{2p-1}, \frac{2}{3}\right\}.$$

In [NP11] Naor and Peres also prove the following result when the base group is of polynomial volume growth at least quadratic. Let H be a non-trivial finitely generated amenable group and Γ a group of polynomial volume growth. Suppose the volume growth rate of Γ is at least quadratic. Then [NP11, Th. 3.1] states that for every $p \in [1, 2]$,

$$\alpha_p^{\#}(H \wr \Gamma) = \min\left\{\frac{1}{p}, \alpha_p^{\#}(H)\right\}.$$

One central idea in these works is the Markov type method that connects compression exponents of G to the speed exponent of certain random walks on G. The aim of this section is to obtain the following generalization of the aforementioned results of Naor and Peres.

66

THEOREM 7.1. Let $p \in [1,2]$, and let H be a finitely generated infinite group. Then the equivariant L_p -compression exponent of $H \wr \mathbb{Z}$ is

$$\alpha_p^{\#}(H \wr \mathbb{Z}) = \min\left\{\frac{\alpha_p^{\#}(H)}{\alpha_p^{\#}(H) + \left(1 - \frac{1}{p}\right)}, \alpha_p^{\#}(H)\right\}.$$

Remark 7.2. It will be transparent in the proof of Theorem 7.1 that there is a dichotomy of the kind of obstruction that $H \wr \mathbb{Z}$ observes, depending on $\alpha_p^{\#}(H)$ as indicated in the formula. Namely, when $\alpha_p^{\#}(H) > \frac{1}{p}$, then the sequence of subsets X_n as defined in the proof of Proposition 7.3 captures distorted elements under the embedding; when $\alpha_p^{\#}(H) \leq \frac{1}{p}$, then one single copy of H already provides sufficient distortion, and in the latter case we have $\alpha_p^{\#}(H \wr \mathbb{Z}) = \alpha_p^{\#}(H)$.

7.1. Upper bound of compression function of $H \wr \mathbb{Z}$. Let $\Psi : H \wr \mathbb{Z} \to \mathfrak{X}$ be an equivariant embedding of $G = H \wr \mathbb{Z}$ into metric space $(\mathfrak{X}, d_{\mathfrak{X}})$. Recall that the group H is naturally identified with the lamp group over site 0,

$$i: H \hookrightarrow G,$$

 $h \to (h\delta_0, 0).$

Then the embedding Ψ induces an equivariant embedding of H into \mathfrak{X} . We denote it by ψ_H

(35)
$$\psi_H(h) = \Psi \circ i(h) = \Psi ((h\delta_0, 0)).$$

Denote the compression function of $\psi_H : H \to \mathfrak{X}$ by ρ_{ψ_H} . Since distortion of the inclusion map *i* is 1, $|h|_H = |i(h)|_G$, it follows that

(36)
$$\rho_{\Psi}(t) \le \rho_{\psi_H}(t) \text{ for all } t \ge 1.$$

We now explain how to apply the spectral method to derive a second upper bound on ρ_{Ψ} when Ψ is a uniform embedding of G into L_p , $p \in (1,2]$. The novelty here is in the choice of Markov kernels on lamplighter graphs.

PROPOSITION 7.3. There exists a constant C > 0 such that for any $p \in (1,2]$ and any 1-Lipschitz equivariant embedding $\Psi : G \to L_p$ of $G = H \wr \mathbb{Z}$ into L_p , the compression function ρ_{Ψ} of Ψ satisfies

(37)
$$\rho_{\Psi}(t) \le C \left(\frac{p}{p-1}\right)^{\frac{1}{p}} \left(2\rho_{\psi_H} \circ \tau^{-1}(t)\right)^{\frac{p}{p-1}} \log^{\frac{1}{p}} \left(2\rho_{\psi_H} \circ \tau^{-1}(t)\right),$$

where ψ_H is the induced embedding of the subgroup H into L_p as in (35) and the function $\tau : \mathbb{R}_+ \to \mathbb{R}_+$ is defined as

$$\tau(x) = 2^{\frac{1}{p-1}} x \left(\rho_{\psi_H}(x) \right)^{\frac{p}{p-1}}.$$

Proof. By definition of the compression function ρ_{ψ_H} , for every $n \ge 1$, there exists an element $h_n \in H$ such that $|h_n|_H \ge n$ and

$$\left\|\psi_{H}\left(h_{n}\right)-\psi_{H}\left(e_{H}\right)\right\|_{p}\leq2\rho_{\psi_{H}}(n).$$

Take X_n to be a finite subset in G defined by

$$X_n = \left\{ (f, z) : z \in [0, m_n - 1], \begin{array}{c} f(x) \in \{e_H, h_n\} & \text{for } x \in [0, m_n - 1] \\ f(x) = e_H & \text{for } x \notin [0, m_n - 1] \end{array} \right\}.$$

The length m_n will be determined later. Note that X_n has the structure of a lamplighter graph; namely, let \mathcal{L}_{m_n} be lamplighter graph over segment $[0, m_n - 1]$ defined in Appendix C. Then there is a bijection

$$\sigma_n : \mathcal{L}_{m_n} \to X_n,$$

$$\sigma_n (f, x) = \left(\tilde{f}, x\right), \text{ where } \tilde{f}(z) = h_n^{f(z)}$$

In Appendix C we define a Markov transition kernel \mathfrak{p}_{m_n} on \mathcal{L}_{m_n} that moves on the base segment with a Cauchy-like step distribution ζ_{m_n} . On X_n , take the Markov kernel K_n to be $\mathfrak{p}_{m_n} \circ \sigma_n^{-1}$; that is,

$$K_n(u,v) = \mathfrak{p}_{m_n} \left(\sigma_n^{-1}(u), \sigma_n^{-1}(v) \right).$$

Denote by π_n the stationary distribution of K_n on X_n , $\pi_n = U_{m_n} \circ \sigma_n^{-1}$. Under the bijection σ_n , the Poincaré inequality that $\Psi \circ \sigma_n : \mathcal{L}_{m_n} \to L_p$ satisfies as in Lemma C.1 implies

(38)
$$\sum_{u,v\in X_n} \|\Psi(u) - \Psi(v)\|_p^p \pi_n(u)\pi_n(v) \\ \leq Cm_n \log m_n \sum_{u,v\in X_n} \|\Psi(u) - \Psi(v)\|_p^p K_n(u,v)\pi_n(u).$$

Now we deduce an upper bound on ρ_{Ψ} from Poincaré inequalities (38) by applying Lemma 5.2. Because of equivariance of Ψ , for any $u \in G$,

$$\|\Psi(u \cdot (h\delta_0, 0)) - \Psi(u)\|_p = \|\Psi((h\delta_0, 0)) - \Psi(e_G)\|_p = \|\psi_H(h) - \psi_H(e_H)\|_p.$$

Recall that K_n moves as a "switch-walk-switch" transition kernel. By the Hölder inequality,

$$\begin{split} &\sum_{u,v \in X_n} \|\Psi(u) - \Psi(v))\|_p^p K_n(u,v) \pi_n(u) \\ &\leq 2 \cdot 3^{p-1} \|\psi_H(e_H) - \psi_H(h_n)\|_p^p \\ &+ 3^{p-1} \sum_{(f,z) \in X_n} \sum_{y \in [0,m_n-1]} \|\Psi((f,z)) - \Psi((f,z) \cdot (\mathbf{e}_H,y))\|_p^p \zeta_{m_n}(z,y) \pi_n((f,z)) \\ &\leq 3^p \left[(2\rho_{\psi_H}(n))^p + \sum_{z,y \in [0,m_n-1]} |z-y|^p \zeta_{m_n}(z,y) \mathcal{C}_{m_n}(z) \right], \end{split}$$

where $C_{m_n}(z)$ denotes the stationary distribution of ζ_{m_n} . The last step used the choice that $\|\psi_H(h_n) - \psi_H(e_H)\|_p \leq 2\rho_{\psi_H}(n)$ and the assumption that Ψ is 1-Lipschitz. From the explicit formula that defines ζ_{m_n} , for p > 1, we have

$$\sum_{y \in [0, m_n - 1]} |z - y|^p \zeta_{m_n}(z, y) \mathcal{C}_{m_n}(z) \le C m_n^{p-1}.$$

Set

z

$$m_n = \left\lceil (2\rho_{\psi_H}(n))^{\frac{p}{p-1}} \right\rceil,$$

so that the two terms in the L_p -energy upper bound are comparable. Then

(39)
$$\sum_{u,v\in X_n} \|\Psi(u) - \Psi(v)\|_p^p K_n(u,v)\pi_n(u) \le 3^p (1+C)m_n^{p-1}$$

From the explicit lamplighter structure, one checks that the set X_n satisfies $\operatorname{diam}_G(X_n) = (2 + |h_n|_H) m_n$, and that for any $u \in X_n$,

(40)
$$\sum_{v \in X_n} \mathbf{1}_{\left\{ d_G(u,v) \ge \frac{1}{2} \left(|h_n|_H + 2 \right) m_n \right\}} \pi_n(v) \ge \frac{1}{5}.$$

Applying (38), (39), and (40) in Lemma 5.2, we have

$$\rho_{\Psi}\left(\frac{(2+|h_n|_H)m_n}{2}\right) \le \left(C'm_n^{p-1} \cdot m_n\log m_n\right)^{\frac{1}{p}}.$$

Since $|h_n|_H \ge n$, it follows that there exist $u, v \in X_n$, $d_G(u, v) \ge \frac{1}{2}(n+1)m_n$, and

$$\|\Psi(u) - \Psi(v)\|_{p} \le C' m_{n} \log^{\frac{1}{p}} m_{n}$$

Note that by definition of τ , $\tau(n) = \frac{1}{2}n (2\rho_{\psi_H}(n))^{\frac{p}{p-1}}$, with the choice of $m_n = \left\lceil (2\rho_{\psi_H}(n))^{\frac{p}{p-1}} \right\rceil$, the statement follows from rewriting the inequality

$$\rho_{\Psi}\left(\frac{1}{2}nm_n\right) \le C'm_n \log^{\frac{1}{p}} m_n.$$

The Markov type method can also be applied to this situation; it actually yields more general results. We presented the proof for L_p -compression of Gwith $p \in (1, 2]$ using the Poincaré inequalities because spectral gap considerations motivate the choice of the α -stable walk on the base with $\alpha = 1$. Now we explain how to apply the Markov type method. Let $(\mathfrak{X}, d_{\mathfrak{X}})$ be a metric space of Markov type p, p > 1, and let $\Psi : H \wr \mathbb{Z} \to \mathfrak{X}$ be a 1-Lipschitz equivariant embedding. Make the same choice of a sequence of finite subsets X_n and K_n as in the proof of Proposition 7.3 with

$$m_n = \left\lceil \left(2\rho_{\psi_H}(n)\right)^{\frac{p}{p-1}} \right\rceil.$$

Let Z_t denote a Markov chain on \mathcal{L}_{m_n} with transition kernel \mathfrak{p}_{m_n} . Then $\tilde{Z}_t = \sigma_n(Z_t)$ is a Markov chain on X_n with transition kernel K_n . Run the Markov chain \tilde{Z}_t up to time $t_n = m_n \log m_n$. To apply Lemma 5.5, we need a lower

bound for $\mathbf{E}_{\pi_n} \left[d_G \left(\widetilde{Z}_{t_n}, \widetilde{Z}_0 \right)^p \right]$. The bijection $\sigma_n : \mathcal{L}_{m_n} \to X_n$ induces a metric on \mathcal{L}_{m_n} by

$$d_{\sigma_n}(u,v) = d_G(\sigma_n(u),\sigma_n(v)).$$

Direct inspection shows that this metric d_{σ_n} coincides with the metric $d_{\mathbf{w}_n}$ with $\mathbf{w}_n = (1, |h_n|_H)$ introduced in Appendix C. By Lemma C.2, we have that for $t_n = m_n \log m_n$,

$$\mathbf{E}_{\pi_n} \left[d_G \left(\widetilde{Z}_{t_n}, \widetilde{Z}_0 \right)^p \right] = \mathbf{E}_{U_{\alpha, m_n}} \left[d_{\mathbf{w}_n} \left(Z_{t_n}, Z_0 \right)^p \right] \\ \geq \mathbf{E}_{U_{\alpha, m_n}} \left[d_{\mathbf{w}_n} \left(Z_{t_n}, Z_0 \right) \right]^p \\ \geq \left(c \left(1 + \left| h_n \right|_H \right) m_n \right)^p.$$

In the proof of Proposition 7.3 we checked that

$$\mathbf{E}_{\pi_n} \left[d_{\mathfrak{X}} \left(\Psi \left(\widetilde{X}_1 \right), \Psi \left(\widetilde{X}_0 \right) \right)^p \right] = \sum_{u, v \in X_n} d_{\mathfrak{X}} \left(\Psi(u), \Psi(v) \right)^p K_n(u, v) \pi_n(v) \\ \leq 3^p \left[(2\rho_{\psi_H}(n))^p + Cm_n^{p-1} \right].$$

Choose $m_n = \left\lceil (2\rho_{\psi_H}(n))^{\frac{p}{p-1}} \right\rceil$, and plug these estimates into Lemma 5.5. Then we have

$$\rho_{\Psi}\left(\frac{c}{2}\left(1+|h_{n}|_{H}\right)m_{n}\right)\leq C'M_{p}\left(\mathfrak{X}\right)m_{n}\log^{\frac{1}{p}}m_{n}.$$

Recall that $|h_n|_H \ge n$, and therefore

$$\rho_{\Psi}\left(\frac{c}{2}nm_{n}\right) \leq C'M_{p}\left(\mathfrak{X}\right)m_{n}\log^{\frac{1}{p}}m_{n}.$$

The result given by the Markov type method is recorded in the following proposition. By [NPSS06], L_p space with p > 2 has Markov type 2. Proposition 7.4 applies to L_p with p > 2 as well and therefore is more general than Proposition 7.3.

PROPOSITION 7.4. There exists a constant C > 0 such that the following holds. Let $(\mathfrak{X}, d_{\mathfrak{X}})$ be a metric space of Markov type p, p > 1, and let Ψ : $H \wr \mathbb{Z} \to \mathfrak{X}$ be a 1-Lipschitz equivariant embedding. The compression function ρ_{Ψ} satisfies

(41)
$$\rho_{\Psi}(t) \leq C\left(\frac{p}{p-1}\right)^{\frac{1}{p}} M_p(\mathfrak{X}) \left(2\rho_{\psi_H} \circ \tau^{-1}(t)\right)^{\frac{p}{p-1}} \log^{\frac{1}{p}} \left(2\rho_{\psi_H} \circ \tau^{-1}(t)\right),$$

where ψ_H is the induced embedding of the subgroup H into \mathfrak{X} as in (35), $M_p(\mathfrak{X})$ is the Markov-type p constant of \mathfrak{X} , and the function $\tau : \mathbb{R}_+ \to \mathbb{R}_+$ is defined as

$$\tau(x) = \frac{1}{C} x \left(2\rho_{\psi_H}(x) \right)^{\frac{p}{p-1}}.$$

7.2. L_p -compression exponent of $H \wr \mathbb{Z}$. For the lower bound in the L_p compression gap of $H \wr \mathbb{Z}$, $p \in [1, 2]$, we use the embedding constructed in [NP08, Th. 3.3]. An explicit description of an equivariant embedding is included here
as a warm-up for Section 8. Given a good equivariant embedding φ of the
group H into L_p ,

$$\|\varphi(h_1) - \varphi(h_2)\|_p \ge \rho_- (d_H(h_1, h_2)),$$

we exhibit an embedding of $G = H \wr \mathbb{Z}$ into L_p with φ as building blocks. Recall that for an element $(f, z) \in G$, the word distance is given by

$$|(f,z)|_G = |\omega| + \sum_{x \in \mathbb{Z}} |f(x)|_H,$$

where ω is a path of shortest length that starts at 0, visits every point x in the support of f, and ends at z. We refer to such a path as a traveling salesman path for (f, z). The embedding of G into L_p consists of two parts: Part I captures the length of ω , and Part II embeds the lamp configurations $\{f(x)\}_{x\in\mathbb{Z}}$ using the embedding of H into L_p .

Part I. This part is essentially the same as an embedding of $\mathbb{Z}_2 \wr \mathbb{Z}$ into L_p . Consider the following sequence of functions on $G = H \wr \mathbb{Z}$:

(42)
$$\phi_n\left((f,z)\right) = \mathbf{1}_{\{\text{supp} f \subseteq [-2^n, 2^n]\}} \max\left\{1 - \frac{|z|}{2^n}, 0\right\}.$$

Note that if $g \in G$ is in the support of ϕ_n , then

$$|\omega_g| \le 4 \cdot 2^n$$

where ω_q denotes a traveling salesman path for g. Also,

 $(\operatorname{supp}\phi_n)(\operatorname{supp}\phi_n)^{-1} \subseteq \operatorname{supp}\phi_{n+1}.$

As in [Tes11], given a non-decreasing function $\gamma : \mathbb{N} \to \mathbb{R}_+$ with $\gamma(1) = 1$ and $\sum_{t=0}^{\infty} \left(\frac{1}{\gamma(t)}\right)^p = C_p(\gamma) < \infty$, take a cocycle $b_\gamma : G \to L_p$ defined as

(43)
$$b_{\gamma}(g) = \bigoplus_{n=1}^{\infty} \frac{1}{\gamma(n)} \left(\frac{\tau_g \phi_n - \phi_n}{\mathcal{E}_p(\phi_n)^{1/p}} \right),$$

where τ_g denotes the right translation of a function $\phi: G \to \mathbb{R}$,

$$\tau_g \phi(u) = \phi \left(ug^{-1} \right).$$

Because of the normalization in the definition of b_{γ} , one readily checks that b_{γ} is $(2C_p(\gamma))^{\frac{1}{p}}$ -Lipschitz.

Given $g = (f, x) \in G$, we say a path ω on \mathbb{Z} is a traveling salesman path for g if it starts at 0, visits every $z \in \mathbb{Z}$ where $f(z) \neq e_H$, and ends at x. Let ω_g be a shortest traveling salesman path for g. Suppose $|\omega_g| > 2^{n+3}$. Then $g \notin (\operatorname{supp}\phi_n) (\operatorname{supp}\phi_n)^{-1}$, and it follows that in this case,

$$\frac{\left\|\tau_{g}\phi_{n}-\phi_{n}\right\|_{p}^{p}}{\mathcal{E}_{p}\left(\phi_{n}\right)}=\frac{2\left\|\phi_{n}\right\|_{p}^{p}}{\mathcal{E}_{p}\left(\phi_{n}\right)}\geq\left(2^{n}\right)^{p}.$$

Putting the components together, we have that

$$\left\|b_{\gamma}(g)\right\|_{p} \geq \frac{\left|\omega_{g}\right|}{8\gamma\left(\log_{2}\left|\omega_{g}\right|\right)}.$$

Part II. Let $\varphi : H \to L_p$ be a 1-Lipschitz equivariant embedding of H into L_p . Define a map $\Xi_{\varphi} : G \to L_p$ as

$$\Xi_{\varphi}((f,z)) = \bigoplus_{x \in \mathbb{Z}} \varphi(f(x)).$$

Since the map Ξ_{φ} factors through the projection $G \to \bigoplus_{x \in \mathbb{Z}} H$ and φ is equivariant, it follows that Ξ_{φ} is a cocycle. By construction, Ξ_{φ} is 1-Lipschitz and

$$\|\Xi_{\varphi}((f,z))\|_p^p = \sum_{x \in \mathbb{Z}} \|\varphi(f(x))\|_p^p.$$

Combining the two parts, we define

(44)
$$\Phi: G \to L_p,$$
$$\Phi(g) = b_{\gamma}(g) \bigoplus \Xi_{\varphi}(g).$$

Then we have that Ψ is $(2C_p(\gamma))^{1/p}$ -Lipschitz and

$$\begin{split} |\Phi(g)||_p^p &= \|b_{\gamma}(g)\|_p^p + \sum_{x \in \mathbb{Z}} \|\varphi(f(x))\|_p^p \\ &\geq \frac{|\omega_g|^p}{8\gamma \left(\log_2 |\omega_g|\right)} + c \sum_{x \in \mathbb{Z}} \rho_- \left(|f(x)|_H\right)^p \end{split}$$

Let $p \in (1, 2]$. The bounds (35), (36) and the embedding constructed above provide rather detailed information about the L_p -compression gap of $G = H \wr \mathbb{Z}$ in terms of L_p -compression gap of H. We now derive the formula relating the L_p -compression exponents of G and H stated in Theorem 7.1.

Proof of Theorem 7.1. We first treat the case $p \in (1, 2]$. Let $\Psi : G \to L_p$ be a 1-Lipschitz equivariant embedding of $G = H \wr \mathbb{Z}$ into L_p and ψ_H its induced embedding $H \to L_p$. From (36), it is always true that $\alpha_p^{\#}(H \wr \mathbb{Z}) \leq \alpha_p^{\#}(H)$. Now we apply Proposition 7.3 to prove the upper bound. By definition of the compression exponent, there exists a constant $C = C(\psi_H) > 0$ and an increasing sequence $n_i \in \mathbb{N}$ with $n_i \to \infty$ such that

$$\rho_{\psi_H}(n_i) \le C n_i^{\alpha_p^{\#}(H)}.$$
Along the sequence $\{n_i\}$, by Proposition 7.3,

$$\rho_{\Psi}(n_i) \le C_1(C, p) n_i^{\frac{p\alpha}{p-1+\alpha p}} \log^{\frac{1}{p}} \left(n_i^{\frac{p\alpha}{p-1+\alpha p}} \right),$$

where $\alpha = \alpha_p^{\#}(H)$ and $C_1(C, p)$ is a constant depending on C and p. Therefore

$$\alpha_p^{\#}(H \wr \mathbb{Z}) \le \frac{\alpha_p^{\#}(H)}{\alpha_p^{\#}(H) + \left(1 - \frac{1}{p}\right)}.$$

We have proved the upper bound.

Note that if $\alpha_p^{\#}(H) = 0$, then $\alpha_p^{\#}(G) = 0$. Thus in the lower bound direction, we consider the case $\alpha_p^{\#}(H) > 0$. For any $0 < \varepsilon < \alpha_p^{\#}(H)$, let $\varphi: H \to L_p$ be a 1-Lipschitz equivariant embedding such that

$$\rho_{\varphi}(t) \ge (ct)^{\alpha_p^{\#}(H) - \varepsilon},$$

and set $\gamma(n) = \log^{\frac{1+\varepsilon}{p}}(1+n)$. Take the embedding $\Phi: G \to L_p$ as defined in (44). Then

$$\begin{split} \|\Phi(g)\|_p^p &= \|b_{\gamma}(g)\|_p^p + \sum_{x \in \mathbb{Z}} \|\varphi(f(x))\|_p^p \\ &\geq \frac{|\omega_g|^p}{8\log^{1+\varepsilon} (1+|\omega_g|)} + \sum_{x \in \mathbb{Z}} (c |f(x)|_H)^{\left(\alpha_p^{\#}(H) - \varepsilon\right)p} \end{split}$$

Such an embedding Φ is analyzed in [NP08]. As ε is arbitrarily small, the proof of [NP08, Th. 3.3] applies to show that if $\alpha_p^{\#}(H) \leq \frac{1}{p}$, then $\alpha_p(\Phi) \geq \alpha_p^{\#}(H)$, and if $\alpha_p^{\#}(H) > \frac{1}{p}$, then

$$\alpha_p^{\#}(\Phi) \ge \frac{\alpha_p^{\#}(H)}{\alpha_p^{\#}(H) + 1 - 1/p}$$

Finally when p=1, from $\rho_{\Psi}(t) \leq \rho_{\psi_H}(t)$ and the explicit embedding $\Phi: G \to L_1$ given a good embedding $\varphi: H \to L_1$, we deduce that

$$\alpha_1^{\#}(G) = \alpha_1^{\#}(H). \qquad \Box$$

8. Compression of Δ constructed with dihedral groups

Throughout this section Δ denotes a diagonal product with input { $\Gamma_s = D_{2l_s}$ }. The main result of this section is the following.

THEOREM 8.1. Let Δ be the diagonal product with $\{\Gamma_s = D_{2l_s}\}$ and parameters (k_s) , and set

$$\theta := \limsup_{s \to \infty} \frac{\log l_s}{\log k_s}.$$

Assume that (k_s) satisfies the growth assumption 2.11. Then

(i) for $p \in [1, 2]$,

$$\alpha_p^*\left(\Delta\right) = \max\left\{\frac{1}{1+\theta}, \frac{2}{3}\right\};$$

(ii) for $q \in (2, \infty)$,

$$\alpha_q^*(\Delta) = \frac{1}{1+\theta} \quad \text{if } 0 \le \theta \le \frac{1}{q},$$
$$\max\left\{\frac{\theta+1-\frac{2}{q}}{\left(2-\frac{1}{q}\right)\theta+1-\frac{2}{q}}, \frac{2}{3}\right\} \le \alpha_q^*(\Delta) \le \frac{2\theta+1-\frac{2}{q}}{3\theta+1-\frac{2}{q}} \quad \text{if } \theta > \frac{1}{q}.$$

When $p \in [1, 2]$, the upper bound on $\alpha_p^*(\Delta)$ is a consequence of the Mendel-Naor metric cotype inequality cited in Section 5.4; in the lower bound direction we construct an explicit embedding $\Delta \to \ell^2$. The case of $p \in (2, \infty)$ is more involved. The proof of Theorem 8.1 is completed in Section 8.4.

Since Δ is 3-step solvable, in particular it is amenable. By [NP11, Th. 1.6], we have

$$\alpha_p^*(\Delta) = \alpha_p^{\#}(\Delta) \text{ for all } p \in [1, \infty).$$

8.1. Upper bounds on compression functions. We first explain why it is necessary to examine the distortion in a block of side length k_s in Δ_s . For notational convenience, assume that (k_s) , (l_s) are multiples of 4. As in Section 2.3, consider the subset $\Pi_s^{k_s/2}$ of Δ defined as in (2). Note that $\Pi_s^{k_s/2}$ is isomorphic to the direct product of $k_s/2$ copies of $D'_{2l_s} \simeq \mathbb{Z}_{l_s/2}$. Denote by $\vartheta_s : \mathbb{Z}_{l_s/2}^{k_s/2} \to \Pi_s^{k_s/2}$ the isomorphism. Write elements of $\Pi_s^{k_s/2}$ as vectors $u = (u(0), \ldots, u(k_s - 1)), u(j) \in \mathbb{Z}_{l_s/2}$.

Now consider the induced metric on $\Pi_s^{k_s/2}$ of the word metric d_{Δ} on Δ . Then by Lemma 2.16, we have that for $u \in \Pi_s^{k_s/2}$,

$$|u|_{\Delta} \simeq_{72} k_s \max_{0 \le j \le k_s/2 - 1} |u(j)|_{\mathbb{Z}/l_s \mathbb{Z}}$$

Therefore the induced metric $|\cdot|_{\Delta}$ on $\Pi_s^{k_s/2}$ can be viewed as the ℓ^{∞} metric being dilated by k_s .

Let $[m]_{\infty}^{k}$ denote the set $\{0, 1, \ldots, m\}^{k}$ equipped with the metric induced by ℓ^{∞} ,

$$d_{\infty}(x, x') := \max_{0 \le j \le k-1} |x_j - x'_j|, \ x = (x_0, \dots, x_{k-1}).$$

Let \mathfrak{X} be a Banach space of non-trivial type and cotype q. Then by [MN08, Th. 1.12], there exists a constant $c(\mathfrak{X}, q)$ depending only on \mathfrak{X} and q such that the distortion of embedding of $[m]_{\infty}^{k}$ into \mathfrak{X} satisfies

$$c_{\mathfrak{X}}\left(\left[m\right]_{\infty}^{k}\right) \geq c(\mathfrak{X},q)\left(\min\left\{k^{\frac{1}{q}},m\right\}\right).$$

74

We now explain how to apply this distortion lower bound and the Austin lemma to derive an upper bound on $\alpha^*_{\mathfrak{X}}(\Delta)$. Define

$$m_s = \left\lfloor \min\left\{k_s^{\frac{1}{q}}, \frac{1}{4}l_s\right\} \right\rfloor,\,$$

and consider $\{0, 1, \ldots, m_s\}$ as elements in \mathbb{Z}_{l_s} . The grid $[m_s]_{\infty}^{k_s}$ is embedded in the group Δ via the map ϑ_s . Let $\theta := \limsup_{s \to \infty} \frac{\log l_s}{\log k_s}$, and suppose $\theta \in (0, \infty)$. For any $\epsilon > 0$, select a subsequence s_n such that $l_{s_n} \ge C_{\epsilon} k_{s_n}^{\theta - \epsilon}$ along this subsequence. To apply Lemma 5.1 using the sequence of finite metric spaces $([m_{s_n}]_{\infty}^{k_{s_n}}, d_{\infty})$, we check that

- diam_{d_{\infty} $\left(\left[m_{s_n} \right]_{\infty}^{k_{s_n}} \right) = m_{s_n} + 1;$}
- the word distance on Δ relates to the metric d_{∞} on $[m_{s_n}]^{k_{s_n}}_{\infty}$ by

$$d_{\Delta}(\vartheta_{s_n}(u,o),\vartheta_{s_n}(u',0)) \simeq_{72} k_{s_n} d_{\infty}(u,u')$$

and

$$k_{s_n} \le C'_{\epsilon} \left(\operatorname{diam}_{d_{\infty}} \left([m_{s_n}]_{\infty}^{k_{s_n}} \right); \right)^{\min\left\{q, \frac{1}{\theta - \epsilon}\right\}};$$

• the distortion of $[m_{s_n}]^{k_{s_n}}_{\infty}$ satisfies

$$c_{\mathfrak{X}}\left(\left[m_{s_{n}}\right]_{\infty}^{k_{s_{n}}}\right) \geq c(\mathfrak{X},q)m_{s_{n}} \geq \frac{c\left(\mathfrak{X},q\right)}{2}\operatorname{diam}_{d_{\infty}}\left(\left[m_{s_{n}}\right]_{\infty}^{k_{s_{n}}}\right).$$

Then by Lemma 5.1, we have that if \mathfrak{X} is a Banach space of non-trivial type and cotype q, then

(45)
$$\alpha_{\mathfrak{X}}^*(\Delta) \le 1 - \frac{1}{1 + \min\left\{q, \frac{1}{\theta}\right\}} = \max\left\{\frac{1}{1+\theta}, \frac{q}{1+q}\right\}.$$

Note that the \mathfrak{X} -distortion of the grid $[m_{s_n}]_{\infty}^{k_{s_n}}$ selected is comparable to its d_{∞} -diameter. Unlike the case with $\{\Gamma_s\}$ taken to be expanders, the size of m_s is constrained by $k_s^{\frac{1}{q}}$. We will see in Section 8.2 that when $q_{\mathfrak{X}} > 2$ and \mathfrak{X} is of Markov type p < 2, this upper bound on $\alpha_{\mathfrak{X}}^*(\Delta)$ can be improved. In the rest of this subsection we give a more detailed description of distorted elements and derive an upper bound for the compression functions.

8.1.1. A first upper bound by the metric cotype inequality.

PROPOSITION 8.2. Let ϑ_s and $\Pi_s^{k_s/2}$ be introduced as above, and suppose \mathfrak{X} is a Banach space of non-trivial type and cotype q. Then there exists a constant $C = C(\mathfrak{X}, q)$ such that for any 1-Lipschitz equivariant embedding $\varphi : \Delta \to \mathfrak{X}$,

$$\rho_{\varphi}\left(\frac{1}{8}k_s\min\left\{l_s,k_s^{\frac{1}{q}}\right\}\right) \leq C(\mathfrak{X},q)k_s.$$

This proposition improves on (45) as it applies to functions. Its proof will also be useful to "locate" the obstructions and derive a better upper bound in the next subsection.

Proof. Take $m \leq l_s/4$, and consider $0, \ldots, m+1$ as elements of $\mathbb{Z}_{l_s/2}$. By [MN08, Lemma 6.12], for each $\epsilon > 0$, \mathbb{Z}_{2m}^n equipped with an ℓ^{∞} metric embeds with distortion $1 + 6\epsilon$ into $[m+1]_{\infty}^{(\lceil 1/\epsilon \rceil + 1)n}$. Take $\epsilon = 1$, and fix a 1-Lipschitz embedding

$$\psi_s: \mathbb{Z}_{2m}^{k_s/2} \to [m+1]_{\infty}^{k_s}$$

with distortion $c(\psi_s) \leq 8$. Let \tilde{d}_{Δ} be the induced metric by d_{Δ} on $\mathbb{Z}_{2m}^{k_s/2}$

$$d_{\Delta}(u,v) = d_{\Delta} \left(\vartheta_s \circ \psi_s(u), \vartheta_s \circ \psi_s(v)\right),$$

and let $\tilde{\varphi}$ be the induced embedding

$$\tilde{\varphi} = \varphi \circ \vartheta_s \circ \psi_s : \mathbb{Z}_{2m}^{k_s/2} \to \mathfrak{X}$$

Let U_s be the uniform measure on $\mathbb{Z}_{2m}^{k_s/2}$ and σ_s be the uniform measure on $\{-1, 0, 1\}^{k_s/2}$. Let $\{\mathbf{e}_j\}_{j=0}^{k_s/2-1}$ be the standard basis of $\mathbb{R}^{k_s/2}$. Since \mathfrak{X} is a *K*-convex Banach space of cotype q, then by the metric cotype inequality in [MN08, Th. 4.2] (cited in Section 5.4),

$$(46)$$

$$\sum_{j=0}^{k_s/2-1} \sum_{u \in \mathbb{Z}_{2m}^{k_s/2}} \|\tilde{\varphi}(u) - \tilde{\varphi}(u + m\mathbf{e}_j)\|_{\mathfrak{X}}^q U_s(u)$$

$$\leq \Omega^p \sum_{u \in \mathbb{Z}_{2m}^{k_s/2}} \sum_{\varepsilon \in \{-1,0,1\}^{k_s/2}} \left\|\tilde{\varphi}(u) - \tilde{\varphi}\left(u + \sum_{j=0}^{k_s/2-1} \varepsilon_j \mathbf{e}_j\right)\right\|_{\mathfrak{X}}^q \sigma(\varepsilon) U_s(u),$$

where

$$\Omega = 5 \max\left\{ C\left(\mathfrak{X}, q\right) m, \left(\frac{k_s}{2}\right)^{\frac{1}{q}} \right\}$$

and $C(\mathfrak{X},q)$ is a constant that only depends on the cotype constant and K_q convexity constant of \mathfrak{X} .

Since φ is 1-Lipschitz, by Lemma 2.16 we have

$$\left\| \tilde{\varphi}(u,0) - \tilde{\varphi} \left(u + \sum_{j=0}^{k_s/2-1} \varepsilon_j \mathbf{e}_j, 0 \right) \right\|_{\mathfrak{X}} \leq \tilde{d}_{\Delta} \left(u + \sum_{i=0}^{k_s/2-1} \varepsilon_i \mathbf{e}_i, u \right) \leq 72k_s.$$

Plug in (46):

$$\sum_{j=0}^{k_s/2-1} \sum_{u \in (\mathbb{Z}/2m\mathbb{Z})^{k_s/2}} \|\tilde{\varphi}(u) - \tilde{\varphi}\left(u + m\mathbf{e}_j\right)\|_{\mathfrak{X}}^q U_s(u)$$
$$\leq \left(5 \max\left\{C(\mathfrak{X})m, k_s^{\frac{1}{q}}\right\}\right)^q (72k_s)^q$$

It follows that there exist $u \in (\mathbb{Z}/2m\mathbb{Z})^{k_s/2}$ and $j_0 \in \{0, \ldots, k_s/2 - 1\}$ such that

$$\|\tilde{\varphi}(u) - \tilde{\varphi}\left(u + m\mathbf{e}_{j_0}\right)\|_{\mathfrak{X}} \le 360 \max\left\{C(\mathfrak{X})mk_s^{1-\frac{1}{q}}, k_s\right\}$$

To obtain the upper bound on ρ_{φ} , choose $m = \left\lfloor \frac{1}{4} \min \left\{ l_s, k_s^{\frac{1}{q}} \right\} \right\rfloor$. By Lemma 2.16,

$$\tilde{d}_{\Delta}\left(u, u + m\mathbf{e}_{j_0}\right) \ge \frac{1}{2}mk_s,$$

and it follows that

$$\rho_{\varphi}\left(\frac{1}{2}k_{s}m\right) \leq 360C(\mathfrak{X})k_{s}.$$

Remark 8.3. Since L_1 has trivial type, embeddings $\varphi : \Delta \hookrightarrow L_1$ are not covered by the lemma. However it is true that there exists constant C > 0 such that for $\varphi : \Delta \to L_1$ a 1-Lipschitz embedding,

$$\rho_{\varphi}\left(\frac{1}{C}k_s\min\left\{l_s,k_s^{\frac{1}{2}}\right\}\right) \le Ck_s.$$

To see this, as pointed out in [MN08, Rem. 7.5], since L_1 equipped with the metric $\sqrt{\|x - y\|_1}$ is isomorphic to a subset of Hilbert space, [MN08, Th. 4.2] applied to the Hilbert space gives

$$\sum_{j=0}^{k_s/2-1} \sum_{u \in u \in (\mathbb{Z}/2m\mathbb{Z})^{k_s/2}} \|\tilde{\varphi}(u) - \tilde{\varphi}(u + m\mathbf{e}_j)\|_{L_1} U_s(u) \le C^2 \max\left\{m^2, k_s\right\}$$
$$\cdot \sum_{u \in u \in (\mathbb{Z}/2m\mathbb{Z})^{k_s/2}} \sum_{\varepsilon \in \{-1,0,1\}^{k_s}} \left\|\tilde{\varphi}(u) - \tilde{\varphi}\left(u + \sum_{j=0}^{k_s-1} \varepsilon_j \mathbf{e}_j\right)\right\|_{L_1} \sigma(\varepsilon) U_s(u),$$

which implies the stated bound.

8.2. A more refined upper bound when \mathfrak{X} has cotype > 2. In this subsection we develop an improvement of the compression upper bound in Proposition 8.2. The idea is that when $q_{\mathfrak{X}} > 2$, we can further apply the Markov type method to find obstruction in lamplighter graphs with elements in blocks of side length k_s considered as lamp configurations. The argument is similar to the one for the wreath product $H \wr \mathbb{Z}$ in Section 7.1.

We continue to use notation introduced at the beginning of Section 8.1. Let $\varphi : \Delta \to \mathfrak{X}$ be an equivariant 1-Lipschitz embedding of the group Δ into \mathfrak{X} , and assume that \mathfrak{X} is a Banach space of cotype q and non-trivial type p > 1. From the proof of Proposition 8.2, we have that there exists an element $h_0^s = \vartheta_s \left(\frac{l_s}{2} \mathbf{e}_{j_0}, 0\right)$ satisfying $|h_0^s|_{\Delta} \geq \frac{1}{4} k_s l_s$:

(47)
$$\|\varphi(h_0^s) - \varphi(e_{\Delta})\|_{\mathfrak{X}} \le C(\mathfrak{X}, q) \max\left\{l_s k_s^{1-\frac{1}{q}}, k_s\right\}.$$

The element h_0^s is in the zero section of Δ_s , and it is supported at site j_0 in the interval $[0, k_s - 1)$. Let h_j^s denote the translation of h_0^s to the block $[jk_s, (j+1)k_s)$,

$$h_j^s(x) = h_0^s(x - jk_s).$$

Consider the following subset (not a subgroup) in Δ_s :

$$L_m^s = \begin{cases} f_s \upharpoonright_{[jk_s,(j+1)k_s)} \in \left\{ \mathbf{0}, h_j^s \upharpoonright_{[jk_s,(j+1)k_s)} \right\}, \ 0 \le j \le m-1 \\ (f_s, z) : & \sup p_s \subseteq [0, mk_s) \\ & z \in \{0, k_s, \dots, (m-1)k_s\}. \end{cases} \end{cases}.$$

Again L_m^s is naturally embedded in Δ , and we identify it with its embedded image and consider L_m^s as a subset of Δ . The subset L_m^s has the structure of a lamplighter graph over a segment, and the lamp configuration is divided into blocks of side length k_s . In each block it is either identically zero or it coincides with h_j^s . As explained in Section 7.1, we can apply the Markov type method to derive a lower bound for distortion of L_m^s .

PROPOSITION 8.4. Let \mathfrak{X} be a Banach space of cotype q and Markov type p such that $2 < q < \infty$, p > 1. Then there exists a constant C > 0 such that for any 1-Lipschitz equivariant embedding $\varphi : \Delta \to \mathfrak{X}$, for each $s \in \mathbb{N}$, we have

• if $l_s \leq k_s^{\frac{1}{q}}$, then

$$\rho_{\varphi}\left(\frac{1}{4}k_{s}l_{s}\right) \leq C(\mathfrak{X},q)k_{s};$$

• if $l_s > k_s^{\frac{1}{q}}$, then

$$\rho_{\varphi}\left(\frac{1}{2C}k_{s}l_{s}\left(l_{s}k_{s}^{-\frac{1}{q}}\right)^{\frac{p}{p-1}}\right)$$

$$\leq C\left(\frac{p}{p-1}\right)^{\frac{1}{p}}M_{p}\left(\mathfrak{X}\right)C(\mathfrak{X},q)k_{s}\left(l_{s}k_{s}^{-\frac{1}{q}}\right)^{\frac{p}{p-1}}\log^{\frac{1}{p}}\left(l_{s}k_{s}^{-\frac{1}{q}}\right).$$

Proof. The case where $l_s \leq k_s^{\frac{1}{q}}$ is covered by Proposition 8.2.

In the case where $l_s > k_s^{\overline{q}}$, we apply the Markov type method. Let L_m^s be defined as above. There is a natural bijection σ_m^s between the lamplighter

graph \mathcal{L}_m over the segment $\{0, \ldots, m-1\}$ and L_m^s ; explicitly,

$$\begin{split} \sigma_m^s &: \mathcal{L}_m \to L_m^s, \\ \sigma_m^s(u,x) &= (f^u,mx) \text{ where } f^u \upharpoonright_{[jk_s,(j+1)k_s)} = \left(h_j^s\right)^{u(j)} \end{split}$$

Let \mathfrak{p}_m be the lamplighter kernel on \mathcal{L}_m defined in Appendix C. Under the bijection σ_m^s , let $K_m^s = \mathfrak{p}_m \circ (\sigma_m^s)^{-1}$ be the corresponding Markov kernel on L_m^s . Now we run the Markov chain with transition kernel K_m^s up to time $t = m \log m$. Lemma 5.5 implies

(48)

$$\begin{aligned}
\rho_{\varphi}\left(\left(\frac{1}{2}\mathbf{E}_{\pi}d_{\Delta}\left(Z_{t},Z_{0}\right)^{p}\right)^{\frac{1}{p}}\right) \\
&\leq \left(2M_{p}^{p}(\mathfrak{X})t\operatorname{diam}_{\Delta}\left(L_{m}^{s}\right)^{p}\frac{\mathbf{E}_{\pi}d_{\mathfrak{X}}\left(\varphi\left(Z_{1}\right),f\left(Z_{0}\right)\right)^{p}}{\mathbf{E}_{\pi}d_{\Delta}\left(Z_{t},Z_{0}\right)^{p}}\right)^{\frac{1}{p}},
\end{aligned}$$

where π is the stationary distribution of K_m^s and Z_t is a stationary Markov chain on L_m^s with transition kernel K_m^s .

Now we estimate the quantities that appear in the inequality. Let d_{Δ} be the metric on \mathcal{L}_m induced by word metric on Δ ,

$$d_{\Delta}(u,v) = d_{\Delta}\left(\sigma_m^s(u), \sigma_m^s(v)\right).$$

Then direct inspection shows that there exists an absolute constant C > 0 such that

$$\frac{1}{C}d_{\mathbf{w}}(u,v) \le d_{\Delta}(u,v) \le Cd_{\mathbf{w}}(u,v), \ \mathbf{w} = (k_s,k_sl_s).$$

It follows that diam_{Δ} $(L_m^s) \leq 2C (k_s + k_s l_s) m$. By Lemma C.3, we have that for $t = m \log m$,

$$\mathbf{E}_{\pi_n} \left[d_\Delta \left(Z_t, Z_0 \right)^p \right] \ge \left(c \left(k_s + k_s l_s \right) m \right)^p$$

For the other term, using Lemma C.2 and (47) when $l_s^q > k_s$, we have

$$\mathbf{E}_{\pi} \left[d_{\mathfrak{X}} \left(\varphi \left(Z_{1} \right), \Psi \left(Z_{0} \right) \right)^{p} \right] = \sum_{u, v \in X_{n}} d_{\mathfrak{X}} \left(\varphi(u), \varphi(v) \right)^{p} K_{n}(u, v) \pi_{n}(v)$$
$$\leq 3^{p} \left[C k_{s}^{p} m^{p-1} + \left(C(\mathfrak{X}) l_{s} k_{s}^{1-\frac{1}{q}} \right)^{p} \right].$$

With the choice

$$m = \left\lceil \left(l_s k_s^{-\frac{1}{q}} \right)^{\frac{p}{p-1}} \right\rceil,$$

(48) implies

$$\rho_{\Psi}\left(\frac{1}{2C}k_{s}l_{s}m\right) \leq C\left(\frac{p}{p-1}\right)^{\frac{1}{p}} M_{p}\left(\mathfrak{X}\right)C(\mathfrak{X},q)k_{s}\left(l_{s}k_{s}^{-\frac{1}{q}}\right)^{\frac{p}{p-1}}\log^{\frac{1}{p}}\left(l_{s}k_{s}^{-\frac{1}{q}}\right).$$

The upper bound on the compression function immediately yields the following upper bound on compression exponent.

COROLLARY 8.5. Let \mathfrak{X} be a Banach space of cotype q and Markov type pwith $2 < q < \infty$ and p > 1. Let Δ be the diagonal product with $\Gamma_s = D_{2l_s}$,

$$\theta := \limsup_{s \to \infty} \frac{\log l_s}{\log k_s}.$$

Then

$$\alpha_{\mathfrak{X}}^{\#}\left(\Delta\right) \leq \begin{cases} \frac{1}{1+\theta} & \text{if } \theta \leq \frac{1}{q},\\ \frac{p\theta+p-1-\frac{p}{q}}{(2p-1)\theta+p-1-\frac{p}{q}} & \text{if } \theta > \frac{1}{q}. \end{cases}$$

8.3. An explicit embedding of Δ into L_q , $q \geq 2$. We construct an embedding of Δ into L_q in two parts, similar to the embedding of wreath products in Section 7.2.

We first recall a standard embedding of finite dihedral groups into the Euclidean plane \mathbb{R}^2 . The (unlabelled) Cayley graph of D_{2l} is the same as a cycle of size 2l. One can embed it as vertices of a regular 2l-gon in the plane. For each element $\gamma \in D_{2l}$, fix a word of minimal length in a and b such that the word represents γ and starts with the letter a. Let $k(\gamma)$ be the length of such a chosen word, and let $k_a(\gamma)$ (resp. $k_b(\gamma)$) be the number of occurrence of a (resp. b) in this word. Take $\theta_l : D_{2l} \to \mathbb{R}^2$ as

$$\theta_l(\gamma) = \frac{1}{2\sin(\pi/2l)} \left(\cos\left(\frac{\pi k(\gamma)}{l}\right), \sin\left(\frac{\pi k(\gamma)}{l}\right) \right).$$

It is clear that θ_l is 1-Lipschitz and equivariant. We also consider maps to vertices of *l*-gons. Let $\theta_l^{(a)}: D_{2l} \to \mathbb{R}^2$ be the map given by

$$\theta_l^{(a)}(\gamma) = \frac{1}{2\sin\left(\pi/l\right)} \left(\cos\left(\frac{2\pi k_a(\gamma)}{l}\right), \sin\left(\frac{2\pi k_a(\gamma)}{l}\right) \right).$$

The map $\theta_l^{(b)} : D_{2l} \to \mathbb{R}^2$ is defined in the same way with $k_a(\gamma)$ replaced by $k_b(\gamma)$. Since $|k_a(\gamma) - k_b(\gamma)| \leq 1$ for any element $\gamma \in D_{2l}$, by definition of $\theta_l^{(a)}$, $\theta_l^{(b)}$ we have

$$\left\|\theta_l^{(a)}(\gamma) - \theta_l^{(b)}(\gamma)\right\|_2 \le 1.$$

Recall the classical fact that ℓ^2 embeds isometrically in L_q for all $q \ge 1$; see [AK06, Prop. 6.4.2]. For each q > 2, fix an isometric embedding $i_q : \ell^2 \to L_q$, and set $b_{\gamma,q} = i_q \circ b_{\gamma}$. Similarly, write $\theta_{l,q}^{(a)} = i_q \circ \theta_l^{(a)}$, $\theta_{l,q}^{(b)} = i_q \circ \theta_l^{(b)}$.

Direct inspection on $\theta_l^{(a)}, \theta_l^{(b)}$ shows the following.

FACT 8.6. For all $\gamma, \gamma' \in D_{2l_s}$,

$$\left\|\theta_{l_{s},2}^{(a)}(\gamma\gamma') - \theta_{l_{s},2}^{(a)}(\gamma)\right\|_{2} = \left\|\theta_{l_{s},2}^{(a)}(\gamma') - \theta_{l_{s},2}^{(a)}\left(e_{D_{2l_{s}}}\right)\right\|_{2}$$

The same equality holds with a replaced by b.

Now we introduce a weight function. For $s \in \mathbb{N}$, let $w_s : \mathbb{Z} \to [0, 1]$ be the function defined as

$$w_s(y) = \begin{cases} \frac{1}{2} & \text{for } y \le -\frac{1}{2}k_s \text{ or } y \ge \frac{3}{2}k_s, \\ \frac{|y|}{k_s} & \text{for } -\frac{1}{2}k_s < y < k_s, \\ 1 - \frac{y - k_s}{k_s} & \text{for } k_s \le y < \frac{3}{2}k_s. \end{cases}$$

It is a piecewise linear function taking value 1/2 outside $[-k_s/2, 3k_s/2]$, 0 at 0 and 1 at k_s , and its slope is in $\{\pm \frac{1}{k}, 0\}$. For $x \in \mathbb{Z}$, write $\tau_x w_s$ for the translation of w_s by x,

$$\tau_x w_s(y) = w_s(y - x).$$

Define the map $\Phi_{s,p} : \Delta_s \to L_q = \left(\bigoplus_{y \in \mathbb{Z}} (L_q)_y\right)_q$ by setting for each $y \in \mathbb{Z}$, (49)

$$\left[\Phi_{s,q}\left(f_{s},z\right)\right](y) = k_{s}^{1-\frac{1}{q}}\left((\tau_{z}w_{s})(y)\theta_{l_{s},q}^{(a)}\left(f_{s}(y)\right) + (1-(\tau_{z}w_{s})(y))\theta_{l_{s},q}^{(b)}(f_{s}(y))\right).$$

In other words, at each coordinate y, the image of $f_s(y)$ is a linear combination of $\theta_{l_s,q}^{(a)}$ and $\theta_{l_s,q}^{(b)}$ with the weights depending on the relative position between y and the cursor z.

Recall the 1-cocycle constructed in Section 6.2.3. Fix a choice of an increasing function $\gamma : \mathbb{N} \to \mathbb{R}_+$ such that $\gamma(1) = 1$, $C(\gamma) = \sum_{n=1}^{\infty} \gamma(n)^{-2} < \infty$. Let $b_{\gamma} : \Delta \hookrightarrow L_2$ be the 1-cocycle defined by (34) with p = 2 using the basic test functions. Finally define an embedding $\Phi_{\gamma,q} : \Delta \to L_q$ by

(50)
$$\Phi_{\gamma,q}\left((f_s),z\right) = \left(\bigoplus_{s=0}^{\infty} \left(\frac{1}{\gamma(s)}\Phi_{s,q}\left(f_s,z\right)\right)\right) \bigoplus b_{\gamma,q}\left((f_s),z\right),$$

where \bigoplus is direct sum in L_q .

We now check some basic properties of the map $\Phi_{\gamma,q}$.

LEMMA 8.7. Let $\gamma : \mathbb{N} \to \mathbb{R}_+$ be a function such that $\gamma(1) = 1$, $C(\gamma) = \sum_{n=1}^{\infty} \gamma(n)^{-2} < \infty$. Assume $q \ge 2$. The map $\Phi_{\gamma,q} : \Delta \to L^q$ defined in (50) is *C*-Lipschitz with *C* only depending on $C(\gamma)$.

Proof. It suffices to check that for any $u = ((f_s), z) \in \Delta$ and $s \in \{\tau, \alpha, \beta\}$ a generator, the increment $\|\Phi_{\gamma,p}(us) - \Phi_{\gamma,p}(u)\|_q$ is bounded by C.

For the generator α , $((f_s), z) \alpha = ((f'_s), z)$, where $f'_s(y) = f'_s(y)$ for all $y \neq z$ and $f'_s(z) = f_s(z)a(s)$. Recall that by definition, the weight function w_s satisfies $\tau_z w_s(z) = 0$, and the map $\theta_{l_s,q}^{(b)}$ satisfies $\theta_{l_s,q}^{(b)}(\gamma a(s)) = \theta_{l_s,q}^{(b)}(\gamma)$ for all $\gamma \in D_{2l_s}$. Then by (49),

$$\Phi_{s,q}\left(\left(f_{s},z\right)\alpha\right) = \Phi_{s,q}\left(\left(f_{s},z\right)\right).$$

Therefore in the embedding (50),

$$\left\|\Phi_{\gamma,q}\left(u\alpha\right)-\Phi_{\gamma,q}\left(u\right)\right\|_{q}=\left\|b_{\gamma}\left(u\alpha\right)-b_{\gamma}\left(u\right)\right\|_{2}\leq\sqrt{2C(\gamma)}.$$

The last inequality uses the fact that the 1-cocycle b_{γ} is Lipschitz; see Section 6.2.3. Similarly, since $\tau_z w_s (z + k_s) = 1$ and $\theta_{l_s,q}^{(a)}(\gamma b(s)) = \theta_{l_s,q}^{(a)}(\gamma)$ for all $\gamma \in D_{2l_s}$, we have $\|\Phi_{\gamma,q}(u\beta) - \Phi_{\gamma,q}(u)\|_q = \|b_{\gamma}(u\beta) - b_{\gamma}(u)\|_2$ as well. For the generator τ ,

$$\Phi_{s,q}(u\tau) = \left(k_s^{1-\frac{1}{q}} \left(w_s(y-z-1)\theta_{l_s,q}^{(a)}(f_s(y)) + (1-w_s(y-z-1))\theta_{l_s,q}^{(b)}(f_s(y))\right)\right)_{y\in\mathbb{Z}}.$$

Then

$$\begin{aligned} &\|\Phi_{s,q}(u\tau) - \Phi_{s,q}(u)\|_q^q \\ &= \sum_{y \in \mathbb{Z}} k_s^{q-1} \, |w_s(y-z-1) - w_s(y-z)|^q \, \left\|\theta_{l_s,q}^{(a)}\left(f_s(y)\right) - \theta_{l_s,q}^{(b)}\left(f_s(y)\right)\right\|_q^q. \end{aligned}$$

Recall that $\left\|\theta_{l_s}^{(a)}(\gamma) - \theta_{l_s}^{(b)}(\gamma)\right\|_2 \leq 1$ for all $\gamma \in D_{2l_s}$, $w_s(y-z) \neq w_s(y-z-1)$ only if $y-z \in \left[-\frac{k_s}{2}, \frac{3k_s}{2}\right]$, and in this interval $|w_s(y-z) - w_s(y-z-1)| = \frac{1}{k_s}$. Therefore

$$\|\Phi_{s,q}(u\tau) - \Phi_{s,q}(u)\|_q^q \le 2k_s k_s^{q-1} \left(\frac{1}{k_s}\right)^q = 2.$$

Summing up in the embedding (50),

$$\begin{split} \|\Phi_{\gamma,q}(u\tau) - \Phi_{\gamma,q}(u)\|_{q}^{q} &= \sum_{s} \frac{1}{\gamma(s)^{q}} \|\Phi_{s}(u\tau) - \Phi_{s}(u)\|_{q}^{q} + \|b_{\gamma}(u\tau) - b_{\gamma}(u)\|_{2}^{q} \\ &\leq 2C(\gamma) + (2C(\gamma))^{\frac{q}{2}}. \end{split}$$

Because of the presence of the weight function w_s , the embedding $\Phi_{\gamma,q}$: $\Delta \to L_q$ fails to be equivariant. But the increment $\|\Phi_{\gamma,q}(uv) - \Phi_{\gamma,q}(u)\|_q$ is actually comparable to $\|\Phi_{\gamma,q}(v)\|_q$.

LEMMA 8.8. There exists a constant c > 0 depending only on $C(\gamma)$ such that for $q \ge 2$,

$$\left\|\Phi_{\gamma}\left(uv\right)-\Phi_{\gamma}\left(u\right)\right\|_{q} \geq c \left\|\Phi_{\gamma}\left(v\right)\right\|_{q}.$$

Proof. By formula (49), which defines $\Phi_{s,q}$, $\Phi_{s,q}(uv)$ is

$$\left(k_s^{1-\frac{1}{q}} \left(w_s(y-z-z')\theta_{l_s,q}^{(a)}\left(f_s(y)f_s'(y-z)\right)\right. \\ \left. + (1-w_s(y-z-z'))\theta_{l_s,q}^{(b)}\left(f_s(y)f_s'(y-z)\right)\right)\right)_{y\in\mathbb{Z}}.$$

Then by the triangle inequality and Fact 8.6,

$$\begin{split} \|\Phi_{s,q}(uv) - \Phi_{s,q}(u)\|_{q} &\geq \|\Phi_{s,q}(v)\|_{q} \\ &- \left(k_{s}^{q-1}\sum_{y\in\mathbb{Z}}\left|w_{s}(y-z) - w_{s}(y-z-z')\right|^{q}\left\|\theta_{l_{s}}^{(a)}(f_{s}(y)) - \theta_{l_{s}}^{(b)}(f_{s}(y))\right\|_{2}^{q}\right)^{\frac{1}{q}}. \end{split}$$

Since $\left\|\theta_{l_s}^{(a)}(\gamma) - \theta_{l_s}^{(b)}(\gamma)\right\|_2 \le 1$ for all γ , and

$$\sum_{y \in \mathbb{Z}} |w_s(y-z) - w_s(y-z-z')|^q \le \left(\frac{1}{k_s}\right)^q \min\{|z'|, 2k_s\},\$$

we have $\|\Phi_{s,q}(uv) - \Phi_{s,q}(u)\|_2 \ge \|\Phi_{s,q}(v)\|_2 - 2^{\frac{1}{q}}$. Using $(a-b)^q_+ \ge (a/2)^q - b^q$ for $a, b \ge 0$,

$$\begin{split} \|\Phi_{\gamma,q}(uv) - \Phi_{\gamma,q}(u)\|_{q}^{q} &\geq \sum_{s} \frac{1}{\gamma(s)^{q}} \left(\|\Phi_{s,q}(v)\|_{q} - 2^{\frac{1}{q}} \right)_{+}^{q} + \|b_{\gamma,q}(v)\|_{q}^{q} \\ &\geq \frac{1}{4C(\gamma)} \left(\sum_{s} \frac{1}{\gamma(s)^{q}} \left(\frac{1}{2^{q}} \|\Phi_{s,q}(v)\|_{q}^{q} - 2 \right) \right) + \|b_{\gamma,q}(v)\|_{q}^{q} \\ &\geq \frac{1}{2^{2+q}C(\gamma)} \sum_{s} \frac{1}{\gamma(s)^{q}} \|\Phi_{s,q}(v)\|_{q}^{q} + \|b_{\gamma,q}(v)\|_{q}^{q} - \frac{1}{2} \\ &\geq \frac{1}{2^{2+q}C(\gamma)} \|\Phi_{\gamma,q}(v)\|_{q}^{q}. \end{split}$$

8.4. The compression exponent $\alpha_p^*(\Delta)$. In this subsection we estimate the L_p -compression exponent of Δ . Recall that for $p \in [1, 2]$, L_p has Markov type p and cotype 2; and for $p \in (2, \infty)$, L_p has Markov type 2 and cotype p; see [NPSS06] and references therein.

Proof of Theorem 8.1. Upper bound in (i). For the upper bound on $\alpha_p^*(\Delta)$, $p \in [1, 2]$, when $\theta = 0$, the bound is trivially true. Assume now that $\theta \in (0, \infty)$. For any $\varepsilon > 0$ sufficiently small, take a subsequence $\{s_i\}_{i \in \mathbb{N}}$ such that

$$\frac{\log l_{s_i}}{\log k_{s_i}} > \theta - \varepsilon.$$

Recall that the cotype of L_p is 2 for $p \in [1, 2]$. We apply Proposition 8.2, or the remark after it for L_1 , along this subsequence of (k_{s_i}, l_{s_i}) . We obtain the required upper bound after sending $\varepsilon \to 0$. The argument also extends to $\theta = \infty$.

Upper bound in (ii). The upper bound on $\alpha_q^*(\Delta) = \alpha_q^{\#}(\Delta)$ is covered by Corollary 8.5.

Lower bound in (ii). In the case $\theta \leq \frac{1}{q}$, $q \geq 2$, simply take the 1-cocyle b_{γ} : $\Delta \rightarrow \ell^2$ defined in (34) using the basic test functions (32), with $\gamma(n) = n^{-\frac{1}{2}-\epsilon}$. Then Lemma 6.9 implies that the compression exponent of Δ satisfies

$$\alpha_q^*(\Delta) \ge \alpha_2^*(\Delta) \ge \liminf_{s \to \infty} \frac{\log k_s}{\log \left(k_s l_s\right)} \ge \frac{1}{1+\theta}$$

Now we focus on the case $\theta > \frac{1}{q}$. Consider the explicit embedding $\Phi_{\gamma,q} : \Delta \to L_q$ defined in (50) with γ taken to be $\gamma(n) = (1+n)^{-\frac{1+\epsilon}{2}}$. By Lemma 8.7, $\Phi_{\gamma,q}$ is Lipschitz. By Lemma 8.8, there exists a constant $c = c(C(\gamma))$ such that for any $u, v \in \Delta$,

$$\left\|\Phi_{\gamma,q}(uv) - \Phi_{\gamma,q}(u)\right\|_{q} \ge c \left\|\Phi_{\gamma,q}(v)\right\|_{q}.$$

Let $v = ((f_s), z)$ be an element of Δ . At each site y, from the definition of $\theta_{l_s,q}^{(a)}$, $\theta_{l_s,q}^{(b)}$,

$$\begin{split} \left\| (\tau_z w_s)(y) \theta_{l_s,q}^{(a)} \left(f_s(y) \right) + (1 - (\tau_z w_s)(y)) \theta_{l_s,q}^{(b)}(f_s(y)) \right\|_q \\ &\geq \frac{1}{\sin(\pi/l_s)} \sin\left(\frac{\pi}{2l_s} \left(|f_s(y)|_{D_{2l_s}} - 1 \right)_+ \right) \\ &\geq \frac{1}{\pi} \left(|f_s(y)|_{D_{2l_s}} - 1 \right)_+. \end{split}$$

From the explicit formula (49), which defines $\Phi_{s,q}$,

$$\|\Phi_{s,q}(f_s,z)\|_q^q \ge \sum_{j\in\mathbb{Z}} \left(\frac{1}{\pi} k_s^{1-\frac{1}{q}} \max_{y\in I_j^s} \left(|f_s(y)|_{D_{2l_s}} - 1\right)_+\right)^q$$

In what follows we write R = |Range(v)|, and $\tilde{f}_s(y) = (|f_s(y)|_{D_{2l_s}} - 1)_+$. We have

$$\begin{split} \|\Phi_{\gamma,q}(v)\|_{q}^{q} &\geq \sum_{s \leq s_{0}(v)} \frac{1}{\gamma(s)^{q}} \sum_{j \in \mathbb{Z}} \left(\frac{1}{\pi} k_{s}^{1-\frac{1}{q}} \max_{y \in I_{j}^{s}} \tilde{f}_{s}(y) \right)^{q} + \|b_{\gamma}(v)\|_{2}^{q} \\ &\geq \sum_{s \leq s_{0}(v)} \frac{1}{\gamma(s)^{q}} \sum_{j \in \mathbb{Z}} \left(\frac{1}{\pi} k_{s}^{1-\frac{1}{q}} \max_{y \in I_{j}^{s}} \tilde{f}_{s}(y) \right)^{q} + \left(\frac{cR}{\gamma \circ \log_{2}(R)} \right)^{q}. \end{split}$$

The last step used Lemma 6.9 and the fact that since $k_{s+1} \ge 2k_s$ for all s, $s_0(v) \le \log_2 R$. Note that in the factor Δ_s , the number of intervals I_j^s with $\max_{x \in I_j^s} \tilde{f}_s(y) \ne 0$ is bounded from above by $2R/k_s$. Therefore by the Hölder inequality,

$$\sum_{j\in\mathbb{Z}}\max_{y\in I_j^s}\tilde{f}_s(y)^q \ge \left(\frac{2R}{k_s}\right)^{1-q} \left(\sum_{j\in\mathbb{Z}}\max_{y\in I_j^s}\tilde{f}_s(y)\right)^q = 2Rk_s^{-1}\ell(s)^q,$$

where

$$\ell(s) = \frac{k_s \sum_{j \in \mathbb{Z}} \max_{y \in I_j^s} \hat{f}_s(y)}{2R}.$$

Consider the following three cases:

• If $0 \le \ell(s) \le R^{\frac{1}{q}}$, then

$$\sum_{j \in \mathbb{Z}} \left(\frac{1}{2} k_s^{1-\frac{1}{q}} \max_{y \in I_j^s} \tilde{f}_s(y) \right)^q + (cR)^q \ge (cR)^q \ge (c/3)^q \left(R + 2R\ell(s) \right)^{\frac{q^2}{q+1}}.$$

• If $R^{\frac{1}{q}} \leq \ell(s) \leq R^{1-\frac{1}{q}} k_s^{\frac{2}{q}-1}$, then it is necessary that $R \leq k_s$. From the metric description in Section 2.2.3, $R \leq k_s$ implies $\sum_{j \in \mathbb{Z}} \max_{y \in I_j^s} \tilde{f}_s(y) \leq 1$. Then in this case,

$$\sum_{j \in \mathbb{Z}} \left(\frac{1}{2} k_s^{1-\frac{1}{q}} \max_{y \in I_j^s} \tilde{f}_s(y) \right)^q + (cR)^q \ge (cR)^q \ge (c/3)^q \left(R + 2R\ell(s) \right)^q.$$

• If $\ell(s) > R^{1-\frac{1}{q}} k_s^{\frac{2}{q}-1}$, from the second item we only need to consider $R > k_s$. Recall that $\ell(s) \le l_s \le C_{\varepsilon} k_s^{\theta+\varepsilon}$. It follows that $R \le \left(C_{\varepsilon} k_s^{\theta+\varepsilon+1-\frac{2}{q}}\right)^{\frac{q}{q-1}}$. Then in this case,

$$\sum_{j \in \mathbb{Z}} \left(\frac{1}{2} k_s^{1-\frac{1}{q}} \max_{y \in I_j^s} \tilde{f}_s(y) \right)^q + (cR)^q \ge \frac{1}{2^{q-1}} k_s^{q-1} R k_s^{-1} \ell(s)^q \ge c_{\varepsilon}' \left(R \ell(s) \right)^{q \gamma(\varepsilon)},$$

where

$$\gamma(\varepsilon) = \frac{\theta + \varepsilon + 1 - \frac{2}{q}}{\left(2 - \frac{1}{q}\right)\left(\theta + \varepsilon\right) + 1 - \frac{2}{q}}$$

In the last inequality, we used $\ell(s) > R^{1-\frac{1}{q}} k_s^{\frac{2}{q}-1}$ and $R \le \left(C_{\varepsilon} k_s^{\theta+\varepsilon+1-\frac{2}{q}}\right)^{\frac{q}{q-1}}$.

Note that since $\theta > \frac{1}{q}$, when ε is sufficiently small, $\gamma(\varepsilon) < \frac{q}{q+1}$. Thus the worst case is represented by the third item, and we have

$$\|\Phi_{\gamma,q}(v)\|_q^q \ge \sum_{s \le s_0(v)} \frac{c_{\varepsilon}'}{\gamma(s)^q} \left(R + R\ell(s)\right)^{q\gamma(\varepsilon)}.$$

Recall that by the metric estimate Proposition 2.14,

$$|v|_{\Delta} \le 500 \left(R + \sum_{s \le s_0(v)} k_s \sum_{j \in \mathbb{Z}} \max_{y \in I_j^s} \tilde{f}_s(y) \right) = 500 \left(R + \sum_{s \le s_0(v)} 2R\ell(s) \right),$$

and $s_0(v) \leq \log_2 R$ by Assumption 2.11. Combining with Lemma 8.8, we have that for any $u \in \Delta$,

$$\begin{split} \|\Phi_{\gamma,q}(u) - \Phi_{\gamma,q}(uv)\|_{2} &\geq \frac{c'}{\log_{2}^{\frac{1}{2}+\epsilon}(R)} \max_{s \leq s_{0}(v)} \left(R + R\ell(s)\right)^{\gamma(\varepsilon)} \\ &\geq \frac{c'}{\log_{2}^{\frac{1}{2}+\epsilon}(R)} \left(\frac{|v|_{\Delta}}{1000 \log_{2}(R)}\right)^{\gamma(\varepsilon)}, \end{split}$$

where c' > 0 is a constant depending on θ and ϵ . Sending $\varepsilon \to 0$, we conclude that when $\theta \in \left(\frac{1}{q}, \infty\right]$,

$$\alpha_q^*(\Delta) \ge \frac{\theta + 1 - \frac{2}{q}}{\left(2 - \frac{1}{q}\right)\theta + 1 - \frac{2}{q}}.$$

Note that the formula is simplified when q = 2, namely, $\alpha_2^*(\Delta) \ge \frac{2}{3}$. Combining with the fact that $\alpha_q^*(\Delta) \ge \alpha_2^*(\Delta)$, we obtain the statement.

Lower bound in (i). Since ℓ^2 embeds isometrically in all $L_p, p \ge 1$, it follows that

$$\alpha_p^*(\Delta) \ge \alpha_2^*(\Delta) \ge \max\left\{\frac{1}{1+\theta}, \frac{2}{3}\right\}.$$

This completes the proof of Theorem 8.1.

Example 8.9 (Proof of Theorem 1.4). Considering the construction of Δ with $\Gamma_s = D_{2l_s}$, the parameters $\{k_s\}$, $\{l_s\}$ are chosen to be $k_s = 2^{\beta s}$, $l_s = 2^{\iota s}$ with $\beta > 1, \iota \ge 0$. Then $\theta = \iota/\beta$.

For $p \in [1, 2]$, Theorem 8.1 implies that

$$\alpha_p^*(\Delta) = \max\left\{\frac{1}{1+\theta}, \frac{2}{3}\right\},\,$$

which can take any value in $\left[\frac{2}{3}, 1\right]$.

For q > 2, $\theta > \frac{1}{q}$, the upper and lower bound in Theorem 8.1 do not match up. But in some region of parameters we can still compare it to the Hilbert compression exponent. For $\theta \in (\frac{1}{q}, 1)$, we have

$$\alpha_q^*(\Delta) \ge \frac{\theta + 1 - \frac{2}{q}}{\left(2 - \frac{1}{q}\right)\theta + 1 - \frac{2}{q}} > \alpha_2^*(\Delta) = \max\left\{\frac{1}{1 + \theta}, \frac{2}{3}\right\}.$$

In particular, we can take $\theta = \frac{1}{2}$. Then the corresponding diagonal product Δ_1 satisfies

$$\alpha_q^*(\Delta_1) \ge \frac{3q-4}{4q-5} > \alpha_2^*(\Delta_1) = \frac{2}{3}.$$

-	-	-	

9. Discussion and some open problems

The groups of Theorem 1.1 are diagonal products of lamplighter groups. In particular, they contain many torsion elements and admit many quotients.

Problem 9.1. Find solutions to the inverse problems for speed, entropy, return probability, isoperimetric profile or compression in the class of torsion-free groups or in the class of simple groups.

In Theorem 3.8, we imposed the regularity assumption on ρ that $\rho(n)/\sqrt{n}$ is non-decreasing. This is not always satisfied; it is possible to construct examples of groups where the speed function is roughly constant over certain long time intervals. The following question asks if this regularity assumption can be dropped.

Problem 9.2. Let $\varrho : [1, \infty) \to [1, \infty)$ be a non-decreasing subadditive function satisfying $\varrho(x) \ge \sqrt{x}$ for all x. Are there a group G and a symmetric probability measure μ of finite support on G such that

$$L_{\mu}(n) \simeq \varrho(n)?$$

Proposition 3.17 only partially answers the question of what joint behavior of speed and entropy can occur. Further, the question of possible joint behavior of speed, entropy and return probability, even restricting to group of exponential volume growth, is wide open; see [Ami17, Question 6]. Solving the following problem would be a step in this direction.

Problem 9.3. Find an open set \mathcal{O} in $(0,1)^3$ such that for any point $(\alpha, \beta, \gamma) \in \mathcal{O}$, there exist a finitely generated group G and a symmetric probability measure of finite support on G, such that (α, β, γ) is the exponent of $(L_{\mu}(n), H_{\mu}(n), -\log \mu^{(2n)}(e))$.

In [Gou16], Gournay showed that if a simple random walk on G satisfies that for some C > 0, $\gamma \in (0,1)$, $\phi^{(2n)}(e) \ge \exp(-Cn^{\gamma})$, and an off-diagonal decay bound

(51)
$$\phi^{(2n)}(g) \le C\phi^{(2n)}(e) \exp\left(-\frac{C|g|^2}{n}\right) \text{ for all } g \in G, \ n \in \mathbb{N},$$

then we have the Hilbert compression exponent $\alpha_2^{\#}(G) \geq 1 - \gamma$. The offdiagonal decay assumption (51) is difficult to check in general. We illustrate a family of examples where it is not valid. In Figure 1, take the diagonal product Δ with parameters $(k_s = 2^{2s})$ and $\{\Gamma_s\}$ expanders with diam $(\Gamma_s) \simeq 2^{2\theta}$. When $\theta > 1$, we have

$$-\log q^{(2n)}(e) \simeq n^{\frac{1+\theta}{3+\theta}}, \ \alpha_2^*(\Delta) = \frac{1}{1+\theta} < 1 - \frac{1+\theta}{3+\theta}.$$

Therefore we deduce from [Gou16, Th. 1.4] that in this case the simple random walk on Δ fails the off-diagonal upper bound (51). On the other hand, we have the strict inequality

$$\alpha_{2}^{*}(\Delta) = \frac{1}{1+\theta} > \frac{1}{2+\theta} = \frac{1-\gamma}{1+\gamma}$$

showing the gap is far from the lower bound of [Goul6, Th. 1.1]. A better understanding of the relation between return probability and compression remains open.

Problem 9.4. Let G be a finitely generated infinite group such that for some $\gamma \in (0, 1)$, the simple random walk satisfies $\phi^{(2n)}(e) \ge \exp(-Cn^{\gamma})$ for some C > 0. Find the sharp lower bound for $\alpha_2^{\#}(G)$ in terms of γ and explicit examples where the bound is sharp.

In Theorem 7.1 we give an explicit formula that relates equivariant L_p compression exponents of $H \wr \mathbb{Z}$ and H when $p \in [1, 2]$. Less is known about
compression exponent of embeddings into L_p with p > 2. In particular, the
following problem is open.

Problem 9.5. For p > 2, is there an explicit formula that connects equivariant compression exponents $\alpha_p^{\#}(H \wr \mathbb{Z})$ and $\alpha_p^{\#}(H)$?

The problem of determining $\alpha_p^{\#}(\Delta)$ for Δ constructed with dihedral groups as discussed in Section 8 is related to the previous problem.

Problem 9.6. Determine the equivariant compression exponent $\alpha_p^{\#}(\Delta)$, p > 2, where Δ is the diagonal product constructed with dihedral groups $\{D_{2l_s}\}$.

Appendix A. Some auxiliary facts about excursions

In this section we recall some classical facts about local time and excursions of standard simple random walk on \mathbb{Z} . Let $\{S_k\}$ denote the standard simple random walk on \mathbb{Z} , starting at $S_0 = 0$. Let L(x, n) denote the local time of the random walk at site x,

$$L(x, n) = \# \{k : 0 < k \le n, S_k = x\}.$$

The distribution of L(x, n) is known explicitly ([Rao12, Th. 9.4]) for x = 0, 1, 2, ...

$$P_0(L(x,n) = m) = \begin{cases} \frac{1}{2^{n-m+1}} \binom{n-m+1}{(n+x)/2} & \text{if } n+x \text{ if even,} \\ \frac{1}{2^{n-m}} \binom{n-m}{(n+x-1)/2} & \text{if } n+x \text{ if odd.} \end{cases}$$

Let $\rho_0 = 0$ and $\rho_m = \min \{j > \rho_{m-1} : S_j = 0\}$. Then $\rho_1, \rho_2 - \rho_1, \ldots$ record the time duration of the excursions from 0, and they form a sequence of independent and identically distributed random variables with distribution

$$P_0(\rho_1 > 2n) = P_0(S_{2n} = 0) \sim \frac{1}{\sqrt{4\pi n}}$$

The chance that an excursion from 0 crosses k is ([Rao12, Th. 9.6])

$$P_0\left(\max_{0 \le i \le \rho_1} S_i \ge k\right) = \frac{1}{2k}, \ k = 1, 2, \dots$$

For $x \in \mathbb{Z}$, let $\rho_1(x) = \min\{j > 0: S_j = x\}$ denote the first time the random walk visits x, and let T(k, x, n) be the number of excursions away from x that cross x - k and are completed before time n. We need estimates on the moments $E_0[T(k, x, n)^q], 0 < q \leq 1$.

LEMMA A.1. There exists constant C > 0 such that for all $k, n \in \mathbb{N}, x \in \mathbb{Z}$,

$$E_0[T(k,x,n)] \le \frac{C\sqrt{n}}{k} \exp\left(-\frac{x^2}{2n}\right)$$

Proof. Let $\rho_m(x) = \min\{j > \rho_{m-1}(x) : S_j = x\}$ be the *m*-th time the random walk visits *x*. Then

$$E_0 [T(k, x, n)] \le \sum_{j \ge 0} E_0 \left[T(k, x, \rho_{2^j+1}(x)) \mathbf{1}_{\left\{\rho_{2^j}(x) \le n < \rho_{2^j+1}(x)\right\}} \right]$$
$$\le \sum_{j \ge 0} E_0 \left[T(k, x, \rho_{2^j+1}(x)) \mathbf{1}_{\left\{\rho_{2^j}(x) \le n\right\}} \right].$$

Conditioned on the event $\{\rho_{2^{j}}(x) \leq n\}$, the random variable $T(k, x, \rho_{2^{j}+1}(x))$ is stochastically dominated by a binomial random variable with parameter 2^{j+1} and $\frac{1}{2k}$. Therefore

$$\begin{split} \sum_{j\geq 0} E_0 \left[T\left(k, x, \rho_{2^j+1}(x)\right) \mathbf{1}_{\left\{\rho_{2^j}(x)\leq n\right\}} \right] \\ &\leq \sum_{j\geq 0} \frac{2^{j+1}}{2k} P_0\left(\rho_{2^j}(x)\leq n\right) \\ &\leq \sum_{j\geq 0} \frac{2^{j+1}}{2k} P_0\left(\rho_1(x)\leq n\right) P_0\left(L(n,0)\geq 2^j\right) \\ &\leq \frac{1}{2k} P_0(\rho_1(x)\leq n) E_0\left(4L(n,0)\right). \end{split}$$

Plugging in the estimates

$$P_0(\rho_1(x) \le n) = P_0\left(\max_{0 \le t \le n} S_t \ge |x|\right) \le 2\exp\left(-\frac{x^2}{2n}\right),$$
$$E_0(L(n,0)) = \sum_{t=0}^n P_0(S_t = 0) \le Cn^{\frac{1}{2}},$$

we obtain the statement.

LEMMA A.2. There exists a constant c > 0 such that for all $x \in \mathbb{Z}, k, n \in \mathbb{N}$ satisfying $k \leq c^2 n^{\frac{1}{2}}$,

$$P_0\left(T(k,x,n) \ge \frac{c\sqrt{n}}{4k}\right) \ge \frac{1}{2}P_0\left(L(x,n/2) \ge 1\right).$$

Proof. Conditioned on the event $\{\rho_1(x) = t\}, 0 \le t < n$, the distribution of T(k, x, n) is the same as T(k, 0, n - t). Therefore for any m > 0,

$$P_0(T(k,x,n) \ge m) \ge P_0(T(k,0,n/2) \ge m) P_0(L(x,n/2) \ge 1).$$

Now we show that there exists a constant c > 0 such that for $k \le c^2 n^{\frac{1}{2}}$,

$$P_0\left(T(k,0,n/2) \ge \frac{c\sqrt{n}}{4k}\right) \ge \frac{1}{2}.$$

Note that

$$\left\{T(k,0,n/2) \ge m\right\} \supset \left\{\rho_{\left\lfloor cn^{\frac{1}{2}}\right\rfloor}(0) \le \frac{n}{2}\right\} \cap \left\{T\left(k,0,\rho_{\left\lfloor cn^{\frac{1}{2}}\right\rfloor}\right) \ge m\right\}.$$

For ease of notation, in what follows write $l = \lfloor cn^{\frac{1}{2}} \rfloor$. Since $\rho_l(0)$ is sum of l independent and identically distributed random variables with distribution

$$P_0(\rho_1 > 2t) = P_0(S_{2t} = 0) \sim \frac{1}{\sqrt{4\pi t}},$$

by classical theory of sum of independent and identically distributed α -stable variables (here $\alpha = \frac{1}{2}$), there exists constant $C = C(\alpha)$ such that

$$P_0(\rho_l(0) \ge t) \le C \frac{l}{t^{\frac{1}{2}}}$$

Therefore

$$P_0\left(\rho_l(0) \ge \frac{n}{2}\right) \le \sqrt{2}Cc.$$

For the other term, $T(k, 0, \rho_l)$ is binomial with parameters l and 1/2k, and therefore by Bernstein inequality (see, for example, [Rao12, Th. 2.3]),

$$P_0\left(T\left(k,0,\rho_l\right) \le \frac{cn^{\frac{1}{2}}}{4k}\right) \le 2\exp\left(-\frac{l}{4^3k}\right).$$

Then

$$P_0\left(T(k,0,n/2) \ge \frac{c\sqrt{n}}{4k}\right) \ge 1 - \sqrt{2}Cc - 2\exp\left(-\frac{cn^{\frac{1}{2}}}{4^{3}k}\right)$$

Choosing c sufficiently small so that $\sqrt{2}Cc \leq 1/4$, $\exp\left(-\frac{1}{4^3c}\right) \leq 1/8$, we obtain the statement.

Appendix B. Approximation of functions

For $p_1, p_2 \ge 0$, consider the following space \mathcal{C}_{p_1,p_2} of continuous functions between x^{p_1} and x^{p_2} ,

$$\mathcal{C}_{p_1,p_2} = \left\{ f: [1,\infty) \to [1,\infty): \begin{array}{c} f \text{ is continuous, } f(1) = 1\\ \frac{f(x)}{x^{p_1}} \text{ is non-decreasing}\\ \frac{x^{p_2}}{f(x)} \text{ is non-increasing} \end{array} \right\}$$

Equivalently, C_{p_1,p_2} is the set of functions with f(1) = 1 satisfying

$$a^{p_1}f(x) \le f(ax) \le a^{p_2}f(x)$$
 for all $a, x \ge 1$.

We aim to approximate functions in C_{p_1,p_2} up to multiplicative constants by piecewise extremal functions.

Given two unbounded non-decreasing sequences $(k_s), (l_s)$ of real numbers, possibly finite with last value infinity, define the function

(52)
$$\tilde{f}(x) = \tilde{f}_{(k_s),(l_s)}(x) = \begin{cases} l_s & \text{for } k_s l_s \le x \le k_{s+1} l_s, \\ \frac{x}{k_{s+1}} & \text{for } k_{s+1} l_s \le x \le k_{s+1} l_{s+1}. \end{cases}$$

Similarly, define

(53)
$$\bar{f}(x) = \bar{f}_{(k_s),(l_s)}(x) = l_s + \frac{x}{k_{s+1}} \text{ for } k_s l_s \le x \le k_{s+1} l_{s+1}.$$

LEMMA B.1. For any f in $C_{0,1}$ and for any $m_0 > 1$, there exist two sequences $(k_s), (l_s)$ of real numbers, possibly finite with last value infinity, such that $k_{s+1} \ge m_0 k_s$ and $l_{s+1} \ge m_0 l_s$ for all s. The functions defined above satisfy

 $\tilde{f}(x) \simeq_{m_0} f(x)$ and $\bar{f}(x) \simeq_{2m_0} f(x)$.

Moreover, if for some $\alpha > \alpha_0 > 0$ the function $\frac{f(x)}{\log^{\alpha}(x)}$ is non-decreasing, it is possible to find such functions with sequences $(k_s), (l_s)$ satisfying $\log k_s \leq l_s^{\frac{1}{\alpha_0}}$ for all s.

Proof. The proof is best understood looking at Figure 2. Observe that by construction, $\tilde{f}(x)$ is continuous and non-decreasing, which is not necessarily true for $\bar{f}(x)$. By induction, assume k_s, l_s already known with $\tilde{f}(k_s l_s) = l_s = f(k_s l_s)$. The hypothesis on f gives that $l_s \leq f(x) \leq \frac{x}{k_s}$ for all $x \geq k_s l_s$.



Figure 2. Approximation of functions in $C_{0,1}$ by functions piecewise constant and linear, as in the proof of Lemma B.1.

We consider the minimal $y \ge m_0^2 k_s l_s$ such that $m_0 l_s \le f(y) \le \frac{y}{m_0 k_s}$. Assume first that such a y exists. By continuity of f, there are two cases.

Case I. If $f(y) = \frac{y}{m_0 k_s}$, set $k_{s+1} = m_0 k_s$ and $l_{s+1} = \frac{y}{m_0 k_s} \ge m_0 l_s$. Then for all $k_s l_s \le x \le k_{s+1} l_{s+1} = y$,

$$\frac{x}{m_0 k_s} \le \tilde{f}(x) \le \frac{x}{k_s},$$

and the same inequalities hold for f by sublinearity.

Case II. If $f(y) = m_0 l_s$, set $l_{s+1} = m_0 l_s$ and $k_{s+1} = \frac{y}{m_0 l_s} \ge m_0 k_s$. Then for all $k_s l_s \le x \le k_{s+1} l_{s+1} = y$,

$$l_s \le \tilde{f}(x) \le m_0 l_s,$$

and the same inequalities hold for f (non-decreasing).

If such a y does not exist, there are again two cases:

- either $f(x) \ge \frac{x}{m_0 k_s}$ for all $x \ge k_s l_s$ then set $k_{s+1} = m_0 k_s$ and $l_{s+1} = \infty$ so $\tilde{f}(x) = \frac{x}{m_0 k_s}$ for all $x \ge k_s l_s$, generalizing Case I;
- or $f(x) \leq m_0 l_s$ for all $x \geq k_s l_s$ then set $k_{s+1} = \infty$ so $\tilde{f}(x) = l_s$ for all $x \geq k_s l_s$, generalizing Case II.

Concerning the function \overline{f} , a routine inspection (of the intervals $[k_s l_s, k_{s+1} l_s]$ and $[k_{s+1} l_s, k_{s+1} l_{s+1}]$ in both cases) shows that $\widetilde{f}(x) \leq \overline{f}(x) \leq 2\widetilde{f}(x)$ for any x. To check the last statement, it is sufficient to check that

$$f(m_0 l_s \exp((m_0 l_s)^{\frac{1}{\alpha_0}})) \ge m_0 l_s.$$

Thus in Case II we take $l_{s+1} = m_0 l_s$ and get $y \leq l_{s+1} \exp(l_{s+1}^{\frac{1}{\alpha_0}})$ and $k_{s+1} = \frac{y}{l_{s+1}} \leq \exp(l_{s+1}^{\frac{1}{\alpha_0}})$. Our assumptions give

$$f(l_{s+1}\exp(l_{s+1}^{\frac{1}{\alpha_0}})) \ge f(k_s l_s) \left(\frac{\log(l_{s+1}\exp(l_{s+1}^{\frac{1}{\alpha_0}}))}{\log(k_s l_s)}\right)^{\alpha}$$
$$\ge l_s \left(\frac{(m_0 l_s)^{\frac{1}{\alpha_0}}}{\log k_s + \log l_s}\right)^{\alpha}$$
$$\ge l_s m_0^{\frac{\alpha}{\alpha_0}} \left(\frac{l_s^{\frac{1}{\alpha_0}}}{l_s^{\frac{1}{\alpha_0}} + \log l_s}\right)^{\alpha},$$

and the last parenthesis tends to 1 as l_s tends to infinity.

PROPOSITION B.2. Let $C_1 > 0$ and $K, L \subset [1, \infty]$ such that for any x in $[1, \infty]$, there exist $k \in K$ and $l \in L$ with $k \simeq_{C_1} x$ and $l \simeq_{C_1} x$. For any f in $C_{0,1}$ and for any $m_0 > 1$, there exist two sequences $(k_s), (l_s)$ taking values in K and L respectively such that $k_{s+1} \ge m_0 k_s$ and $l_{s+1} \ge m_0 l_s$ for all s. The functions defined in (52) and (53) satisfy

$$f(x) \simeq_{m_0 C_1^5} f(x) \text{ and } f(x) \simeq_{2m_0 C_1^5} f(x).$$

Moreover, if for some $\alpha > \alpha_0 > 0$ the function $\frac{f(x)}{\log^{\alpha}(x)}$ is non-decreasing, it is possible to find such functions with sequences $(k_s), (l_s)$ satisfying $\log k_s \leq l_s^{\frac{1}{\alpha_0}}$ for all s.

Proof. We apply Lemma B.1 with $m'_0 > C_1^2 m_0$, and obtain two sequences $(k_s), (l_s)$ of real numbers satisfying $k_{s+1} \ge m_0 C_1^2 k_s$ and $l_{s+1} \ge m_0 C_1^2 l_s$. The hypothesis on K, L permits us to find two sequences $(k'_s), (l'_s)$ with $k'_s \simeq_{C_1} k_s$ and $l'_s \simeq_{C_1} l_s$. The choice of m'_0 guarantees that $k'_{s+1} \ge m_0 k'_s$ and $l'_{s+1} \ge m_0 l'_s$.

Denote by \tilde{f}' and \bar{f}' the functions defined by (52) and (53) with the sequences $(k'_s), (l'_s)$. It is sufficient to check that $\tilde{f}'(x) \simeq_{C_1^3} \tilde{f}(x)$.

When $k'_s l'_s \leq x \leq k'_{s+1} l'_s$, then $\tilde{f}'(x) = l'_s \simeq_{C_1} l_s$. On the other hand, $\frac{k_s l_s}{C_1^2} \leq x \leq C_1^2 k_{s+1} l_s$ so

$$\frac{l_s}{C_1^2} = \frac{\tilde{f}(k_s l_s)}{C_1^2} \le \tilde{f}(\frac{k_s l_s}{C_1^2}) \le \tilde{f}(x) \le \tilde{f}(C_1^2 k_{s+1} l_s) \le C_1^2 \tilde{f}(k_{s+1} l_s) = C_1^2 l_s.$$
us $\tilde{f}'(x) \sim_{C_1^3} \tilde{f}(x)$

Thus $\tilde{f}'(x) \simeq_{C_1^3} \tilde{f}(x)$.

When
$$k'_{s+1}l'_s \leq x \leq k'_{s+1}l'_{s+1}$$
, set $x = \lambda k'_{s+1}l'_s + (1-\lambda)k'_{s+1}l'_{s+1}$. Then
 $\tilde{f}'(x) = \lambda l'_s + (1-\lambda)l'_{s+1} \simeq_{C_1} \lambda l_s + (1-\lambda)l_{s+1}$.

On the other hand, $x \simeq_{C_1^2} \lambda k_{s+1} l_s + (1-\lambda) k_{s+1} l_{s+1}$ so $\tilde{f}(x) \simeq_{C_1^2} \tilde{f}(\lambda k_{s+1} l_s + (1-\lambda) k_{s+1} l_{s+1}) = \lambda l_s + (1-\lambda) l_{s+1}$. Thus $\tilde{f}'(x) \simeq_{C_1^3} \tilde{f}(x)$.

COROLLARY B.3. Let $C_1 > 0$ and $K, L \subset [1, \infty]$ such that for any x in $[1, \infty]$, there exist $k \in K$ and $l \in L$ with $k \simeq_{C_1} x$ and $l \simeq_{C_1} x$. For any ϱ in C_{p_1,p_2} and for any $m_0 > 1$, there exist two sequences $(k_s), (l_s)$ taking values in K and L respectively such that $k_{s+1} \ge m_0 k_s$ and $l_{s+1} \ge m_0 l_s$ for all s. The function defined by

$$\bar{\varrho}(x) = x^{p_1} l_s + \frac{x^{p_2}}{k_{s+1}} \text{ for } (k_s l_s)^{\frac{1}{p_2 - p_1}} \le x \le (k_{s+1} l_{s+1})^{\frac{1}{p_2 - p_1}}$$

satisfies

$$\bar{\varrho}(x) \simeq_{2m_0 C_1^5} \varrho(x).$$

Moreover, if for any $\alpha > \alpha_0 > 0$ the function $\frac{\varrho(x)}{x^{p_1} \log^{\alpha}(x)}$ is non-decreasing, it is possible to find such functions with sequences $(k_s), (l_s)$ satisfying $\log k_s \leq l_s^{\frac{1}{\alpha_0}}$ for all s.

Proof. The application $T_{p_1,p_2}: \mathcal{C}_{0,1} \to \mathcal{C}_{p_1,p_2}$ given by

$$T_{p_1,p_2}f(x) = x^{p_1}f(x^{p_2-p_1})$$

is a bijection. Take $f = T_{p_1,p_2}^{-1} \rho$, apply Proposition B.2, and set $\bar{\rho} = T_{p_1,p_2} \bar{f}$. \Box

Appendix C. Symmetric α -stable like walk on lamplighter over a segment

For m > 1, let I_m be a subgraph of one dimensional lattice \mathbb{Z} with vertex set $\{0, 1, \ldots, m-1\}$. Consider the lamplighter graph \mathcal{L}_m over the segment I_m . Formally, $\{(f, x) : f : I_m \to \{0, 1\}, x \in I_m\}$ is the vertex set of \mathcal{L}_m , and an edge connects (f, x) and (f', x') if $f \equiv f'$ and $x \sim x'$ in I_m , or $x = x', f \neq f'$ and f differs from f' only at site x. Random walks on lamplighter graphs have been studied in [Ers03], [PR04]; see also references therein.

Random walk on \mathbb{Z} driven by step distribution

$$\nu_{\alpha}(x) \simeq \frac{1}{(1+|x|)^{1+\alpha}}, \ \alpha \in (0,2)$$

is often referred to as a symmetric α -stable like walk on Z. By abuse of terminology, we call it an α -stable walk. General theory regarding heat kernel estimates of α -stable walks on volume doubling graphs is available; see [CK08] and references therein. The connection between α -stable walks and the Markov type method for a bounding compression exponent was first introduced in [NP11], where the *p*-stable walk on \mathbb{Z} is used to determine the L_p -compression exponent of $\mathbb{Z} \wr \mathbb{Z}$, p > 1. For our purpose, we focus on the case of the stable walk of the index $\alpha = 1$ on the base graph.

On I_m , define transition kernel

$$\zeta_m(x, x') = \frac{c_{x,x'}}{\sum_{x' \in I_m} c_{x,x'}}, \ x, x' \in I_m,$$

where

$$c_{x,x'} = \frac{1}{1 + |x - x'|^2}$$

Then ζ_m is a Markov transition kernel on I_m with stationary distribution $\mathcal{C}_m(x) = \frac{\sum_{x' \in I_m} c_{x,x'}}{\sum_{x,x' \in I_m} c_{x,x'}}$. One readily checks that $\frac{1}{5m} \leq \mathcal{C}_m(x) \leq \frac{5}{m}$ for all $x \in I_m$. Now consider a random walk on the lamplighter graph \mathcal{L}_m with transition

Now consider a random walk on the lamplighter graph \mathcal{L}_m with transition ζ_m on the base. Let \mathfrak{p}_m be the transition kernel in \mathcal{L}_m such that for $x \neq x'$, $\mathfrak{p}_m((f,x),(f',x')) = \frac{1}{4}\zeta_m(x,x')$ if f(z) = f'(z) for all $z \notin \{x,x'\}$; for x = x', $\mathfrak{p}_m((f,x),(f',x')) = \frac{1}{2}\zeta_m(x,x)$ if f(z) = f'(z) for all $z \neq x$. In other words, in each step the walker first randomizes the lamp configuration at the current location, then moves according to the transition kernel ζ_m and randomizes the lamp at the arrival location. This Markov chain is reversible with stationary distribution

$$U_m(f, x) = 2^{-m} \mathcal{C}_m(x).$$

From the upper bound on relaxation time in [PR04, Th. 1.2], we have

$$T_{rel}(\mathcal{L}_m, \mathfrak{p}_m) = rac{1}{\lambda_2(\mathcal{L}_m, \mathfrak{p}_m)} \leq \max_{x, y \in I_m} \mathbf{E}_x \tau_y,$$

where $\tau_y = \min\{t : X_t = y\}$, X_t denotes the Markov chain on \mathcal{L}_m with transition kernel \mathfrak{p}_m . Note that although in the statement of [PR04, Th. 1.2] it is assumed that the Markov chain on the base is a lazy simple random walk on a transitive graph, the coupling argument that proves the relaxation time upper bound is completely general, and it applies to any reversible Markov chain on the base graph. The quantity $\max_{x,y\in I_m} \mathbf{E}_x \tau_y$ is known as the maximal hitting time of the chain \mathfrak{p}_m . By [AF02, Lemma 4.1],

$$\max_{x,y\in I_m} \mathbf{E}_x \tau_y = \frac{1}{2} \left(\sum_{z,z'\in I_m} c_{z,z'} \right) \max_{x,y} R_{x,x'},$$

where $R_{x,x'}$ denotes the effective resistance between vertices x and x' in the electric network on I_m with edge conductances $c_{z,z'}$ between the pair of vertices z, z'. Estimates of effective resistance of α -stable walks follow from classical methods. For the particular transition kernel ζ_m with $\alpha = 1$, there exists a constant C > 0

$$R_{x,x'} \le C \log \left| x - x' \right|;$$

see, for example, [CFG09, App. B.2]. We conclude that for the Markov chain with transition kernel \mathfrak{p}_m on \mathcal{L}_m ,

$$\lambda_2(\mathcal{L}_m, \mathfrak{p}_m) \ge \frac{c}{m \log m}.$$

Equivalently, we have the following Poincaré inequality: for any function $f : \mathcal{L}_m \to \mathbb{R}$,

$$\sum_{u,v \in \mathcal{L}_m} (f(u) - f(v))^2 U_m(u) U_m(v)$$

$$\leq \frac{2m \log m}{c} \sum_{u,v \in \mathcal{L}_m} (f(u) - f(v))^2 \mathfrak{p}_m(u,v) U_m(u).$$

By Matoušek's extrapolation lemma for Poincaré inequalities (see [Mat97] and [NS11, Lemma 4.4]), we deduce the following l_p -Poincaré inequalities.

LEMMA C.1. In the setting introduced above, there exists an absolute constant C > 0 such that for any $f : \mathcal{L}_m \to \ell^p$,

• if $1 \le p \le 2$, then

$$\sum_{u,v\in\mathcal{L}_m} \|f(u) - f(v)\|_p^p U_m(u) U_m(v)$$

$$\leq Cm \log m \sum_{u,v\in\mathcal{L}_m} \|f(u) - f(v)\|_p^p U_m(u)\mathfrak{p}_m(u,v);$$

• if p > 2, then

$$\sum_{u,v \in \mathcal{L}_m} \|f(u) - f(v)\|_p^p U_m(u) U_m(v) \\ \leq (Cm \log m)^{\frac{p}{2}} (2p)^p \sum_{u,v \in \mathcal{L}_m} \|f(u) - f(v)\|_p^p U_m(u)\mathfrak{p}_m(u,v).$$

Now we introduce a distance function on \mathcal{L}_m . Let $\mathbf{w} = (w_1, w_2) \in \mathbb{R}_2^+$. In the lamplighter graph \mathcal{L}_m , let $w(e) = w_1$ if the edge e connects (f, x) and (f, x'), where $x \sim x'$ (edges of first type); let $w(e) = w_2$ if the edge e connects (f, x) and (f', x), where f, f' only differs at site x (edges of second type). Define $d_{\mathbf{w}}$ to be distance on \mathcal{L}_m

$$d_{\mathbf{w}}(u,v) = \min\left\{\sum_{e \in P} w(e): P \text{ is a path in } \mathcal{L}_m \text{ connecting } u, v\right\}.$$

From definition of \mathfrak{p}_m , it is straightforward to check that the following.

LEMMA C.2. There exists a constant C > 0 such that for all p > 1,

$$\sum_{u,v\in\mathcal{L}_m} d_{\mathbf{w}}(u,v)^p U_m(u)\mathfrak{p}_m(u,v) \le C\left(w_2^p + \frac{1}{p-1}w_1^p m^{p-1}\right)$$
$$\sum_{d_{\mathbf{w}}(u,v)\ge\frac{1}{4}(w_2+w_1)m} d_{\mathbf{w}}(u,v)^p U_m(u)U_m(v) \ge \frac{1}{C}\left(w_2+w_1\right)m.$$

These ingredients allow us to carry out the Poincaré inequality method to the upper bound L_p -compression function of $H \wr \mathbb{Z}$. It can also be used in the study of the diagonal product Δ with dihedral groups. Alternatively, we may apply the Markov type method. To this end, the following speed lower estimate is needed.

LEMMA C.3. Let X_t be a stationary Markov chain on \mathcal{L}_m with transition kernel \mathfrak{p}_m reversible with stationary measure U_m . Then there exists c > 0 such that

$$\mathbf{E}_{U_m} \left[d_{\mathbf{w}} \left(X_t, X_0 \right) \right] \ge c \left(w_1 + w_2 \right) \frac{t}{\log_* t} \text{ for all } 1 \le t \le m \log m.$$

Proof. Let $S_{[0,t]} = \{S_n, 0 \le n \le t\}$ denote the sites visited by the induced random walk $\{S_t\}$ on I_m . Since in each step the chain randomizes the lamp at the current and new locations, and any path in the graph \mathcal{L}_m that connects X_0 to X_t must visit all the sites where the lamp configurations of X_0 and X_t differ, we have that for any $u \in \mathcal{L}_m$,

$$\mathbf{E}_{u}\left[d_{\mathbf{w}}\left(X_{t}, X_{0}\right)\right] \geq \frac{1}{2}(w_{1} + w_{2})\mathbf{E}_{u}\left[\left|S_{[0,t]}\right|\right].$$

Thus the question is reduced to the range of the ζ_m -random walk on the base I_m . Methods to estimate the expected size of range of the random walk go back to Dvoretzky and Erdos [DE51]. Here we include a straightforward adaptation of the argument in [NP08, Lemma 6.3] for completeness.

In what follows \mathbb{E} means taking expectation with the law of random walk S_n on I_m with step distribution ζ_m . For any $k \in \{1, \ldots, m\}$, denote by V_1, \ldots, V_k the first k elements of I_m that are visited by the random walk S_n . Let

$$Y_k(t) = |\{0 \le n \le t : S_n \in \{V_1, \dots, V_k\}\}|.$$

Note that $\{Y_k(t) < t+1\} = \{|S_{[0,t]}| > k\}$. For any starting point $z \in I_m$,

$$\mathbb{E}_{z}\left[Y_{k}(t)\right] = \sum_{l=1}^{k} \mathbb{E}_{z}\left[\left|\left\{0 \le n \le t : S_{n} = V_{l}\right\}\right|\right] \le k \sum_{n=0}^{t} \max_{x \in I_{m}} \mathbb{P}_{x}\left(S_{n} = x\right).$$

Therefore

(54)
$$\mathbb{E}_{z}\left(\left|S_{[0,t]}\right|\right) \geq k\mathbb{P}_{z}\left(\left|S_{[0,t]}\right| > k\right) \geq k\left(1 - \frac{\mathbb{E}_{z}\left[Y_{k}(t)\right]}{t+1}\right)$$
$$\geq k\left(1 - \frac{k\sum_{n=0}^{t}\max_{x \in I_{m}}\mathbb{P}_{x}\left(S_{n}=x\right)}{t+1}\right).$$

The argument in [CK08, Th. 3.1] implies that the chain (I_m, ζ_m) satisfies a Nash inequality that there exists an absolute constant C > 0,

$$\theta\left(\|u\|_{2}^{2}\right) \leq C\mathcal{E}_{\zeta_{m}}\left(u,u\right) \text{ for all } u:I_{m} \to \mathbb{R}, \text{ where } \theta(r)=r^{2}.$$

This Nash inequality implies on-diagonal decay upper bound (see [DSC96])

$$\mathbb{P}_x(S_l = x) \le \frac{c_2}{l}$$
 for all $l \le m, x \in I_m$.

For l > m, by monotonicity, $\mathbf{P}_x(S_l = x) \leq \mathbf{P}_x(S_m = x) \leq c_2 l^{-1}$. It follows that for $k \in \{1, \ldots, m\}$,

$$\sum_{l=0}^{k \log k} \max_{x \in I_m} \mathbb{P}_x \left[S_l = x \right] \le c_3 \log k.$$

Together these estimates imply that

$$\mathbf{E}_{u}\left[d_{\mathbf{w}}\left(X_{k\log_{*}k}, X_{0}\right)\right] \geq c\left(w_{1}+w_{2}\right)k \text{ for any } k \in \{1, \dots, m\}, u \in \mathcal{L}_{m}. \quad \Box$$

The same method can be used to estimate the speed of random walks on lamplighter graphs over other choices of the base graph.

LEMMA C.4. Let $\Gamma = \mathbb{Z}_2 \wr D_{\infty}^d$, $d \ge 3$ as in the second item of *Example* 2.4, marked with generating subgroups $A = \mathbb{Z}_2 \wr \langle a_j, 1 \le j \le d \rangle$, $B = \mathbb{Z}_2 \wr \langle b_j, 1 \le j \le d \rangle$. Fix an increasing sequence $n_s \in \mathbb{N}$, and let $\Gamma_s = \mathbb{Z}_2 \wr D_{2n_s}^d$ be a finite quotient of Γ . Let A(s), B(s) denote the projection of A and B to Γ_s . There exists a constant $\sigma_d > 0$ only depending on d such that $\{\Gamma_s\}$ satisfies the $(\sigma_d, (2n_s)^d)$ -linear speed assumption.

Proof. Let $\bar{A}(s) = \langle a_j, 1 \leq j \leq d \rangle$, $\bar{B}(s) = \langle b_j, 1 \leq j \leq d \rangle$. Consider a random alternating word $X_t^{(s)}$ in A(s) and B(s) of length t, and let $\bar{X}_t^{(s)}$ be its projection to $D_{2n_s}^d$. In other words, if the last letter in $X_t^{(s)}$ is a random element in A(s), then to get to $X_{t+1}^{(s)}$, the lamp configuration in the neighborhood $\bar{X}_t^{(s)}\bar{B}(s)$ is randomized, and the walker on the base $D_{2n_s}^d$ is multiplied by a random element in $\bar{B}(s)$. Similarly if $X_t^{(s)}$ ends with B(s), then the next move is uniform in A(s). From this description, we have that the lamps over the sites visited by $\bar{X}_t^{(s)}$ are randomized,

$$\left| X_{2t}^{(s)} \right|_{\Gamma_s} \ge \frac{1}{8} \left| \mathcal{R}_{[[0,2t]]}^{(s)} \right|,$$

where $\mathcal{R}_{[[0,2t]]}^{(s)} = \left\{ x \in D_{2n_s}^d : \bar{X}_{2l}^{(s)} = x \text{ for some } 0 \leq l \leq t \right\}$. Comparing $\left\{ \bar{X}_{2l}^{(s)} \right\}$ to the standard simple random walk on $D_{2n_s}^d$, we have that there exists a constant $C_d > 0$ such that

$$\mathbf{P}\left(\bar{X}_{2l}^{(s)} = e\right) \le C_d(2l)^{-\frac{d}{2}} \text{ for all } 1 \le l \le 2n_s^2.$$

It follows that

(

$$\sum_{l=0}^{2n_s)^d} \mathbf{P}\left(\bar{X}_{2l}^{(s)} = e\right) \le 1 + \sum_{l=1}^{2n_s^2} C_d (2l)^{-\frac{d}{2}} + (2n_s)^d (4n_s^2)^{-d/2} \le 2C_d + 2.$$

To estimate $\mathbf{E} \left| \mathcal{R}_{[[0,2t]]}^{(s)} \right|$, we apply the argument as in Lemma C.3. For $t \leq (2n_s)^d$, in the inequality (54) choose $k = \frac{t}{4C_d+4}$. We conclude that there exists a constant $\sigma_d > 0$,

$$\mathbf{E} \left| X_{2t}^{(s)} \right|_{\Gamma_s} \ge \sigma_d t.$$

References

- [AK06] F. ALBIAC and N. J. KALTON, Topics in Banach Space Theory, Grad. Texts in Math. 233, Springer-Verlag, New York, 2006. MR 2192298.
 Zbl 1094.46002. https://doi.org/10.1007/0-387-28142-8.
- [AF02] D. ALDOUS and J. A. FILL, Reversible Markov chains and random walks on graphs, 2002, available on D. Aldous' website.
- [Ami17] G. AMIR, On the joint behaviour of speed and entropy of random walks on groups, *Groups Geom. Dyn.* 11 no. 2 (2017), 455–467. MR 3668047.
 Zbl 1368.20021. https://doi.org/10.4171/GGD/403.
- [AV17] G. AMIR and B. VIRÁG, Speed exponents of random walks on groups, Int. Math. Res. Not. IMRN no. 9 (2017), 2567–2598. MR 3658209. Zbl 1405.
 60007. https://doi.org/10.1093/imrn/rnv378.
- [AGS06] G. N. ARZHANTSEVA, V. S. GUBA, and M. V. SAPIR, Metrics on diagram groups and uniform embeddings in a Hilbert space, *Comment. Math. Helv.* 81 no. 4 (2006), 911–929. MR 2271228. Zbl 1166.20031. https://doi.org/10.4171/CMH/80.
- [ADS09] G. ARZHANTSEVA, C. DRUŢU, and M. SAPIR, Compression functions of uniform embeddings of groups into Hilbert and Banach spaces, J. Reine Angew. Math. 633 (2009), 213–235. MR 2561202. Zbl 1258.20032. https: //doi.org/10.1515/CRELLE.2009.066.
- [Aus11] T. AUSTIN, Amenable groups with very poor compression into Lebesgue spaces, *Duke Math. J.* 159 no. 2 (2011), 187–222. MR 2824482. Zbl 1226.
 43001. https://doi.org/10.1215/00127094-1415860.
- [ANP09] T. AUSTIN, A. NAOR, and Y. PERES, The wreath product of \mathbb{Z} with \mathbb{Z} has Hilbert compression exponent $\frac{2}{3}$, *Proc. Amer. Math. Soc.* **137** no. 1

(2009), 85–90. MR 2439428. Zbl 1226.20032. https://doi.org/10.1090/S0002-9939-08-09501-4.

- [Ave76] A. AVEZ, Harmonic functions on groups, in Differential Geometry and Relativity, Math. Phys. Appl. Math. 3, 1976, pp. 27–32. MR 0507229.
 Zbl 0345.31004. https://doi.org/10.1007/978-94-010-1508-0_4.
- [BFGM07] U. BADER, A. FURMAN, T. GELANDER, and N. MONOD, Property (T) and rigidity for actions on Banach spaces, Acta Math. 198 no. 1 (2007), 57–105. MR 2316269. Zbl 1162.22005. https://doi.org/10.1007/ s11511-007-0013-0.
- [Bal92] K. BALL, Markov chains, Riesz transforms and Lipschitz maps, Geom. Funct. Anal. 2 no. 2 (1992), 137–172. MR 1159828. Zbl 0788.46050. https: //doi.org/10.1007/BF01896971.
- [BE14] L. BARTHOLDI and A. ERSCHLER, Groups of given intermediate word growth, Ann. Inst. Fourier (Grenoble) 64 no. 5 (2014), 2003–2036. MR 3330929. Zbl 1317.20043. https://doi.org/10.5802/aif.2902.
- [BE17] L. BARTHOLDI and A. ERSCHLER, Distortion of imbeddings of groups of intermediate growth into metric spaces, *Proc. Amer. Math. Soc.* 145 no. 5 (2017), 1943–1952. MR 3611311. Zbl 1431.20027. https://doi.org/ 10.1090/proc/13453.
- [Bau72] G. BAUMSLAG, On finitely presented metabelian groups, Bull. Amer. Math. Soc. 78 (1972), 279. MR 0291260. Zbl 0253.20052. https://doi. org/10.1090/S0002-9904-1972-12958-6.
- [BdlHV08] B. BEKKA, P. DE LA HARPE, and A. VALETTE, Kazhdan's Property (T), New Math. Monogr. 11, Cambridge Univ. Press, Cambridge, 2008. MR 2415834. Zbl 1146.22009. https://doi.org/10.1017/ CBO9780511542749.
- [BD06] G. C. BELL and A. N. DRANISHNIKOV, A Hurewicz-type theorem for asymptotic dimension and applications to geometric group theory, *Trans. Amer. Math. Soc.* **358** no. 11 (2006), 4749–4764. MR 2231870. Zbl 1117. 20032. https://doi.org/10.1090/S0002-9947-06-04088-8.
- [BPS12] A. BENDIKOV, C. PITTET, and R. SAUER, Spectral distribution and L²isoperimetric profile of Laplace operators on groups, Math. Ann. 354 no. 1 (2012), 43–72. MR 2957618. Zbl 1268.43001. https://doi.org/10.1007/ s00208-011-0724-6.
- [BH97] S. G. BOBKOV and C. HOUDRÉ, Isoperimetric constants for product probability measures, Ann. Probab. 25 no. 1 (1997), 184–205. MR 1428505.
 Zbl 0878.60013. https://doi.org/10.1214/aop/1024404284.
- [Bri13] J. BRIEUSSEL, Behaviors of entropy on finitely generated groups, Ann. Probab. 41 no. 6 (2013), 4116–4161. MR 3161471. Zbl 1280.05123. https: //doi.org/10.1214/12-AOP761.
- [Bri14] J. BRIEUSSEL, Growth behaviors in the range $e^{r^{\alpha}}$, Afr. Mat. **25** no. 4 (2014), 1143–1163. MR 3277875. Zbl 1337.20041. https://doi.org/10. 1007/s13370-013-0182-2.

SPEED OF RANDOM WALKS

- [CFG09] P. CAPUTO, A. FAGGIONATO, and A. GAUDILLIÈRE, Recurrence and transience for long range reversible random walks on a random point process, *Electron. J. Probab.* 14 no. 90 (2009), 2580–2616. MR 2570012. Zbl 1191.60120. https://doi.org/10.1214/EJP.v14-721.
- [CK08] Z.-Q. CHEN and T. KUMAGAI, Heat kernel estimates for jump processes of mixed types on metric measure spaces, *Probab. Theory Related Fields* 140 no. 1-2 (2008), 277–317. MR 2357678. Zbl 1131.60076. https://doi. org/10.1007/s00440-007-0070-5.
- [CSV12] Y. CORNULIER, Y. STALDER, and A. VALETTE, Proper actions of wreath products and generalizations, *Trans. Amer. Math. Soc.* **364** no. 6 (2012), 3159–3184. MR 2888241. Zbl 1283.20049. https://doi.org/10. 1090/S0002-9947-2012-05475-4.
- [CT13] Y. CORNULIER and R. TESSERA, Dehn function and asymptotic cones of Abels' group, J. Topol. 6 no. 4 (2013), 982–1008. MR 3145147. Zbl 1311.
 20034. https://doi.org/10.1112/jtopol/jtt019.
- [dCTV07] Y. DE CORNULIER, R. TESSERA, and A. VALETTE, Isometric group actions on Hilbert spaces: growth of cocycles, *Geom. Funct. Anal.* 17 no. 3 (2007), 770–792. MR 2346274. Zbl 1129.22004. https://doi.org/10.1007/ s00039-007-0604-0.
- [Cou96] T. COULHON, Ultracontractivity and Nash type inequalities, J. Funct. Anal. 141 no. 2 (1996), 510–539. MR 1418518. Zbl 0887.58009. https: //doi.org/10.1006/jfan.1996.0140.
- [CG97] T. COULHON and A. GRIGOR'YAN, On-diagonal lower bounds for heat kernels and Markov chains, *Duke Math. J.* 89 no. 1 (1997), 133–199. MR 1458975. Zbl 0920.58064. https://doi.org/10.1215/ S0012-7094-97-08908-0.
- [CSC93] T. COULHON and L. SALOFF-COSTE, Isopérimétrie pour les groupes et les variétés, Rev. Mat. Iberoamericana 9 no. 2 (1993), 293–314. MR 1232845. Zbl 0782.53066. https://doi.org/10.4171/RMI/138.
- [CSC95] T. COULHON and L. SALOFF-COSTE, Variétés riemanniennes isométriques à l'infini, *Rev. Mat. Iberoamericana* 11 no. 3 (1995), 687–726. MR 1363211.
 Zbl 0845.58054. https://doi.org/10.4171/RMI/190.
- [Der76] Y. DERRIENNIC, Lois "zéro ou deux" pour les processus de Markov. Applications aux marches aléatoires, Ann. Inst. H. Poincaré Sect. B (N.S.)
 12 no. 2 (1976), 111–129. MR 0423532. Zbl 0353.60075.
- [DSC96] P. DIACONIS and L. SALOFF-COSTE, Nash inequalities for finite Markov chains, J. Theoret. Probab. 9 no. 2 (1996), 459–510. MR 1385408.
 Zbl 0870.60064. https://doi.org/10.1007/BF02214660.
- [DE51] A. DVORETZKY and P. ERDÖS, Some problems on random walk in space, in Proceedings of the Second Berkeley Symposium on Mathematical Statistics and Probability, 1950, Univ. of California Press, Berkeley and Los Angeles, 1951, pp. 353–367. MR 0047272. Zbl 0044.14001. Available at https://projecteuclid.org/euclid.bsmsp/1200500239.

JÉRÉMIE BRIEUSSEL and TIANYI ZHENG

- [Enf69] P. ENFLO, On the nonexistence of uniform homeomorphisms between L_p -spaces, Ark. Mat. 8 (1969), 103–105. MR 0271719. Zbl 0196.14002. https: //doi.org/10.1007/BF02589549.
- [Ers03] A. ERSCHLER, On drift and entropy growth for random walks on groups, *Ann. Probab.* **31** no. 3 (2003), 1193–1204. MR 1988468. Zbl 1043.60005. <u>https://doi.org/10.1214/aop/1055425775</u>.
- [Ers06] A. ERSCHLER, Piecewise automatic groups, Duke Math. J. 134 no. 3 (2006), 591–613. MR 2254627. Zbl 1159.20019. https://doi.org/10.1215/S0012-7094-06-13435-X.
- [Ers10] A. ERSCHLER, Poisson-Furstenberg boundaries, large-scale geometry and growth of groups, in *Proceedings of the International Congress of Mathematicians. Volume II*, Hindustan Book Agency, New Delhi, 2010, pp. 681–704. MR 2827814. Zbl 1258.60009. https://doi.org/10.1142/ 9789814324359 0069.
- [EJZ10] M. ERSHOV and A. JAIKIN-ZAPIRAIN, Property (T) for noncommutative universal lattices, *Invent. Math.* **179** no. 2 (2010), 303–347. MR 2570119.
 Zbl 1205.22003. https://doi.org/10.1007/s00222-009-0218-2.
- [EFW12] A. ESKIN, D. FISHER, and K. WHYTE, Coarse differentiation of quasiisometries I: Spaces not quasi-isometric to Cayley graphs, Ann. of Math.
 (2) 176 no. 1 (2012), 221–260. MR 2925383. Zbl 1264.22005. https://doi. org/10.4007/annals.2012.176.1.3.
- [EFW13] A. ESKIN, D. FISHER, and K. WHYTE, Coarse differentiation of quasiisometries II: Rigidity for Sol and lamplighter groups, Ann. of Math. (2) 177 no. 3 (2013), 869–910. MR 3034290. Zbl 1398.22012. https://doi. org/10.4007/annals.2013.177.3.2.
- [FM98] B. FARB and L. MOSHER, A rigidity theorem for the solvable Baumslag-Solitar groups, *Invent. Math.* 131 no. 2 (1998), 419–451, With an appendix by Daryl Cooper. MR 1608595. Zbl 0937.22003. https://doi.org/10.1007/ s002220050210.
- [FM99] B. FARB and L. MOSHER, Quasi-isometric rigidity for the solvable Baumslag-Solitar groups. II, *Invent. Math.* 137 no. 3 (1999), 613–649.
 MR 1709862. Zbl 0931.20035. https://doi.org/10.1007/s002220050337.
- [FM00] B. FARB and L. MOSHER, Problems on the geometry of finitely generated solvable groups, in *Crystallographic Groups and their Generalizations (Kortrijk*, 1999), *Contemp. Math.* 262, Amer. Math. Soc., Providence, RI, 2000, pp. 121–134. MR 1796128. Zbl 0983.20038. https: //doi.org/10.1090/conm/262/04170.
- [Gou16] A. GOURNAY, The Liouville property and Hilbertian compression, Ann. Inst. Fourier (Grenoble) 66 no. 6 (2016), 2435–2454. MR 3580176. Zbl 1397.20062. https://doi.org/10.5802/aif.3067.
- [Gri85] R. I. GRIGORCHUK, Degrees of growth of finitely generated groups and the theory of invariant means, *Izv. Math.* 25 no. 2 (1985), 259–300. MR 764305. Zbl 0583.20023. https://doi.org/10.1070/ IM1985v025n02ABEH001281.

SPEED OF RANDOM WALKS

- [GN12] R. I. GRIGORCHUK and P. W. NOWAK, Diameters, distortion, and eigenvalues, *European J. Combin.* 33 no. 7 (2012), 1574–1587. MR 2923470.
 Zbl 1246.05046. https://doi.org/10.1016/j.ejc.2012.03.019.
- [Gro81] M. GROMOV, Groups of polynomial growth and expanding maps, Inst. Hautes Études Sci. Publ. Math. no. 53 (1981), 53–73. MR 0623534. Zbl 0474.20018. https://doi.org/10.1007/BF02698687.
- [Gro93] M. GROMOV, Asymptotic invariants of infinite groups, in *Geometric Group Theory*, Vol. 2 (Sussex, 1991), London Math. Soc. Lecture Note Ser. 182, Cambridge Univ. Press, Cambridge, 1993, available on author's webpage, pp. 1–295. MR 1253544. Zbl 0841.20039.
- [GK04] E. GUENTNER and J. KAMINKER, Exactness and uniform embeddability of discrete groups, J. London Math. Soc. (2) 70 no. 3 (2004), 703–718. MR 2160829. Zbl 1082.46049. https://doi.org/10.1112/ S0024610704005897.
- [HLW06] S. HOORY, N. LINIAL, and A. WIGDERSON, Expander graphs and their applications, Bull. Amer. Math. Soc. (N.S.) 43 no. 4 (2006), 439–561. MR 2247919. Zbl 1147.68608. https://doi.org/10.1090/ S0273-0979-06-01126-8.
- [JV14] P.-N. JOLISSAINT and A. VALETTE, L^p-distortion and p-spectral gap of finite graphs, Bull. Lond. Math. Soc. 46 no. 2 (2014), 329–341. MR 3194751.
 Zbl 1297.46021. https://doi.org/10.1112/blms/bdt096.
- [KY06] G. KASPAROV and G. YU, The coarse geometric Novikov conjecture and uniform convexity, Adv. Math. 206 no. 1 (2006), 1–56. MR 2261750.
 Zbl 1102.19003. https://doi.org/10.1016/j.aim.2005.08.004.
- [KP13] M. KASSABOV and I. PAK, Groups of oscillating intermediate growth, Ann. of Math. (2) 177 no. 3 (2013), 1113–1145. MR 3034295. Zbl 1283.
 20027. https://doi.org/10.4007/annals.2013.177.3.7.
- [KV83] V. A. KAĬMANOVICH and A. M. VERSHIK, Random walks on discrete groups: boundary and entropy, Ann. Probab. 11 no. 3 (1983), 457–490.
 MR 0704539. Zbl 0641.60009. https://doi.org/10.1214/aop/1176993497.
- [Kes59] H. KESTEN, Full Banach mean values on countable groups, Math. Scand. 7 (1959), 146–156. MR 0112053. Zbl 0092.26704. https://doi.org/10.7146/ math.scand.a-10568.
- [Laf08] V. LAFFORGUE, Un renforcement de la propriété (T), Duke Math. J. 143 no. 3 (2008), 559–602. MR 2423763. Zbl 1158.46049. https://doi.org/10. 1215/00127094-2008-029.
- [LS88] G. F. LAWLER and A. D. SOKAL, Bounds on the L² spectrum for Markov chains and Markov processes: a generalization of Cheeger's inequality, *Trans. Amer. Math. Soc.* **309** no. 2 (1988), 557–580. MR 0930082. Zbl 0716.60073. https://doi.org/10.2307/2000925.
- [LP13] J. R. LEE and Y. PERES, Harmonic maps on amenable groups and a diffusive lower bound for random walks, Ann. Probab. 41 no. 5 (2013), 3392–3419. MR 3127886. Zbl 1284.05250. https://doi.org/10. 1214/12-AOP779.

JÉRÉMIE BRIEUSSEL and TIANYI ZHENG

- [LPW09] D. A. LEVIN, Y. PERES, and E. L. WILMER, Markov chains and mixing times, Amer. Math. Soc., Providence, RI, 2009, with a chapter by James G. Propp and David B. Wilson. MR 2466937. Zbl 1160.60001. https://doi.org/10.1090/mbk/058.
- [LLR95] N. LINIAL, E. LONDON, and Y. RABINOVICH, The geometry of graphs and some of its algorithmic applications, *Combinatorica* 15 no. 2 (1995), 215– 245. MR 1337355. Zbl 0827.05021. https://doi.org/10.1007/BF01200757.
- [LM00] N. LINIAL and A. MAGEN, Least-distortion Euclidean embeddings of graphs: products of cycles and expanders, J. Combin. Theory Ser. B 79 no. 2 (2000), 157–171. MR 1769197. Zbl 1026.05032. https://doi.org/10. 1006/jctb.2000.1953.
- [LMN02] N. LINIAL, A. MAGEN, and A. NAOR, Girth and Euclidean distortion, *Geom. Funct. Anal.* 12 no. 2 (2002), 380–394. MR 1911665. Zbl 0991. 05037. https://doi.org/10.1007/s00039-002-8251-y.
- [LP16] R. LYONS and Y. PERES, Probability on Trees and Networks, Cambridge Ser. Stat. Prob. Math. 42, Cambridge Univ. Press, New York, 2016. MR 3616205. Zbl 1376.05002. https://doi.org/10.1017/9781316672815.
- [Mat97] J. MATOUŠEK, On embedding expanders into l_p spaces, *Israel J. Math.* **102** (1997), 189–197. MR 1489105. Zbl 0947.46007. https://doi.org/10. 1007/BF02773799.
- [Maz29] S. MAZUR, Une remarque sur l'homéomorphie des champs fonctionnels, Studia Mathematica 1 no. 1 (1929), 83–85. Zbl 55.0242.01. https://doi. org/10.4064/sm-1-1-83-85.
- [MN08] M. MENDEL and A. NAOR, Metric cotype, Ann. of Math. (2) 168 no. 1 (2008), 247–298. MR 2415403. Zbl 1187.46014. https://doi.org/10.4007/ annals.2008.168.247.
- [Mim15] M. MIMURA, Sphere equivalence, Banach expanders, and extrapolation, Int. Math. Res. Not. IMRN no. 12 (2015), 4372–4391. MR 3356758.
 Zbl 1334.46020. https://doi.org/10.1093/imrn/rnu075.
- [NP08] A. NAOR and Y. PERES, Embeddings of discrete groups and the speed of random walks, Int. Math. Res. Not. IMRN (2008), Art. ID rnn 076, 34. MR 2439557. Zbl 1163.46007. https://doi.org/10.1093/imrn/rnn076.
- [NP11] A. NAOR and Y. PERES, L_p compression, traveling salesmen, and stable walks, Duke Math. J. 157 no. 1 (2011), 53–108. MR 2783928. Zbl 1268.
 20044. https://doi.org/10.1215/00127094-2011-002.
- [NPSS06] A. NAOR, Y. PERES, O. SCHRAMM, and S. SHEFFIELD, Markov chains in smooth Banach spaces and Gromov-hyperbolic metric spaces, *Duke Math. J.* **134** no. 1 (2006), 165–197. MR 2239346. Zbl 1108.46012. https://doi. org/10.1215/S0012-7094-06-13415-4.
- [NS11] A. NAOR and L. SILBERMAN, Poincaré inequalities, embeddings, and wild groups, *Compos. Math.* 147 no. 5 (2011), 1546–1572. MR 2834732.
 Zbl 1267.20057. https://doi.org/10.1112/S0010437X11005343.
- [NR03] I. NEWMAN and Y. RABINOVICH, A lower bound on the distortion of embedding planar metrics into Euclidean space, *Discrete Comput. Geom.*

29 no. 1 (2003), 77–81. MR 1946795. Zbl 1050.68154. https://doi.org/ 10.1007/s00454-002-2813-5.

- [OO13] A. Y. OLSHANSKII and D. V. OSIN, A quasi-isometric embedding theorem for groups, *Duke Math. J.* 162 no. 9 (2013), 1621–1648. MR 3079257.
 Zbl 1331.20054. https://doi.org/10.1215/00127094-2266251.
- [PR04] Y. PERES and D. REVELLE, Mixing times for random walks on finite lamplighter groups, *Electron. J. Probab.* 9 no. 26 (2004), 825–845. MR 2110019.
 Zbl 1064.60095. https://doi.org/10.1214/EJP.v9-198.
- [PSC00] C. PITTET and L. SALOFF-COSTE, On the stability of the behavior of random walks on groups, J. Geom. Anal. 10 no. 4 (2000), 713–737.
 MR 1817783. Zbl 0985.60043. https://doi.org/10.1007/BF02921994.
- [PSC02] C. PITTET and L. SALOFF-COSTE, On random walks on wreath products, Ann. Probab. 30 no. 2 (2002), 948–977. MR 1905862. Zbl 1021.60004. https://doi.org/10.1214/aop/1023481013.
- [Rao12] M. M. RAO, Random and Vector Measures, Ser. Multivariate Anal.
 9, World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ, 2012. MR 2840012. Zbl 1259.46002. https://doi.org/10.1142/8678.
- [SC97] L. SALOFF-COSTE, Lectures on finite Markov chains, in Lectures on Probability Theory and Statistics (Saint-Flour, 1996), Lecture Notes in Math. 1665, Springer-Verlag, Berlin, 1997, pp. 301–413. MR 1490046. Zbl 0885.60061. https://doi.org/10.1007/BFb0092621.
- [SCZ16] L. SALOFF-COSTE and T. ZHENG, Random walks and isoperimetric profiles under moment conditions, Ann. Probab. 44 no. 6 (2016), 4133–4183.
 MR 3572333. Zbl 1378.60019. https://doi.org/10.1214/15-AOP1070.
- [Tes11] R. TESSERA, Asymptotic isoperimetry on groups and uniform embeddings into Banach spaces, *Comment. Math. Helv.* 86 no. 3 (2011), 499–535.
 MR 2803851. Zbl 1274.43009. https://doi.org/10.4171/CMH/232.
- [Tes13] R. TESSERA, Isoperimetric profile and random walks on locally compact solvable groups, *Rev. Mat. Iberoam.* 29 no. 2 (2013), 715–737. MR 3047434. Zbl 1267.43004. https://doi.org/10.4171/RMI/736.
- [Yu00] G. YU, The coarse Baum-Connes conjecture for spaces which admit a uniform embedding into Hilbert space, *Invent. Math.* 139 no. 1 (2000), 201–240. MR 1728880. Zbl 0956.19004. https://doi.org/10.1007/ s002229900032.

(Received: July 8, 2016)

(Revised: September 9, 2020)

UNIVERSITÉ DE MONTPELLIER, INSTITUT MONTPELLIÉRAIN ALEXANDER GROTHENDIECK, MONTPELLIER, FRANCE *E-mail*: jeremie.brieussel@umontpellier.fr

DEPARTMENT OF MATHEMATICS, STANFORD UNIVERSITY, STANFORD, CA, USA *Current address*: DEPARTMENT OF MATHEMATICS, UC SAN DIEGO, CA, USA *E-mail*: tzheng2@math.ucsd.edu